

This is to certify that the dissertation entitled

Textual Ideology, Textual Practice: A Discourse-Centered Approach to Protestant Bible Study

presented by

James S. Bielo

has been accepted towards fulfillment of the requirements for the

Ph.D.	degree in	Anthropology
	Palert K %	iteliuck
•	Major Profe	ssor's Signature
	4/19/	07
		Date

MSU is an affirmative-action, equal-opportunity employer

LIBRARY Michigan State University

PLACE IN RETURN BOX to remove this checkout from your record.

TO AVOID FINES return on or before date due.

MAY BE RECALLED with earlier due date if requested.

DATE DUE	DATE DUE	DATE DUE
	,	
		6/07 - /0/10/10-4-0

6/07 p:/CIRC/DateDue.indd-p.1

TEXTUAL IDEOLOGY, TEXTUAL PRACTICE: A DISCOURSE-CENTERED APPROACH TO PROTESTANT BIBLE STUDY

By

James S. Bielo

A DISSERTATION

Submitted to
Michigan State University
in partial fulfillment of the requirements
for the degree of

DOCTOR OF PHILOSOPHY

Department of Anthropology

2007

ABSTRACT

TEXTUAL IDEOLOGY, TEXTUAL PRACTICE: A DISCOURSE-CENTERED APPROACH TO PROTESTANT BIBLE STUDY

By

James S. Bielo

This dissertation examines the practice of small group Bible study among

American Protestants. The analysis focuses on the relationship between the assumptions
and expectations readers hold about the Bible (textual ideology) and their impact on
various forms of discursive action (textual practice). I argue that the textual ideology
surrounding the Bible acts as both a structuring mechanism in the discursive backdrop of
group life, and an interactional strategy participants invoke in pursuit of specific
hermeneutic and rhetorical aims.

Small group Bible study is a key ethnographic site for observing how contemporary American Protestants interact with their sacred text. Bible study is the most common type of small group in U.S. society. More than 25 million Americans gather each week to read and discuss Biblical texts and related commentaries. Small groups play a unique and vital role as sites of knowledge production within Protestant communities. Unlike other congregational activities, Bible study is a site of open, critical dialogue where members engage in an active, reflective negotiation of their faith. Categories of belief, tradition, identity, and practice are reaffirmed and questioned in this context. How

believers understand their relationship with the Bible is the form of knowledge most evident and central to Protestant ethos.

This dissertation is the result of 19 months of ethnographic fieldwork with 19 Bible study groups in six Protestant congregations. Based on the audio recording and observation of 324 meetings, as well as several complementary data collection techniques, I argue that the textual ideology surrounding the Bible consists of six well-articulated principles that address the authority, relevance, and textuality of the Bible.

Each of the three main analytical chapters considers a distinct form of discursive practice that unfolds in dynamic relation with these principles. First, groups employ three interpretive styles when reading the Bible: finding application, establishing meaning, and explaining Biblical texts through inferencing processes. Second, groups rely on shared and idiosyncratic interpretive resources, each of which embeds ideological principles.

And, lastly, I analyze how group participants recontextualize the Bible. I argue that the preferred style of recontextualization works in conjunction with the predominant interpretive concerns in each setting.

This dissertation draws from discourse-centered, interpretive, and praxis frameworks in the anthropological tradition. The arguments I present contribute to debates surrounding the production of knowledge in institutional discourse, intertextual strategies, and practices of collective reading. This research also contributes to an emerging body of work on the anthropology of Christianity.

Copyright by JAMES STUART BIELO 2007 For John S. Bielo, my father, who first instilled in me the values of earnest work, perseverance, and a nuanced eye

ACKNOWLEDGMENTS

This dissertation would never have been possible without the help of many individuals and institutions. First and foremost, I would like to thank the Bible study groups that allowed me to include them in this research. They were kind, patient, intuitive, insightful, and forgiving; even when I failed to be. I owe a special thank you to Tony and Jan Luttrell. Not only did they provide warm friendship and wise council, but they provided a home when I was in need.

My family – Judith, John, Mary, Roger, RJ, Ed, Julie, and Clay – have been supportive throughout my undergraduate and graduate careers. They maintained a loving attitude, even though it has meant many more months apart than together.

My friends and colleagues at Michigan State University are largely responsible for helping me maintain sanity and humor during my graduate career. I have been especially grateful for Dillon Carr, Timothy and Kaley Hepburn, Kate Patch, Duane Quates, Ryan Tubbs, Mary Megyesi, Aaron Naumann, Jubin Cheruvelli, and Marita Eibl.

The department of anthropology at Michigan State University has also been a constant source of support and encouragement. The administrative assistants – Gail Barricklow, Nancy Smith, Roxanne Moran, and Peggy Medler – provided wisdom, humor, and a generous ear for listening. The entire faculty at MSU has shown their support in various ways. Scholarly exchanges during my participation in the weekly Brown Bag series helped assure me that my ideas made sense to others beside myself.

Scholarly exchanges in other venues have also helped this dissertation rnaterialize. I would like to thank the participants of the SALSA 2006 conference; the SSSR 2006; and the AAA 2006 for feedback and comments. My undergraduate

professors at Radford University – Melinda Wagner and Mary LaLone – have continued to provide help, insight, and encouragement over the years since I left southwest Virginia.

And, personal communication with Judy Pugh, Brian Malley, Jon Bialecki, and Heather Van Wormer have helped shaped my thinking and analysis.

I owe an especially large debt of gratitude to my dissertation advisory committee. Dr. Amy Derogatis joined the committee near the end of my fieldwork. She has been an insightful and reliable reader, and a valuable resource in the field of American religion. She contributed a great deal to this dissertation in only a short time. Dr. Mindy Morgan was utterly consistent in her commitment and encouragement. Her trust and interest in my work has never failed to leave me inspired and reaffirmed. Much of this dissertation would not have materialized without her mentoring. I will be honored to consider myself a colleague of Dr. Morgan's. Dr. David Dwyer has been a source of wisdom, intellect, and humor since I met him in my first class at Michigan State. He has represented the best of what a life in academia can offer, what can be achieved with decades of research and teaching, and how to keep a boyish enthusiasm through the course of it all.

Finally, I owe the greatest debt of all to Dr. Fredric Roberts, my dissertation advisor. His influence on me – as an anthropologist, a scholar, and a person – is more profound than I could ever articulate. From my first day on the Michigan State campus he has challenged me, encouraged me, and demonstrated great faith in my abilities as an ethnographer, analyst, and writer – even when I was undeserving. He is a father to me.

TABLE OF CONTENTS

LIST OF TABLES.	ix
INTRODUCTION: BIBLE STUDY, ETHNOGRAPHY, AND	
ANTHROPOLOGY	1
Theoretical Frameworks	
The Anthropology of Discourse	3
The Anthropology of Christianity	
Research Methodology	14
Sampling	14
Data Collection	16
Data Analysis	
The Dissertation's Structure	24
CHAPTER 1	
SMALL GROUPS AS CULTURAL PRACTICE	27
The Project Sample: a congregational perspective	27
Similarities in Small Group Life	
The Project Sample: eight ethnographic portraits	
How Small Groups Work	62
Interactive Frames	64
Participation Structure	64
Text Selection.	
Using the Bible	71
CHAPTER 2	
TEXTUAL IDEOLOGY AND THE BIBLE	75
What is Textual Ideology?	76
Textual Ideology and the Bible: six core principles	81
Biblical Authority	82
Biblical Relevance	88
Biblical Textuality	
Textual Ideology: responses to four probable questions	99
CHAPTER 3	
INTERPRETING THE BIBLE IN SMALL GROUP DISCOURSE	108
Five Case Study Groups	110
Applying Biblical Texts	111
Personal Application	113
Collective Application	
Global Application	
Establishing Textual Meaning	
Explaining Biblical Texts	
Inter-Relation among Interpretive Styles	125

Small Groups as "Interpretive Communities"	134
CHAPTER 4	
THE INTERPRETIVE MATRIX OF SMALL GROUP DISCOURSE	140
Shared Resources	
Congregations as Resources	
Denominations as Resources	
Denomination as Other	
Denomination as Self	
Group-Specific Resources	
Individual Resources	
The Significance of the Interpretive Matrix	
CHAPTER 5	
RECONTEXTUALIZING THE BIBLE IN SMALL GROUP DISCOURSE	174
Recontextualization as a Discourse Strategy	175
Sampling the Group Participants	
Functional and Interpretive Qualities of Recontextualization	177
Recontextualization and Interactive Function	178
Recontextualization as Meta-Commentary	179
Recontextualization as Application	
Formal Properties of Biblical Recontextualization	183
What Does Bible Study Discourse Tell Us About Biblical	
Recontextualization?	187
What Does Biblical Recontextualization Tell Us About Bible Study	
Discourse?	
Textual Ideology	
Protestant Discourses	
Interactive Norms	
Concluding Remarks	205
CHAPTER 6	
CONCLUSIONS, CONTRIBUTIONS, AND FURTHER QUESTIONS	
Chapter Summaries	
Further Consequences of Textual Ideology, Textual Practice	
The Interpretive Imagination of American Protestantism	
The Structuring Influence of Protestant Traditions	
Small Groups as a Site of Dialogue	
Broader Significance	
Textual Ideology	216
APPENDICES	222
Appendix A: Sample Transcript Page	
Appendix B: Research Questionnaire	
Appendix C: Sample Study Guide Sheet for Tuesday Men	

REFERENCES		220

LIST OF TABLES

TABLE 1: DATA FOR SIX PROJECT CONGREGATIONS27
TABLE 2: SMALL GROUPS ATTENDED IN SIX PROJECT CONGREGATIONS44
TABLE 3: DATA FOR EIGHT GROUPS USED FOR DISCOURSE ANALYSIS (1)45
TABLE 4: DATA FOR EIGHT GROUPS USED FOR DISCOURSE ANALYSIS (2)45
TABLE 5: QUESTIONNAIRE DATA FOR TEXTUAL IDEOLOGY (1)104
TABLE 6: QUESTIONNAIRE DATA FOR TEXTUAL IDEOLOGY (2)106
TABLE 7: STUDY TOPICS FOR FIVE PROJECT GROUPS110
TABLE 8: TUESDAY MEN STUDY GUIDE ANALYSIS128
TABLE 9: TUESDAY MEN INTERPRETIVE RESOURCES147
TABLE 10: INTERPRETIVE RESOURCES OF SEVEN PROJECT GROUPS148
TABLE 11: ICONOCLASTS' INTERPRETIVE RESOURCES160
TABLE 12: MONDAY GROUP INTERPRETIVE RESOURCES
TABLE 13: RECONTEXTUALIZATIONS FOR 14 BIBLE STUDY PARTICIPANTS
TABLE 14: DEFINITIONS FOR RECONTEXTUALIZATION STYLES
TABLE 15: METONYMIC RECONTEXTUALIZATIONS FOR 14 BIBLE STUDY PARTICIPANTS
TABLE 16: RECONTEXTUALIZATION AS INTERACTIONAL SUPPORT AND CHALLENGE

Introduction: Bible Study, Ethnography, and Anthropology

Every week tens of millions of Americans gather in homes and churches for small group Bible study. It is, in fact, the most common type of small group in American society (Wuthnow 1994a). These groups assume a variety of forms. There are men's, women's, couples', and singles' Bible studies. They are organized by age, occupation, and maturity of faith. Some groups study directly from the Bible, while others read non-scriptural texts by Christian authors or devote discussion to a Biblical topic or theme. In any case, small groups are among the most meaningful and important practices for American Protestants.

This dissertation is about the discourse in small group Bible study. My descriptions and analyses are based on 19 months of ethnographic fieldwork conducted among six Protestant congregations in a Midwestern city, and in 19 groups from these churches. The central issue I address is the interaction between discourse participants and their sacred text, the Bible. How individuals and groups interact with Biblical texts is fundamental to understanding what occurs in the small group setting, and why this practice is so significant for American Protestantism.

This moment in American history is an appropriate time for this topic. As I began to write the initial section (which eventually became chapter 5) the Gospel of Judas¹ was released to the public. A few weeks later, the film version of The Da Vinci Code² was released. Both re-ignited debates about what constitutes Christian scripture and identity.

¹ The Gospel of Judas is a Gnostic text dating to the 2nd century A.D. It was first discovered in 1983 and released to the public as an English translation in April of 2006. The text portrays Judas as a confidant of Jesus Christ who obeyed a direct command from Jesus to betray him.

² The Da Vinci Code (2003) is a mystery novel written by Dan Brown. The book's plot includes suggestions that Mary Magdalene and Jesus were married and had children together. This is a heretical belief in the orthodox Christian tradition.

Midway through the year of data collection, controversy swirled around Intelligent
Design theory after court cases considered whether or not this view of human origins
should be taught in public schools alongside Darwinian evolution. This brought attention
to Biblical themes of creation, as well as what role the Bible might play in science
education in public schools. Two months before I started collecting data George W. Bush
was re-elected to the U.S. Presidency. Much was (and is) made about the political capital
of 'Christian Right' voters, and the use of Biblical rhetoric to attract this audience. The
earliest stages of this project came on the heels of Mel Gibson's film The Passion of the
Christ. This too raised questions about the accuracy, consistency, and authority of the
Bible. Biblically oriented books like The Prayer of Jabez (2000). The Purpose Driven
Life (2003), and Your Best Life Now (2004) appeared on religious and secular best-seller
lists prior to and throughout the research. Sales reached to tens of millions, and in some
cases entire industries of products and programs ensued.

Christianity clearly pervades private and public life in America. As a native anthropologist³, I believe ethnographic attention to the lives of Christian believers – what they do, what they say, and what they care about – should reflect this presence. Meeting together in small groups is only one thing that American Protestants do. But, it is among the most significant things they do. Their discursive relation with the Bible is not the only thing they talk about, or care about. But, it is something they spend much of their time talking about, and care a great deal about.

³ I consider myself a native anthropologist in this research in two regards. 1) I am a native-born United States citizen studying a cultural practice by US citizens in the United States of America. 2) I am a Christian believer studying other Christian believers, although many adhere to significantly different theological, social, and political beliefs than I do.

Before presenting the analyses that support these claims, it is important to place my research among other scholarly work. It is equally important to explain how the data for these analyses were collected. First, I discuss the relevant theoretical frameworks to my research. I then review my methodology.

Theoretical Frameworks

I situate my research within two theoretical frameworks, the study of discourse and the study of Christianity. My goal is to demonstrate how these two intellectual traditions made my work possible and how my work contributes to them both.

The Anthropology of Discourse

One of my key theoretical decisions was to view Bible study as a discourse event, and not simply as a general form of cultural practice. This focused my analysis on the linguistic, interactive, and textual features of group meetings. This orientation to studying cultural life has a rich history in anthropology. The inception of the ethnography of speaking in the early 1960s marked a turning point in how anthropologists interested in language approached their object of study.

The ethnography of speaking emerged in conjunction with socio-linguistics (Hymes 1964). Both sought to broaden their field of inquiry to include new questions about the relationship between language, language use, and social action. Linguistic anthropologists were interested in research outside the Whorfian tradition of linguistic relativity, and the grammatical description and classification of non-Indo European languages. The shift toward speech as an object of analysis also occurred in response to structural linguistics, where scholars in the Chomskyian tradition were favoring issues of "competence" over issues of "performance" (ibid). Sociolinguists were interested in

research that went beyond the linking of social and linguistic variables (Duranti 2003). The result was a shift in interest from the structuralist preoccupation with language as a system to the actualization of that system by real people in real situations.

Research in the ethnography of speaking proceeded from the assumption that speech is not a completely individualized phenomenon, but is patterned and shared much like any other form of social practice. As a methodological tool, ethnographers of speaking became interested in speech events and speech communities. For the former, certain types of speech activities were considered central to the (re)production of cultural meaning. And, analysis of how speech is organized in these events can capture the cultural work being accomplished (e.g., Frake 1964). The concept of speech communities clarified the problematic assumption that sharing a language system also meant sharing ways of speaking. In short, not all potential speakers are legitimate speakers for every setting where speech occurs. Two analytical interests developed from these methodological developments. Interest arose about different types of speech, primarily how groups of speakers differentiated among genres (e.g., joking, gossip, storytelling). Interest also arose about how speaking roles were reserved for certain social actors, and the communicative competence needed to adequately fulfill those roles (Bauman and Sherzer 1975).

The outcome of these developments was an interest in the concept of discourse (i.e., language-in-use). Much like researchers in other disciplines, anthropologists have attributed a variety of meanings to "discourse." There are at least three overlapping and mutually reinforcing areas of theoretical concern where the concept of discourse has been found useful (cf., Abu-Lughod and Lutz 1990). Discourse has been conceptualized as the

signifying and communicative behavior that occurs within a single type of bounded event. This approach focuses on what features typify a particular discourse event. It is generally concerned with questions of conversational structure (e.g., Goodwin and Heritage 1990), speaking roles (e.g., Keenan 1974), interactive norms (e.g., Tannen and Wallet 1987), and the incorporation of linguistic devices into speech (e.g., Puckett 2000). In several sections of this dissertation I analyze Bible study as a speech event. For example, chapter 1 illustrates the varying, and at times competing, interactive frames that structure what participants expect from Bible study.

Discourse has also been conceptualized as the ongoing signifying and communicative behavior that occurs in relation to particular social activities. This approach focuses on what discourse is about. It is generally concerned with questions about the dialogical construction of ethos (e.g., Bowen 1992), identities (e.g., Mould 2003), and social relations (e.g., Lutz 1988). This is my primary approach to discourse in this study. For example, the discursive practices analyzed in chapters 3 through 5 are involved in the ongoing articulation of the textual ideology described in chapter 2.

Lastly, discourse has been conceptualized as a society-wide, historical force that weighs heavily on the formation of cultural attitudes and practices. This approach focuses on the impact of discourse in structuring social life. It is generally concerned with questions about the formation of ideological complexes, such as madness, truth, or science (e.g., Foucault 1981). This dissertation is least oriented toward this approach because of the ethnographic (but not necessarily historical) nature of the data, and my lack of overt interest in issues of power⁴. Indeed, the work presented here is probably

⁴ A central assumption in this line of research is that power is a productive force (not necessarily a restrictive one) that is pervasive throughout all of social life. The practice of group Bible study is not

more a product of such discourse than an analysis of it. For example, I accept and employ many scholarly categories (e.g., "discourse," "Protestant," "interpretation," "conservative," "belief," etc.) rather than attempt to deconstruct and situate them within power-laden processes of knowledge production.

All three approaches understand discourse as a process of linguistic production (i.e., as producers of texts). Thus, the analysis of discourse begins with the analysis of texts in order to understand the processes of production. This is the basis for what Sherzer (1987) and Urban (1991) have termed a "discourse-centered approach" to language and culture. They argue that discourse should not be treated as a means toward pursuing other questions, but should itself be the subject of analysis. This argument presumes that discourse is where cultural life unfolds, not simply a reflection of culture. As a result, a 'discourse-centered approach' focuses on formal, stylistic, organizational, and structural features, not solely the semantic content of talk. I do not restrict my analyses to issues of discursive form, but I do emphasize close and detailed attention to discursive practice, and do not just look "through talk" at what it describes, but look at discursive (inter)action.

I also draw heavily upon the Bahktinian tradition of intertextuality: "the relational orientation of a text to other texts" (Bauman 2004: 4). This term originates with Mikhail Bahktin's (1934) concept of dialogism, wherein all utterances are produced and received in relation to their previous and expected uses. Julia Kristeva (1986) coined the term 'intertextuality' as part of her literary analyses, intending to capture the way a single text is the result of, carries the imprint of, and incorporates multiple texts. To understand

exempt from this claim, particularly when we consider processes of text availability, text selection, and facilitator-participant and participant-participant dynamics. However, my interest in the relationship between textual ideology and textual practice shifts the focus away from these issues.

texts as intertextual productions has consequences for meaning, narrativity, authorship, audience reception, and textual appropriation. Anthropologists interested in discourse analysis have found the concept useful for several reasons. Studies of performance, reported speech, genre switching, and recontextualization presume that texts are rarely first time productions, and that all texts enter social life connected to other texts (Bauman 2004).

The intertextual character of small group discourse is important in two ways. As a discourse event, small groups involve the presence of multiple textual artifacts and a litany of intertextual references. Discourse participants consult and read from Bibles (often multiple translations and study Bibles with extensive footnotes), devotional materials, journals and study texts to consult and read from. Participants draw on television and movie segments, recent media stories, radio broadcasts, hymn and praise song lines, favorite Christian authors, recent sermons, poems, books, devotionals, Biblical commentaries, and (of course) Biblical texts. Small group discourse draws upon an intricate, dynamic, and thoroughly intertextual matrix of interpretive resources (cf., Chapter 4).

The intertextual practices of decontextualization and recontextualization – "the process of rendering discourse extractable...[and lifting text] out of its interactional setting" (Bauman and Briggs 1990: 73) – also play a key role in small group discourse. Participants constantly invoke Biblical texts for a variety of purposes. Cross-cultural research has emphasized the importance of recontextualization for garnering social authority and obfuscating social responsibility. For example, Joel Kuipers (1990) examined how Weyewa ritual specialists invoke ancestors' words to repair social

relations damaged by unfulfilled obligations. The appropriation of ancestral words provides them with an indisputable authority, and their repair work ensures that future relations with ancestors are prosperous. Judith Irvine (1996) showed how responsibility in speech is blurred using the Wolof genre of "xaxaar," the insult poem. Speakers understand these poems as dialogically formulated (i.e., not of their own creation), creating social distance between themselves and xaxaar performances. Due to this recontextualized nature, responsibility shifts from the speaker to a shared body of tradition, and culpability remains indeterminate. Webb Keane (2004) has observed recontextualization as a particularly potent strategy in religious discourse because it plays with categories of authorship and authority. In chapter 5, I present an extended analysis of how participants recontextualize Biblical texts in small group discourse.

This research emerges from three other fields of inquiry: the ethnography of reading, praxis theory, and interpretive anthropology. The ethnography of reading emphasizes the relationship between orality and textuality in contexts where reading is performed collectively (Boyarin, ed. 1992). These studies have argued that the interactions between reader-text and reader-reader must account for previous readings and the social processes of multiple readers interacting with texts together. For example, Jonathan Boyarin analyzed how participants in a Jewish yeshiva used their collective reading of Torah and midrashic texts to communicate notions of personhood and Hebraic tradition (Boyarin 1992). My dissertation applies the crucial insight from this framework that texts are not simply records of dialogue, but occasions for joining dialogue. Bible study participants do not just read Biblical texts; they place themselves in relations of parallel and dissonance to Biblical narratives and characters.

This study has been significantly informed by two theoretical traditions linked with the study of discourse. The first are theories of social praxis, articulated most clearly by Pierre Bourdieu (1977). Praxis theory understands cultural life to be the result of dialectical relations between institutional structures and the predispositions that social actors inherit. Individual and collective actors are understood as products of their cultural history and active (re)producers of that cultural universe. Institutions are not immutable structures, but resources that can be appropriated or ignored, accepted as is or endowed with new meaning.

Studies of discourse have applied a praxis approach to numerous research questions. The actor-oriented nature of praxis theory has been useful for understanding variation within discourse, the link between discourse and identity, strategies for textual interpretation, and discourse as a site for power struggles (cf., Fairclough 1989). The analyses that follow demonstrate the structured and structuring relations that participants have with group discourse. Chapter 4 illustrates how individuals and groups use the interpretive resources that are meaningful in their lives.

Discourse studies also are tied closely to the hermeneutic tradition in the social sciences. This seems almost inevitable, given their shared interests in texts, interpretation, and meaning. The anthropology of discourse and interpretive anthropology have roots in hermeneutic philosophy and the philosophy of language, particularly in the work of Paul Ricouer. Ricoeur (1971) articulated best the similarities between interpreting textual artifacts and viewing social action as a textual form. Ricoeur and others were occupied with what constitutes a text, the interaction between text and interlocutors, and the interplay of different interpretive styles. Interpretive anthropology applied these

perspectives to cultural life, asking how actors understand their relation to the surrounding social universe, how they make sense of that universe, how they make it meaningful, and how all these are made public (e.g., Geertz 1973).

Anthropologists interested in discourse share with this tradition the notion that human action is fundamentally symbolic action (Sherzer 1987). The most developed symbol systems, of course, are verbal and non-verbal forms of communicative behavior. They also intersect as enterprises that represent practices of signifying meaning. And, whereas interpretive anthropology analogized action to text, studies of discourse are directly involved in demarcating spoken texts. From its inception, this dissertation has been firmly situated within the interpretive tradition of anthropology. Through this brand of discourse analysis, I hope to make clear what is so meaningful about participating in small groups and what participants mean by certain forms of talk. I also hope this dissertation will provide a window into how these groups view themselves, make sense of their world, understand their relationship with it, and make these sentiments public via small group discourse.

In the following chapters, I employ these frameworks to understand the discourse of small group Bible study. I attend to the content, ways of speaking, formal properties, intertextual strategies, collective reading practices, and hermeneutic features that characterize how group participants interact with one another and with Biblical texts. The small group setting is an ideal venue for applying these frameworks because of its quality as an organized, bounded speech event that incorporates textual, interactive, and interpretive dynamics.

The Anthropology of Christianity

This dissertation also emerges from social science research on Protestant

Christianity. Interestingly, sociologists are responsible for the most coherent body of research in the study of American Protestantism. The ethnographic study of congregational life is most relevant. Local congregations have been described as the "bedrock of American religion" (Warner 1994). They have been studied as "local cultures," institutions that provide a sense of identity and belonging (Becker 1999; Roberts 2005; Wind and Lewis 1994); bearers of denominational traditions (Bass 1994; Roberts 2005); arenas for organizing strategies and resources for community action (Ammerman 1997); and institutions integrated into local religious ecologies (Eiesland 2000). Each of these approaches understands congregational life to be central to the lives of individual believers and to the structure of American Protestantism.

Congregational studies have consistently identified Bible study as vital to the cultural work accomplished by congregations. Robert Wuthnow described the significance of small groups in the broader context of American society, arguing that a small group movement is reshaping Americans' sense of community. His large-scale survey effort identified Bible study⁵ as the most prolific form of small group, with an estimated 30-40 million weekly participants (1994a). Jody Davie, an anthropologist, carried out ethnographic research with one women's Bible study group in a New England Presbyterian church. She described Bible study as unique among all congregational activities because it provides participants with the opportunity for "dialogue [and] active personal negotiation of meaning" (1995:1). Yet, small group Bible study has not received

Wuthnow distinguished between "Sunday school" and "Bible study," but for the purposes of this issertation, they are similar enough activities to be included in the same category.

the research attention to match these claims of personal and collective significance. This dissertation responds to this lack of research by providing in-depth ethnographic and discursive analyses of Bible study in numerous small group settings.

The study of Christianity has been far less popular among anthropologists than among sociologists of religion. Harding (1991) argues this is because Christianity, particularly Fundamentalist Christianity, represents a "repugnant" cultural Other – an Other seen as diametrically opposed to an anthropological Self. Alternatively, Robbins (2004) explains that anthropological studies of Christianity in non-Western societies have de-emphasized Christianity for several reasons. He suggests that such work has traditionally viewed Christianity as either the harbinger of cultural erosion (e.g., Harkin 1997), a syncretized form that has transformed into something other than Christianity (e.g., Burt 1994), or as the bearer of other epistemologies, such as modernity or capitalism (e.g., Comaroff and Comaroff 1991). Robbins includes his own study of the Urapmin of Papua New Guinea as part of a growing body of research taking Christianity seriously, as something that is itself fully culturally and fully worthy of study (cf., Cannell 2006). My research is part of this growing body of work. In particular, I engage Biblicism as an anthropological issue within the study of Christianity.

Simply defined, Biblicism refers to how the Bible is understood and the relation of those understandings to Bible use. Susan Harding (2000) described how the sermonic discourse of Fundamentalist preacher Jerry Falwell aligns with Biblical themes of sin, redemption, salvation and righteousness. Falwell's narrative practices invite listeners to hear and interpret his words as though they were reading the Bible. Vincent Crapanzano (2000) examines American Fundamentalists' claims of interpreting the Bible literally,

and how their hermeneutic approach results in practices (e.g., biblical counseling) and ethos (e.g., conceptualizing time and history). Brian Malley (2004) outlines the epistemological structure of Evangelical Biblicism based on ethnographic observations of a Baptist congregation in the American Midwest. I discuss his framework in detail in chapter 2.

Matthew Engelke (2004) explains the absence of Bible use among Masowe apostolics in Zimbabwe as the shifting of authority in religious knowledge from written texts to performed spoken texts. The emphasis on a "live and direct" experience with the divine leads them to "privilege the spoken word because of its immediacy and its intimacy" (ibid: 87; cf., Pulis 1999 regarding Jamaican Rastafarians). Eva Keller's ethnography of Seven Day Adventists in Madagascar suggests that individual and group Bible study is the most important part of their religious practice and identity. It is not a means to an end; Bible study (with its associated intellectual inquiry) attracts adherents. "The Bible is not regarded as a fetish that has intrinsic power, rather it is thought of as a collection of wise and truthful words that need to be read, analyzed, and understood" (2005: 92). Most recently, Erika Muse (2005) explained Bible use as integral to the ethno-Christian identity of Chinese Evangelicals in America. They use the Bible as a textual resource to construct a sense of self that counters the stereotype of the model minority.

The thread connecting each of these studies is an interest in exploring the interaction that occurs between Christian communities and their sacred text, the Bible. In chapter 2, I introduce the concept of textual ideology, an organizing framework that guides Bible use among believers. Chapters 3, 4, and 5 demonstrate how the discursive

practices of small group Bible study relate to this textual ideology. Thus, this dissertation contributes to the anthropology of Christianity, and particularly the issue of Biblicism.

To critically test these theoretical ideas about discourse and the anthropology of Christianity, I developed a systematic methodology for data collection and analysis. The next section explains the research design, as well as the types of data collection, and analysis I employed. I begin with my rationale for choosing the six congregations that comprise the project sample.

Research Methodology

Sampling

When the year of data collection ended in December 2005, I had attended 324

Bible study meetings among the 19 project groups (see chapter 1 for a description of these groups). I tape-recorded 167 meetings, though many had yet to be transcribed. To borrow from Robert Frost, I was "overtired of the great harvest I myself desired⁶."

The data are the result of fieldwork that began during late May of 2004. I began by compiling a list of Protestant congregations within a 30-minute car ride of the city center. The research site is a Midwestern city with a population between 100,000 and 150,000. The surrounding cities and townships increase the total population to between 200,000 and 250,000. Using up-to-date telephone and Internet listings, I located 354 Protestant churches in the city and surrounding areas. The primary variable for reducing this was to only focus on congregations that offered extensive Bible study programs⁷.

⁷ This sampling rationale produced at least one inherent bias in the resulting data. By focusing on congregations with extensive small group opportunities, I most likely missed congregations that were

⁶ This quote comes from Robert Frost's 1914 poem, "After Apple-Picking."

This ensured that the eventual project churches would provide ample, non-conflicting opportunities for data collection. I narrowed the sampling pool to 48 by examining church websites, talking to regional denominational representatives, and placing phone calls to churches.

Next, I visited these 48 churches and presented the project to the senior pastor. These introductory visits ended in one of three scenarios: 1) the pastor declined participation on behalf of the church, usually because the congregation was in the midst of some type of struggle or controversy. 2) I learned that the small group opportunities were not as abundant as I had thought, which meant removal from the sample. 3) The pastor expressed interest in having their church participate. In these cases, I left a packet of materials describing the project with the pastor for he/she to present to the church board, elders, or other decision-making body. By late September of 2004, I had narrowed the list of 48 to 10. These ten were deemed appropriate for the project based on their small group programs, and had also granted me permission to approach individual groups and request their consent for inclusion in the project.

For the remainder of the fall I visited individual groups to solicit their consent, as well as get a sense for different group environments. This allowed me to further narrow the list of 10 to the six congregations that constituted the final sample. From the beginning of the sampling process, my intention was to divide the final six into three mainline churches and three non-mainline churches. The three mainline churches were all affiliated with the United Methodist denomination. The three non-mainline congregations

struggling with membership numbers. In turn, small group discourse that addressed local church struggles was not a part of my data.

15

were from the Lutheran-Church Missouri Synod, Restoration Movement, and Vineyard Fellowship denominations. (See chapter 1 for descriptions of these Protestant traditions.)

Data Collection

I relied on five techniques for collecting data: audio-recording group meetings; field notes; ethnographic interviewing; a text archive; and a questionnaire. 1) This data derive primarily from the audio recording and transcription of Bible study meetings (see below for a discussion of the transcription process). Discourse analysts use audio-recordings to capture language-in-use and recreate speech events through transcription (cf., Bernard 2006, Johnstone 2000). Tape-recording has several advantages over hand-written notes to represent discourse. It allows for verbatim records of speech events, providing a more accurate and reliable corpus of data. It frees the ethnographer to focus on other interactive and contextual features in his or her fieldnotes (see below for a discussion of field notes). Once the recorder becomes an accepted presence in the group, it gradually becomes less distracting to participants than an ethnographer who is constantly scribbling notes. And, in group situations where I was expected to participate, as a speaker or as a reader, tape recording allowed me to do this without losing valuable data.

As I mentioned above, 167 of the 324 meetings were tape-recorded, including all of those for the eight groups used in the analyses for chapters 2 through 5. The recordings started at the beginning of each discussion, and were stopped when discussion ended. This excluded opening and closing prayers, as well as the interactions that occur before and after the study proper. The informed consent agreement granted individuals the right to request at any point for the tape to be stopped.

I did not record the remaining 157 meetings because the group involved refused to be recorded, or I decided that the length and/or quality of the discussions did not merit recording. These meetings were confined to seven of the project groups. In hindsight, the decision not to record was a poor choice for three of the groups. A Sunday school class in one of the United Methodist churches read through the New Testament book of Romans. A Lenten study in another United Methodist church did a book study that ended up focusing extensively on issues of Biblical accuracy and authenticity. And, a Sunday school class in the third United Methodist church began with a book study, but then switched to a study of the Sermon on the Mount in the Gospel of Matthew.

2) Three types of field notes are an integral part of my data: (a) those taken while recording; (b) those taken while not recording; and (c) those taken before and after meetings. When the tape recorder was running, my field notes focused on three things. First, I noted non-verbal and interactive cues that accompany talk. This encompassed a wide range of behaviors, including: facial gestures, bodily gestures, and deictic references (e.g., the "she" in "as she was saying earlier"). Second, I described behavior of non-speakers during discussions. This included a range of features as well, including: indications of agreement or disagreement; indications of confusion or ambiguity; and locating readings in a Bible or other text. Lastly, I wrote down theoretical questions and ideas that occurred to me during discussions. These included analytical questions about the group I was observing, comparisons with the discourse and interactive features of other groups, and issues applicable to the entire sample. The first two types of notes are what Bernard calls "descriptive notes...the meat and potatoes of fieldwork" (2006: 397).

The third type of notes included both "methodological" and "analytic" notes (ibid: 395-8).

In unrecorded group situations, the goal of my field notes was to sketch the discussion. Capturing the entire conversation verbatim was not possible, so I focused on keeping track of who spoke and in what order. For each speaker's contribution, I indicated his or her central point. Whenever possible, I filled in conversational details and included exact quotations. I also returned to these sketches as soon as possible after the meeting to fill in any details I could recall from memory.

I also have fieldnotes from before and after meetings. These include my interactions with participants and interactions I observed. These notes provided a rich source of information about each group. They include information on who socialized together, what types of conversations they had outside the meeting proper, and recommendations of reading materials. It was common for participants to use this time to loan and recommend different texts to each other.

Field notes were recorded on site in paper notebooks. I reserved one notebook for each congregation. I transcribed these notes as soon as possible following the meetings, usually the same day or in some cases, the day after. Separate files were maintained for descriptive, methodological, and analytic field notes.

3) I employed informal and formal ethnographic interviewing. Bernard describes informal interviewing as "conversations heard during the course of a day in the field" (2006: 211). These happened almost exclusively before and after meetings. This was convenient because all of the participants were in one place, and because the interview topic was also the purpose of being together. In my field notes, I often recorded specific

questions to ask individuals regarding a comment they made during discussion, a text they referenced, or a discursive tendency I was observing about the group as a whole. I also used informal interviewing to gather much of the introductory information about each group: how long they had been meeting, how long individuals had been coming, other studies they have done, and studies they liked best and least. The information gleaned from this mode of informal interviewing proved invaluable when conducting analyses and writing descriptions of the churches and groups.

The semi-structured interviews were intended for group facilitators only. Bernard describes this type of ethnographic interviewing as "freewheeling...[but] based on the use of an interview guide...a list of questions and topics that need to be covered in a particular order" (ibid: 212). My plan was to conduct at least one of these with each facilitator regarding group history, decisions about text selection, idiosyncratic questions pertaining to each group, the participation order, and leadership style. For groups where I observed multiple studies, I hoped to conduct multiple interviews. Ultimately, I conducted these interviews with nine of the 15 facilitators, though not all in the format I planned. In several cases, I conducted the interview via telephone or email, as opposed to the ideal of face-to-face interaction. For these interviews I did not use a tape recorder because hand-written notes were adequate to capture the desired information.

4) I assembled a text archive for each group. Due to my theoretical focus on intertextuality, I devoted methodological attention to a systematic collection of the textual materials I discovered during research. The archive included texts I encountered during Bible study discussions, before and after meetings, and during interviews. By the end of the fieldwork, this archive included texts used for study and those recommended amongst

participants. It also consisted of texts cited during group discourse, including: fiction and non-fiction books, movie and TV references, hymns and praise songs, Bible commentaries, congregational newsletters, Internet printouts, devotional materials, newspaper articles, and periodical articles. I collected bibliographic information about each text and, whenever possible, obtained a copy.

5) I distributed a questionnaire consisting of 10 sections (cf., Appendix 2). There were closed and open-ended questions in each of the following areas: personal background, religious background, church attendance and participation, Bible study participation, media preferences, experience with Christian texts and authors, Bible reading habits, Bible translation preferences, help sources for Bible reading, and attitudes towards the Bible. The sections dealing with the Bible were particularly helpful for gaining insight into the issue of Biblicism; in particular the textual ideology described in chapter 2 and the analysis of Bible interpretation in chapter 3. The questionnaire was distributed toward the end of the fieldwork year and was collected through the spring of 2006. In total, I submitted the questionnaire to 91 group participants with a return rate of 74% (N=67).

Before moving on to the final section of this introduction, some reflexive comments are necessary about my role as a participant observer. The ethnographic descriptions in chapter 1 include a discussion of how I perceived my role in each of the project groups, as well as the potential consequences of my involvement. However, a more general statement is in order here. I made clear to each group when I first requested its participation that I preferred to be a silent observer, but would make whatever adjustments necessary to put them at ease. For some, this meant I could be completely

silent. For others, it meant I would need to be an active participant. I also made clear to each group that they were always welcome to approach me individually about any aspect of my personal, academic or religious background. Participants from each group did so, and I had the benefit of engaging in detailed, intellectual, and emotional conversations. I came to consider many of them as friends and spiritual mentors. In short, I was primarily a silent observer, but in no case was I a distant, unknown observer.

Data Analysis

Transcription is the first step in any discourse-based analysis. Tape-recording ensures a more accurate and reliable corpus of data, but issues of selectivity and representation remain. Transcription is a process laden with the researcher's theoretical, methodological, and analytical interests (Ochs 1979). In accounting for this decision-making process, the transcripts I produced include certain features of talk and interaction, but not others. Due to my theoretical interest in intertextuality, and particularly recontextualization, the transcripts mark any form of intertextual reference. I used a special indication for recontextualizations of the Bible. Paralinguistic features that impact meaning were marked, such as: heightened stress on utterances, changes in intonation, and voice imitation. I also indicated certain interactive features, such as: interruptions, pauses, and laughs. On the other hand, since this is not a linguistic analysis, other features were excluded, such as: phonology, crutch phrases, or breathing and other habits that break up talk (see Appendix 1 for a sample transcript)⁸.

⁸ The transcription process began in January of 2005, and ended in February of 2006. The time gap between recording and transcribing meant there was potential for me to lose track of key observations and insights in evaluating the data. To balance this, I compiled monthly reports beginning in January that reflected the previous month of data collection. Each report included sections on the following: summary of the data for each group; theoretical frameworks appearing relevant in my analytic field notes; methodological possibilities; and fieldwork difficulties. Reports were created every month from January through May, and every other month through December. These eight reports – and my guidance

The analyses for chapters 2 through 5 was carried out separately, but all followed the same six-step process⁹. 1) I constructed an inductive list of questions based on the monthly reports, field notes, and transcripts. These questions addressed the descriptive (what, when, who) and explanatory (how, why) qualities of each chapter topic. Questions were posed about the topic on the level of the entire sample, individual groups and individual speakers. In the analysis of chapter 3, for example, the compiled list included questions about what interpretive styles I observed, their function in group discourse, the discourse strategies they incorporated, and the groups and individuals that used them.

- 2) A round of "open coding" (Bernard 2006) was conducted. Here, I read and coded for themes and patterns relevant to the chapter topic. Any observation about the topic, be it anomalous or recurring, was included in this codebook. Note that this was the fourth complete experience of each meeting: recording, transcription, the initial read, and the open coding read.
- 3) I compared results of the open coding for each group. These results were viewed both qualitatively and quantitatively. Qualitatively, I used *AtlasTi*¹⁰ to view the coded sections side-by-side. This allowed me to consider recurring discursive features for each theme, as well as the conversational contexts of occurrence. The code results were also quantified to identify the distribution of codes throughout the sample. I recorded my observations about the presence and absence of patterns in each group, as well as the isolation of patterns in particular groups and speakers. In the example of chapter 3, again,

committee's responses to them – were an invaluable starting point for organizing analysis and writing plans. After reviewing the collection of monthly reports, and as I was completing transcription, I read my field notes and the completed transcripts. Using the notes I took during this review of the data, I selected the eight groups that appear in the analyses for chapters 2 through 5.

⁹ These six steps were derived partly from my proposed plan of work and partly from the analysis process. ¹⁰ AtlasTi is a computer program designed for qualitative and textual analysis. For more information, see www.atlasti.com.

this stage identified the very different interpretive environments between particular groups due to their reliance on different interpretive styles.

- 4) The open coding results were matched against the original set of inductive questions. My goal was to check whether the initial observations for the chapter topics were confirmed by the coding procedure. In cases where the question was not addressed, new questions were derived. This allowed me to reflect on why a question seemed appropriate during fieldwork, but was not important during analysis. In the case of chapter 3, I had a number of questions regarding participants' use of different interpretive styles when interacting with different genres in the Bible. This line of questioning proved uninformative for analysis.
- 5) A final reading of the transcripts was done to confirm the revised list of questions for each topic. In addition to confirming the relevant themes and patterns, exceptions and exemplary cases were identified. In the chapters to follow, the examples I use are primarily from these listings of clear exemplars. At this stage in the analysis process, the transcripts reflected each of the previous readings. The use of different color notations distinguished which codes and comments occurred at what point in the process. One benefit of working with the same copy is that it allows unmarked sections in the transcripts to be reconsidered. Large sections of talk had not been included in any of the analyses, and this provided a chance to consider why that was.
- 6) The final stage of analysis was also the first stage of the writing process. A report based on the first five steps was created to summarize the relevant questions, the patterns around individual themes, potential exceptions, and the location of the exemplary cases. This was the first step in constructing a writing outline for each chapter.

The Dissertation's Structure

This introductory chapter has presented the dissertation's topic, the theoretical frameworks that have shaped its development, and the methodological framework in which it was conducted. Chapter 1 is primarily descriptive, providing information about the six congregations, focusing on what currently defines the life of the congregation and how the small group program is structured. Ethnographic portraits of the eight groups used for the remaining analyses are then given. This chapter also summarizes the interactive features of group discourse that characterized the entire sample: interactive frames, participation structure, text selection, and Bible use. This chapter provides the ethnographic backdrop for reading the remaining chapters, particularly my use of discourse examples from the project groups.

Chapter 2 explains this dissertation's main analytic theme – the interaction between the textual ideology surrounding the Bible and the discursive uses of the Bible in small groups. I begin this chapter with an introduction to the concept of textual ideology, which I define as the constellation of ideas, beliefs, assumptions, and expectations readers bring to particular texts. The main goal of this chapter is to outline the specific principles that form the core of the textual ideology surrounding the Bible for the project groups. In total, I identify six organizing principles that deal with the authority, relevance, and textuality of the Bible. This chapter provides the analytical backdrop for interpreting my claims in chapters 2 through 5 about the relationship between discursive practices and this textual ideology.

Chapter 3 deals with the issue of Bible interpretation. My analysis of group discourse revealed three interpretive styles used by participants: finding application,

establishing meaning, and explaining Biblical passages through textual inferencing. Each of these styles embeds principles of the textual ideology, as does its use in group discourse. The common practices and functions in how participants invoke these styles are sufficient to consider small groups as interpretive communities. This chapter highlights the importance of Bible interpretation to small group life, as well as its central role in reproducing textual ideology principles.

Chapter 4 addresses the issue of how interpretive resources are drawn on in group discourse. I analyze textual and cultural resources that are shared throughout the entire sample, resources that receive greater emphasis in particular groups, and resources invoked only by specific individuals. Participants' uses of these resources contribute to the notion of small groups as interpretive communities, and further demonstrate how textual ideology helps structure group discourse. This chapter helps connect what occurs in Bible study discourse to the lives of participants outside the group setting.

Chapter 5 analyzes the practice of recontextualizing Biblical texts. My argument in this chapter is twofold: the prevailing patterns of recontextualization are impacted by the nature of group discourse, and the nature of group discourse is impacted by acts of recontextualization. For the former, I argue that the predominant recontextualization style of a group reflects the predominant interpretive concerns of that group. For the latter, I argue that recontextualizing Biblical texts contributes to the articulation of textual ideology principles, larger discourses operative in American Protestantism, and the guiding interactive norms of small group life. This chapter indexes the significance of intertextual strategies for the study of speech events.

Chapter 6 reviews the issues addressed in this dissertation, and argues for further research on the relationship between textual ideology and textual practice. This chapter also describes the relevance of this dissertation for those not interested in anthropologies of discourse or Christianity.

Anthropologists often use particular events to show how macro-level meanings, social relations, identities, and historical processes are evident in the practices of social actors. Indeed, this is a hallmark of the discipline. Analysis of these events shows how individuals and groups enact as well as reflect upon cultural categories. In this dissertation I devote in-depth attention to the practice of small group Bible study because it is a key event in the culture of contemporary American Protestantism. It provides participants an opportunity to engage in dialogue surrounding what it means to be a Christian, and the variety of beliefs and practices encompassed by this identity. At the heart of small group life is the ongoing relationship between believers and their sacred text, the Bible. My aim in the chapters that follow is to articulate the nature of this relationship, and its role in Bible study discourse.

Chapter 1: Small Groups as Cultural Practice

The previous chapter outlined the cultural significance of small group Bible study, and the need for an in-depth ethnographic perspective on the practice. It highlighted the sheer number of small groups in America, as well as their uniqueness as sites for dialogue and the production of knowledge. This chapter is more descriptive. It begins by introducing the six churches from which the sample of small groups was drawn. Next, I describe the project groups, and provide a sense of what makes these groups attractive to members. Lastly, I illustrate how small group Bible study works by exploring the themes that emerged from my fieldwork experience. This includes the prevailing interactive frames (Tannen and Wallet 1979), participation structure, text selection practices, and use of the Bible. Through these analyses, I consider the following questions: why is the small group experience so meaningful? Why do participants find time to meet every week in spite of hectic schedules, foul weather, competing appointments, sick children, and a host of other reasons *not* to meet?

The Project Sample: a congregational perspective

During the 12 months of data collection, I acted as a participant observer in 19 small groups. They were distributed among six congregations from four Protestant denominations. A sketch of these local churches provides a useful and necessary introduction to the individual groups. The description of each church is based on the issues that occupied its attention during my year of research. Each church placed particular emphasis on the small group experience. To capture this, I address the structure of the small group program in each church, and the role of small groups within the life of the congregation. Table 1 presents some basic information about the project churches.

	Reported Membership/ Avg. Sun. Worship	Year Est.	~ No. of Small Groups ¹¹
Suburban UMC	700+/400+	1866	19
Downtown UMC	400+/200+	1851	9
Inner-City UMC	250+/150+	1912	11
LC-MS	700+/350+	1984	22
Restoration	600+/450+	1952	27
Vineyard	200+/100+	1986	6

Table 1: Data for Six Project Congregations

Suburban United Methodist (SUMC): This congregation is located in a previously rural-farm community that began transforming into a residential suburb in the early 1990s. A drive in and around the town limits features a barrage of new and half-finished housing developments. The prevailing demographic is the young married couple with young children. Pastor Rod has made a continual effort to tailor the church to these surroundings. As a result, the church has become a local success story for numerical growth and attracting young families.

The church has been a part of the community for 140 years. When Pastor Rod arrived in 1988 the membership had plummeted to 40, and his assignment was to assess whether the church should permanently close. The changing demographic struck Rod immediately, and he began making adjustments to meet the community's changing needs. The building of two additions – one wing for an expanded children's Sunday school program and one for a nursery – were milestones in this transformation. Currently, membership exceeds 700 and is comprised of working to upper-middle class, white families and retirees. Most live within a five-mile radius of the church building, though a small minority resides in the city.

¹¹ These totals reflect the number of small groups active in each congregation during the fieldwork year. They are not necessarily the number of groups offered when I began the research or when I concluded the research.

Congregational life at SUMC revolves around its mission of being "familyoriented." In addition to the large nursery and Sunday school wings, a game room and
gymnasium have recently been added. Separate leadership positions have been created
for a children's and a student minister, whereas most churches combine these roles into
one. The student ministry is popular enough to require separate junior high and high
school programs. Special events bringing parents and children together are continually
offered (e.g., story hours for younger children, parent-child sports for teenagers). Young
married couples – with and without children – are encouraged to attend marriage
seminars offered by the church several weekends during the year. These are organized
around a professional speaker who conducts the entire weekend. This dynamic of
catering to young families and experiencing growth has been a common trend among
suburban (primarily Evangelical) churches over the past two decades (cf., Ammerman
1997, Eiesland 2000, Elisha 2004).

The current small group program at the church focuses on a set of ~15 home groups. This began in 2002 when SUMC participated in the popular church campaign "40 Days of Purpose¹²." A central component of the Purpose Driven campaign is the formation of a small group network, where every group reads and discusses the same set of materials each week. Following the six-week program, these groups of six to ten members continued meeting, alternating between studies of their own choice and ones chosen by the small group director and pastor.

¹² This program, and its follow-up "40 Days of Community," result from the best-selling <u>The Purpose Driven Life</u> (2003) by Evangelical pastor and author Rick Warren. "40 Days of Purpose" and "40 Days of Community" have been widely popular among Protestant (and even some Catholic) congregations. For a scholarly review of this movement see (Balmer, 2006).

Part of any newcomer's socialization into SUMC is an attempt to place them in the home group that "best meets their spiritual and practical needs." These groups meet at various times of day on various days, some every week and some every other week. They are divided by demographic differences. These classifications include young couples, couples with children, couples with young children, couples with older children living at home, women, elderly adults, and middle-aged empty nesters. These groups have become increasingly important in the life of the congregation, coordinating on outreach and fundraising efforts. Additionally, there are four other groups connected with the church, all of which were well established when the home groups began: a women's Bible study, two men's Bible studies, and a Sunday school class.

Downtown United Methodist (DUMC): This church has been a fixture in the city's downtown for 155 years. The state capital is directly adjacent to the south corner of the building, and Capitol Avenue lies just a few feet from the main entrance. Although the church's physical presence has been constant, the climate of the congregation has been in constant flux. Most everything that characterized DUMC a few decades ago is now being contested, or is already on the wane. In the following I provide a few examples of these changes.

In the mid-1970s DUMC was considered among the city's most affluent churches. Now, the membership includes individuals and families from every socio-economic stratum. DUMC is also the most racially mixed church in the sample, with a significant portion of its membership comprised of African-Americans. As part of the socio-economic change, the church has foregrounded its identity of being open to the downtown community via its "Open Door Ministry." This program keeps the building

open 24 hours a day, and its mission is to "provide sustenance, counseling, assistance and love to those of God's children who find themselves in difficult circumstances and with few places to which they can turn." This mission effort is mainly a product of the church's location in downtown, which is surrounded by several neighborhoods characterized by poverty-level income, crime, and fluid residency.

The brick and stone architecture of the building has become as much a source of consternation as a source of aesthetic interest. Age and deterioration have prompted a "Restoration Task Force," which estimates the total cost of revamping the building to be near a half million dollars. However, the primary source of aesthetic pride remains intact — a series of shining organ pipes that run the length of the pulpit and tower 30 feet against the back wall. The massive organ system has also been symbolic, signifying the identity of the church as a home for traditional, "high" hymnal music. This, too, is changing. Sunday morning worship services now alternate between centuries old hymns that ring from the organ and contemporary praise music that rarely incorporates the instrument.

The prevailing theological attitude has been the congregation's most contested feature in recent years. Like many United Methodist churches across the United States, DUMC experienced significant theological and political changes during the civil rights movement of the 1960s. Keeping with national trends, the last decade has been one of increasing Evangelicalism and heightened internal oppositions, mirroring the struggles at regional and district levels (Kirby, et al. 1998). The theological clashes at DUMC have taken place on numerous fronts, including homosexuality and the Iraq War. Most recently, the use of feminine and gender-neutral references to God during worship services sparked public and private reactions of support and dissent. Leaders and laity

recognize these divisions as a source of some ambivalence, generating anxiety over tension and pride in diversity.

The small group setting, because of its value as a potential site for open dialogue, is an exemplary place to observe the debates that surround these congregational changes. DUMC has not adopted a network of home groups, or relied on church-wide curricula. In the past DUMC has offered "Disciple" courses, a series of four 32-week Bible studies produced by the United Methodist Church¹³. These were considered moderately successful by members, and have not been offered in recent years. Instead, small group life at the church consists of several independent groups, as well as an established adult Sunday school program.

At any given time, between four and six adult classes are offered on Sunday morning. These are divided by topic as opposed to age or gender, and usually last for 6-8 weeks. Recent subjects include Bible studies of Proverbs and Ecclesiastes and a video study by Evangelical apologist Ravi Zacharias. There are three ongoing groups: a men's Bible study, a women's reading group, and a Sunday evening class. Ad hoc groups also arise in accordance with seasons in the religious calendar, or an individual's interest in a particular topic. During the Lenten season of 2005, for example, a class was co-taught on a recent book by a United Methodist pastor, Journey to the Center of the Faith (2001). Leadership and participation in these groups vary, and they too last between six and eight weeks.

Inner-City United Methodist (ICUMC): ICUMC began as a sister church of DUMC in 1912. It is located at two high-traffic cross-streets, several miles from downtown, in the midst of several of the city's worst neighborhoods for crime and

¹³ For an ethnographic description of a group using Disciple materials see Olson (1994).

poverty. The last six years have absorbed the social memory of the congregation, with five different pastors being appointed since 1999. The arrival in 2000 of the church's first female, first African-American, and most controversy-minded pastor prompted a series of schisms. Conflicts arose about the church's role in the local community, as well as sociopolitical issues not previously of explicit concern to the church, including: racial tensions, same-sex relations, and right-to-life issues. Much of the membership left for more stable church environments. A refrain I heard from a variety of Bible study participants was that this pastor's brief tenure "brought the church to its knees." When Pastor Ted arrived in 2002, permanently closing the church was still an option.

Despite the unenviable circumstances, Ted's enthusiasm has led to resurgence in the church. The membership is above 300, and the worshiping community on Sunday mornings is approaching 200. The socio-economic composition of the church is primarily lower to middle-middle class, though there are several very wealthy families who have been active in the church across several generations. Much like DUMC, ICUMC is a multi-racial church, including a substantial population of foreign refugees (see below).

As Ted's tenure matures, two changes are shaping the life of the congregation. In 2003, the arrival of the first foreign refugee foreshadowed a turning point in the church's identity. The city has a substantial refugee population, mostly from Middle Eastern (e.g., Iran and Afghanistan) and African countries (e.g., Ivory Coast, Sudan). Some are Christian and in search of a place to worship. Many others are Muslim and are attracted to the hospitality and material assistance offered by the congregation. The active evangelism of the first wave of refugees has led to a group of 30 adults and children who now attend worship services at ICUMC regularly. As the number of refugees has grown

over the past three years, their relationship with the congregation has evolved. Language barriers, cultural differences, and theological misunderstandings have spurred a relationship of mutual frustration, trust, and appreciation. The presence of the refugees has also prompted the church to be more active in the surrounding neighborhoods, where many refugees have been placed to live. (See chapter 4 for additional information about the relationship between ICUMC and the refugee population).

The other change in the life of the church began midway through the research year. Following his arrival, Ted instituted a second, contemporary worship service on Sunday mornings. He hoped this would attract more teenagers and young adults. The contemporary service experienced only mild success and in the summer of 2005, it was brought to a temporary end. ICUMC then began consulting with the pastor of the fastest growing UMC congregation in the city. Weekly planning sessions were devoted to the launch of a new contemporary service, as well as a new "culture of outreach," for the fall of 2006¹⁴. Ted's thinking in this regard has been structured greatly by his experience with the "emerging church movement." This particular segment of the church growth institutional network emphasizes a "return to first century Christianity" using the "technologies and methods relevant in today's culture" (see chapter 5 for further explanation). The "new launch," as it is known around the church, has received widespread support. In addition to the open planning sessions, Ted has given updates before sermons and made repeat visits to Sunday school classes to discuss potential changes.

One of the first tasks Ted assigned himself was to increase the small group activity at the church. Historically, the church has emphasized its Sunday school

¹⁴ As of the mid-summer of 2006, ICUMC had moved the launch of the new service to January of 2007.

program, which Ted has sustained. Five classes are offered: two for elderly adults, one for twenty and thirty-somethings, one for middle-aged members, and one for the refugees. All but the refugees' class have been meeting together for numerous years.

Newcomers to the church are encouraged to join one of these classes, which invariably correspond with their age. The significance of the Sunday school community highlights the importance of Ted using that time to advocate for the "new launch."

Ted has also introduced the use of several structured small group programs. This began with the UMC Disciple classes. Unlike DUMC, ICUMC has responded well to these lengthy, intensive Bible study courses. Three of the four have been taught at the church, and every fall at least one of them is started anew. Ted has also organized other programs, including ALPHA¹⁵, Becoming a Contagious Christian¹⁶, and both Purpose Driven campaigns. These efforts have been consistently well received by the congregation, but have not been successful in establishing an ongoing network of home groups. Groups that form for these programs disband once the curriculum ends. In fact, there is also one ongoing group at the church, a men's Bible study that meets weekly on Tuesday mornings.

Lutheran-Church Missouri Synod (LCMS): All of the project churches can be described as "healthy," in that their membership is experiencing growth rather than decline. More than the others, though, LCMS had the feel of excitement and enthusiasm at every event I attended. They have experienced continual growth in membership and worship attendance over the past three years. The arrival of a new pastor in 2003

-

¹⁵ ALPHA is a worldwide program. It began in the United Kingdom within the charismatic tradition. It is designed as an introduction to the "basics of the Christian faith," and is intended as an evangelizing tool for churches. Members are encouraged to bring a non-believer with them to the weekly video talks.

¹⁶ This program is produced by the Willow Creek institutional network and is co-authored by the founder of Willow Creek, Bill Hybels. It is an eight-session course designed to cultivate evangelism skills.

accounts for a great deal of this fervor. He is relatively young (mid 40s) and is extremely well liked throughout the congregation. The recent growth has led to the planning of a new building, and groundbreaking scheduled for the summer of 2006¹⁷. Their success is an anomaly within the national context of the denomination. LCMS churches across the U.S., including the Midwest where they are most numerous, have been declining steadily for decades (Jones, et al. 2002).

The church is located in a suburban community east of the city. Its membership, like SUMC, derives primarily from a five-mile radius around the church. However, the struggles of LCMS churches in other parts of the city have attracted new members from neighborhoods further away. The membership is primarily white collar, but there is some representation of the working class. A wide spectrum of ages is present in the life of the church, including a burgeoning population of young families. LCMS was the most ethnically homogenous, with nearly the entire membership being of Western European descent (many from German and Scandinavian Lutheran heritages).

The Lutheran Church in the United States consists of three major Synods:

Missouri (LCMS), Wisconsin (WELS), and the Evangelical Lutheran Church of America (ELCA). Their institutional separation is a result of doctrinal and theological schisms during the past half century (Noll 1992). The ELCA Synod is the largest of the three, a member of the National Council of Churches¹⁸, and is typically understood as the most liberal of the three Synods. The Wisconsin Synod is the smallest and is generally viewed as the least flexible on doctrinal points of orthodoxy. The Missouri Synod occupies the

¹⁷ LCMS continued as planned with the groundbreaking schedule. As I write this in the fall of 2006, excavation of the site is underway.

¹⁸ The National Council of Churches is an institutional body whose membership is recognized as the "mainline" churches in the United States.

middle ground between these two. The pastorate is drawn from two seminaries: one in Ft. Wayne, Indiana, and one in St. Louis, Missouri. These are recognized within the denomination as, respectively, "traditional" and "moderate" on doctrinal and sociopolitical issues (e.g., the proper role of women in church life). As a result, there is variety among Missouri Synod churches; though all remain firmly a part of the conservative Christian sub-culture in America. The project LCMS congregation offers a representative cross-section of this national context.

The structure of small groups in the congregation is diverse. Structured programs are not as likely here as with SUMC and ICUMC, but they have participated in both of the Purpose Driven campaigns. There is a network of home groups that were established soon after Pastor Daniel began his tenure in 2003. The number of these fluctuates between 10 and 15, as leadership responsibilities shift and time commitments change. These groups meet for four to six week sessions throughout the year, broken up by sabbaticals of several weeks to a month. Groups are encouraged, and uniformly agree, to use study materials designed by either the pastor or a senior member in the congregation. These studies are Biblically based, and use a series of handouts that consist of Bible readings, followed by commentary and questions from the study's author. In the spring of 2005, for instance, Pastor Daniel wrote a four-week study of prayer based on four Old Testament examples. The pastor also coordinates his Sunday sermons with the small group study topic. Groups are no bigger than 10-12 members, and are organized by personal preference. As people join or express a desire to get involved with the church, they are encouraged to find a group that suits their "schedule and comfort level."

There are also five ongoing groups. Three are women's Bible studies: one meets on a weekday evening, one for young mothers on Friday mornings, and one for older adults and seniors on Thursday mornings. A men's Bible study led by the pastor meets on Thursday mornings at a local restaurant. And, there is a class for new members that meets on Thursday evenings. Leadership for this class alternates between the pastor and an experienced teacher of Lutheran doctrine. The class is designed as an overview of LCMS theology for "new members," but it has attracted existing members who did not grow up in the Lutheran tradition as well as lifelong Lutherans interested in a "refresher course." All of these groups are well attended, and recognized within the congregation as being quite popular. The remaining small group program consists of two Sunday school classes offered between the two worship services. Pastor Daniel leads one class, which attracts several dozen participants. The other class is led by a church elder, and has roughly a dozen participants. Both work through eight to ten week topical studies, often based on a Biblical text.

Restoration Movement (RMC): Theologically, RMC is a typical Evangelical congregation. The Restoration Movement identifies itself through idioms like "Bible-based" and "New Testament Christianity." RMC advertises itself in similar ways, emphasizing a Triune and personalized God, Jesus Christ as individual and collective "Lord and Savior," the Second Coming of Christ, an active Holy Spirit, an "infallible" and "inspired" Bible, and the necessity of baptism by immersion. In the United States there are over 6,000 Restoration churches with more than 4 million members, as well as an affiliated network of Bible colleges (Jones, et al. 2002). In the research city, there are six Restoration churches and one of the denomination's colleges. RMC is the largest in

the area, with over 600 members. The church moved into its current facility in 2003, after its old building proved inadequate for its growing worship service attendance.

The life of the congregation thrives on a network of "ministries." These ministries cover a wide range of spiritual and practical functions, from building decorations to prayer to discipleship. Several are devoted to developing the Christian education of children and young adults in the church. Some are based on nationwide programs, such as the Women's First Place ministry, which connects bodily and spiritual health through diet, exercise, prayer, and Bible study. Some ministries plan congregational events throughout the year. The prayer ministry organizes several 24-hour prayer campaigns where the church's prayer room has at least one person praying from midnight to midnight. Along with small group opportunities, these ministries are a principal means for members and newcomers to become active in the life of the congregation.

RMC offers a wide variety of small groups. The Sunday school program is more robust than that of any of the other project churches. At any given time, there are between eight and ten classes offered on Sunday morning prior to the worship service. Some are divided and identified by age (e.g., "seniors", "college age"). Others are divided by subject matter, including book and Bible studies. These classes are well attended, attracting roughly one third of those attending worship service.

There are several ongoing classes during the week. Two of these are women's Bible study programs. The first meets at the church on Wednesday afternoons, and consists primarily of older adults. The second is a series of groups who meet as part of the First Place Ministry. There is a four-part series called "Building U" designed by the pastor and other church leaders. "Christianity 101, 201, 301, and 401" are intended as an

introduction to the fundamentals of the Christian faith. Past topics have included "understanding the Bible," "the Book of Genesis," "the Holy Spirit," and "Tales of Truth from Jesus." Every fall the classes start over again, with each section lasting several weeks. In the spring of the following year, anyone who attends all four sections receives a "diploma of completion." All members of the church are encouraged to "graduate" at least once.

The home group program at RMC is also robust, though not as well organized as at the other project churches. The small group director also serves as the assistant pastor, youth minister, and worship leader. This load of responsibilities made keeping an up-to-date list of small groups a more difficult task than usual. The number of groups and list of facilitators shifted often. At any given time, between 10 and 16 home groups were meeting. Groups vacillated between study topics of their own choice and congregation-wide studies. The latter included structured programs, like the Purpose Driven campaigns, and Bible studies designed by the pastor. In 2005, for example, groups chose their own topics for the first half of the year. In July, they were asked to complete a sixweek study of 1 Peter written by the pastor. Following two months on their own schedule, groups were asked in late October to participate in the second Purpose Driven campaign.

<u>Vineyard Fellowship (VFC)</u>: In several ways, VFC was unique in the research sample. It was the smallest in membership number and in average worship attendance. It was the only church where the current pastor is also the founding pastor. It was the only place on Sunday morning where nearly everyone remained after service to socialize.

Theologically, it is the only church that institutionally recognizes the continuing

availability of charismatic gifts. But, like the other project churches, VFC has also experienced growth in recent years.

When the church began in 1986, it met on the nearby university campus. Eight years later it moved into its current building – a two-story warehouse structure designed to look like an "old red barn." In 2004, the church purchased the entire building, which it is now renovating. VFC is a good example of the "new paradigm" churches described by Donald Miller (1997), whose research included a Vineyard church in southern California. Most of the defining features observed by Miller are evident at VFC: contemporary worship, informal and self-revealing preaching style, casual dress code, bodily expression in worship, acceptance of Biblical authority, and an emphasis on small group participation (ibid: 20). VFC is also a typical Vineyard congregation within the American context. The pastor of the church was mentored by a co-founder of the movement, John Wimber. And, in terms of theology, mission focus, and institutional organization, VFC is close to the Vineyard model articulated by Wimber in the early 1980s (cf., Jackson 1999).

The small group structure at VFC was more limited than the other churches because of its smaller size, but it involved a higher percentage of the overall church membership (nearly 2/3). VFC does not offer Sunday morning classes, and there was only one ongoing (women's) group during the week. VFC has used three structured programs: <u>ALPHA</u>, which they offer once a year, and both Purpose Driven campaigns. The primary small group opportunity came via home group meetings. There were five of these groups; though, one met at a local coffee shop instead of the facilitator's home. This emphasis on home groups is also consistent within the Vineyard tradition, whose

"church planting" model originates with home group meetings (Miller 1997). VFC home groups were unique because of their meeting structure and the consistency of different groups in replicating the format.

The VFC home groups began with a period of worship singing that lasted 25-35 minutes. In cases where an able musician was present, (s)he played guitar while others followed along on a printed lyric sheet. In other groups, a DVD of praise songs produced by the Vineyard denomination was used. Following the worship, the worship leader or facilitator led a brief prayer. A 30-60 minute study period came next. The study topic shifted from week to week, rather than lasting for multiple sessions. The role of the facilitator also shifted; another characteristic feature of Vineyard small groups. The intent behind rotating leadership is to cultivate a sense of ownership and involvement in every participant, as opposed to limiting the responsibility and benefit of leading to one person. Topics included Bible study lessons, personal testimonies, discussion of a Vineyard or charismatic periodical article (e.g., Charisma, ElijahList.com), viewing videotapes and listening to audiotapes by Christian speakers. Meetings were concluded by a 20-30 period of prayer requests and group prayer. This always held the potential to be an emotional experience for members. Prayer requests could be very intimate, and often dealt with serious personal and family struggles. Individuals shared these concerns openly, and they were addressed in specific detail during the actual prayer.

For the remainder of this dissertation, I do not draw on the data collected from the Vineyard small groups. My reasoning is threefold. First, given the structure of their weekly meetings, relatively little time was devoted to study discussion. Secondly, all of the home groups met on the same night of the week, which meant my data for all of them

is discontinuous. In hindsight, it would have been more beneficial to choose one of the groups and attend only its meetings. Lastly, for the arguments I make in this dissertation, the data I do have from the Vineyard is consistent with the conclusions I draw based on the other project groups.

Similarities in Small Group Life

This review of the project churches reveals some similar approaches to providing small group opportunities. In each of these churches small groups are emphasized as an essential component in the health of the individual Christian life and the local congregation. Small groups are advertised as places for developing "fellowship" with other believers, growing in one's "relationship with Christ," organizing congregational and community activities, and becoming "stronger in God's Word." Though the number, format, and organization of small groups varied, the high value placed on the small group experience did not.

Based on the above descriptions, small groups can be organized into four overlapping types: Sunday school classes, home groups, ongoing classes and groups, and groups linked with structured programs. This latter category is particularly interesting. All six churches during the research year, or in the recent past, participated in a congregation-wide study curriculum. The most common were Rick Warren's tandem campaigns. In any case – be it a Purpose Driven campaign, ALPHA, Disciple, or a locally designed study – they rely on the formation of a small group network. Part of these programs' attraction is their potential for establishing small groups that will continue meeting after the program has concluded. As evident from these churches, though, this promise is not always fulfilled.

The intentions underlying these four types also have reappearing themes. Several churches offered groups that address the "basics" of the Christian faith and/or the affiliated denomination. The audience in these cases was not only newcomers to the church. Congregations also encouraged longstanding members to participate in these studies. Developing methods and skills for evangelism was another common goal. In most cases, the group itself was viewed as an opportunity for evangelism. Members were continually prompted to invite unchurched and non-believing friends and family to the group; or, at the very least, to solicit prayer requests from them¹⁹. Lastly, all six small group programs persistently stressed Bible study. Some groups read directly from a Biblical text, while others consulted a variety of texts to develop a Biblical theme. In any case, strengthening its knowledge of, and relation with, the Bible was a principal intent for every group. And, as the following chapters make clear, how groups interact with the Bible is the key theme explored in this dissertation.

The Project Sample: eight ethnographic portraits

The six project churches provided ample opportunities for conducting ethnographic research. I visited 25 groups during the sampling process in the fall of 2004, and I collected data from 19 groups. Table 2 summarizes how these were distributed amongst the churches.

	Groups Attended	Sun. Schools	Home-Based	Ongoing	Ad Hoc
Suburban UMC	5	1	2	2	-
Downtown UMC	3	-	-	2	1
Inner-City UMC	3	1	-	1	1
LC-MS	3	-	1	2	-
Restoration	3	1	2	-	-
Vineyard	2	-	2	-	-

Table 2: Small groups attended in six project congregations

_

¹⁹ Group members frequently reported encounters where they invited new people to the group, but there were only a handful of occasions when people from outside the congregation attended.

My tenure in these groups ranged from one meeting to 48 meetings. Clearly, this produces an imbalance in the corpus of data, and in turn, my understanding of the 19 groups. This requires a methodological decision: on which groups do I concentrate for the discourse-centered analyses? The final third of this chapter draws on my experience with the entire sample, but chapters two, three, four, and five rely on those groups from whom I collected the most consistent and reliable data²⁰. Tables 3 and 4 identify some basic information about these eight groups; followed by an ethnographic introduction to each.

Group Name	Congregation	Meeting Time	Frequency	# Meetings	Year Est.
"Iconoclasts"	Suburban UMC	Sun, 6-8pm	Weekly	31	2002
McGee Men	Downtown UMC	Tues, 9-10am	Bi-weekly	18	1985
Tuesday Men	Inner-City UMC	Tues, 7:30- 8:30am	Weekly	29	1992
LCMS Women	LC-MS	Thurs, 7-8am	Weekly	34	1990
LCMS Men	LC-MS	Thurs, 10- 11:30am	Weekly	48	1994
Prayer Circle	Restoration	Thurs, 7-9pm	Bi-weekly	14	2003
"Mind of Christ"	Restoration	Sun, 9-10am	Weekly	11	2005
Monday Group	Restoration	Mon, 7-8pm	Weekly	8	2003

Table 3: Data for eight groups used for discourse analysis (1)

	Total Members	<u>Regular</u> Members ²¹	Male/Female (Regulars)	Ages (Regulars)
"Iconoclasts"	14	11	5/6	31-65
McGee Men	14	7	7/0	55-70
Tuesday Men	8	7	7/0	49-75
LCMS Women	28	10	0/10	55-79
LCMS Men	43	27	27/0	40-80

²⁰ I used several criteria to make this decision. 1) Only groups that were tape-recorded were included. 2) Only groups where I observed at least one complete study were included. 3) Only groups where discussions remained germane to the study text or topic were included (i.e., groups that met mainly for fellowship were not included).

²¹ This designates the number of participants who attended at least half of the observed meetings for the group.

Prayer Circle	14	9	3/6	28-49
"Mind of	18	6	2/4	18-51
Christ"				
Monday	6	3	2/1	49-53
Group				

Table 4: Data for eight groups used for discourse analysis (2)

Suburban United Methodist Church

"Iconoclasts": Once a month the small group director at SUMC, Janet, held a "huddle" following Sunday worship service. Her goal was to gather the facilitators so they could exchange updates on the progress of their groups, and ideas on how to improve. As a researcher, it provided a convenient way to monitor what each group was doing. More important, it was an opportunity to hear reflective comments from the facilitators of the SUMC project groups. I was especially interested in what Charlie would say. He was among the most outspoken and energetic facilitators I had encountered. I realized early in the fieldwork with this group that Charlie spoke his mind freely. And, from our conversations before and after meetings, I knew he was perfectly willing to evaluate the other participants. Most of the facilitators at the huddle were reluctant, or unable, to offer much about the dynamics of their group. However, Charlie's first sentences met Janet's inquiries about the group's progress head on:

"Really, the group is a collection of pseudo-rebels and iconoclasts. They are extremely studious, and intense. They don't want anything to do with popular culture, and they love to nibble on controversial topics. Right now we're reading Philip Yancey's The Jesus I Never Knew, which we love. We did the Ortberg and Warren studies before that, which none of us could stand."

The "Ortberg and Warren studies" Charlie refers to were both congregation-wide programs selected by Janet and the pastor. This type of straightforward comment is vintage Charlie. Despite the Iconoclasts' aversion to her taste in study materials, Janet was quick to recommend them during our first meeting together in the early fall of 2004.

She suspected they would be eager participants in the research, which they were. I remember my first explanation of the project and the absence of any skeptical or tentative eyes. Their initial reaction, in fact, was to be "the best small group in the project." They laughed and warned me of their competitive streak. As usual, Charlie found a way to synthesize everyone's comments: "If all the other groups [at SUMC] bring in fifty cans of beans [to the food drive], we'll bring a hundred." They put me at ease immediately.

The Iconoclasts are a collection of early-30s to early-50s married couples, as well as two single women, one in her 30s and one in her 60s. The group meets in Charlie and Diane's living room: an open space, always immaculately clean, with tall ceilings and neatly arranged furniture. The seats for group were arranged in a circle, with a sofa, armchairs, and dining chairs. The group was familiar with one another when I began my work. They had been meeting since 2002, and half of them had been in a Sunday school class taught by Charlie in prior years.

Lively discussions were part of every meeting. Charlie's leadership style combined with the eager participation of several members to make long silences rare. Brief bouts of intellectual sparring, often around controversial topics, were reliable as well. During the course of reading The Jesus I Never Knew, the group devoted extended discussion to historical and textual proofs for the veracity of the Bible, the problem of theodicy regarding the Asian tsunami, the ideals of purity and holiness, historically condemnable Biblical justifications, the New Testament attitude towards women, the delicate balance of faith and good works, and the relation between science and religious faith.

My role in the group was primarily that of a quiet observer, but I was never forgotten. In the first several meetings Charlie addressed questions directly to me, usually after others had given a response. This happened progressively less throughout the fieldwork, but never disappeared. I was quickly given the moniker of "resident anthropologist," and the accompanying line that "every group needs one." I never felt as though responses were restrained or offered on my account alone. At no point was I asked to pause the tape recorder, and the participation order remained constant throughout the fieldwork. They were in favor of the research, often commenting that it was a "good thing" and that they "look forward to what [1] come up [with]." In short, the group did its best to downplay my presence while still making me feel welcome. Their hospitable attitude was most evident in their willingness to include me in their banter. One evening, during a lighter moment in a discussion about "making it to heaven," one of the men quipped: "Even heaven needs an anthropologist."

Downtown United Methodist Church

McGee Men: This men's Bible study is representative of the discord within DUMC. In particular, two sources of division characterized the group. The first is the congregation's theological divide. Of the seven regular participants, four were decidedly liberal in theology, while the other three were much more conservative. The authority of the Bible on all matters spiritual, moral and practical was the premier ground on which they confronted this difference. Discussion transitioned into debate regarding the precision of Biblical language, the accuracy of Biblical analogies, and the contemporary relevance of verses. A second tension revolved around the purpose of meeting as a small group. Several members gravitated toward an analytical approach to group study. Others

believed the emphasis should be on their personal situations, applying Biblical principles to their daily lives, and sharing testimonies about their struggles.

Both approaches were recognized by the group as reasons why some members stopped coming, as well as why the group remains lively and enjoyable. Maintaining the balance between analysis and intimacy was a continued source of disquiet, frustrating several regular members that people are put off either way. Their decision to change study materials indexes this tension. The group has been meeting for nearly 20 years, and much of that time they used Bible commentaries by the Scottish theologian William Barclay. Barclay termed himself a "liberal evangelical," perhaps because he maintained opinions that typically do not co-exist. He cast serious doubt on the virgin birth of Jesus, as well as the miraculous nature of his miracles. Yet, he insisted on the actuality of Jesus' resurrection, and that Gospel authorship belonged to the four books' namesakes (Jackson 2003). In his approach, Barclay consistently presented his Bible commentary in a scholarly manner, tightly argued and well-documented. Just prior to my arrival, the group agreed to a stylistic change. They chose a noted Fundamentalist commentator, J. Vernon McGee, to study the book of James. McGee's "Thru the Bible" series is based on his 1967-1972 radio addresses, and their tone is conversational and informal. The switch of study material did not alleviate the tension. For reasons stylistic and theological, the book attracted some and repelled others.

The use of Barclay and McGee, both older resources, reflects the age of the group. The pastor is the youngest participant, midway through his 50s, and the only member still active in his career. The other men are in their 60s and early 70s. They are very well educated, with several retired teachers, a Methodist pastor, an engineer, a

surgeon, and a journalist. Many are life-long Methodists, and all have a lengthy history in the congregation, making them very familiar with one another. The pastor was not the facilitator, though he was a regular member. This role belonged to one of the lifelong members in the church.

Unlike the Iconoclasts, I was never directly invited into discussion. They were accustomed to academic endeavors, and were more than happy to treat me as a silent observer. From all I could tell, the nature of their conversations did not change as my time with them progressed. Several of the men were overtly pleased with my being there. The pastor was enthusiastic about the research, and the others enjoyed having someone younger at the table. Unfortunately, my presence proved difficult for one of the regular members. The retired Methodist pastor informed the other participants several months into my research that he was uncomfortable being taped and would not return until my work had concluded. He did not approach me directly, and the regular participants never brought it to my attention. My knowledge of his decision came only from a later e-mail exchange with another group member.

Inner-City United Methodist Church

Tuesday Men: I have always been unsure about the difference between consistency and predictability. The latter has such a negative connotation to me, but it is only possible because of the virtues of the former. The Tuesday Men's Bible study draws me back to this musing. By late October 2005, I had been observing them for seven months. Near the end of discussing Acts 7, Pastor Ted read the study question, "What implications is Stephen making about where God can be worshiped in Acts 7:44-50?" As was typical with the group, a lengthy pause ensued as the men reviewed the answers they

had written on their study sheets. During the silence I ran through my mind the type of response to expect from each person. This was my 24th recording and prediction was a tactic for remaining attentive. I was amazed when Ted began fielding responses. The tone and theme of the contributions, not to mention the order of their occurrence, were nearly identical to my forecast.

One conclusion is that there was a resulting staleness to being with this group.

Nothing could be further from my experience. The group began in 1992, including four of the current members. The pastor at the time was experiencing personal difficulties, and started the group as much for support as for study. When Ted arrived in 2002 he took over facilitating. With the addition of two members, the group has remained as is. Every Tuesday at 7:30am, so long as they are not sick or traveling, they gather in the church's second floor classroom, assume their usual seats, and engage the day's reading and questions with measured eagerness.

The Tuesday Men are very similar to the McGee Men in terms of age. The pastor (in his 40s) is the youngest participant, and the rest are between 65 and 75. However, the age difference was more evident here. In particular, Ted's knowledge base in the "emerging church movement" was a challenge to some. The Tuesday Men also took a different approach to studying the Bible. Instead of using a commentary, they used a series of study questions – the "Life Guide" series published by InterVarsity Press. Each sheet covers a chapter of Bible reading, followed by 10-12 questions (cf., Appendix 3). Pastor Ted has been using this series in various settings for 15 years, and the men reacted positively when he introduced it to them. Two to three meetings were usually required to complete each sheet. Ted always passed out the next sheet the week before starting it,

giving the men a chance to complete it ahead of time. Without fail, they showed up the following week with responses, often multiple paragraphs worth, for each question.

My role in this group varied because several participants perceived me in different ways. Pastor Ted sought my insight on questions about "changing the culture of the church." I attributed this as much to my age, being in the demographic the church wants to attract, as my burgeoning status as an "expert" on congregational life. When he did this, I did my best to offer an honest, but brief response in hopes of not redirecting the discussion. Several men treated me with the same kind of welcoming indifference as the McGee men. Others were less comfortable with the silent observer role and wanted me to contribute more to discussion. They were never anxious about it, but eventually did start addressing me directly.

Lutheran-Church Missouri Synod

LCMS Women: I approached seven women's groups about being part of this research. Three denied my request. One agreed, but denied my request to tape record. One agreed to be tape-recorded, but it disbanded due to lack of attendance after one meeting. One met for only a single six-week study during my fieldwork. The LCMS women were a sharp contrast to these other opportunities. I attended my first meeting with them in November of 2004 to explain the project. When I finished, the group facilitator, Sandy, went around the table in the church library and asked each woman if they had any questions or objections. One by one, all smiling, they repeated a willingness to participate: "I think it'll be fun," "Sure, it's good to mix things up," "Sounds interesting," "I don't mind," "If he can stand being around us old ladies every week."

These women are much more than study partners; they are close friends. Many were charter members who helped plant the church almost 20 years ago. Some vacation together; others baby-sit grandchildren together. The first 15 to 20 minutes of every meeting was spent talking and laughing. And, the last five minutes was spent deciding on where to eat lunch that day. Their discussions reflected this familiarity and closeness. They transitioned seamlessly among the intimate, the supportive, the scholarly, the jovial, and the controversial.

The group was unique for its variety of distinctive personalities. Sandy's role as facilitator was unquestioned. The other ladies thought very highly of her in this regard. They appreciated her desire to involve everyone, and her efforts to consult multiple commentaries for explicating the study lessons. Sandy valued her responsibilities and met them with diligence, though she was sometimes anxious about meeting the women's expectations. Many of our conversations before and after group meetings centered on the pros and cons of the current study material, potential study resources, and different styles of leadership.

Joy, a lifelong LCMS member but a recent arrival to this church, contributed in ways that were anticipated for being insightful, articulate, straightforward, and for providing the "standard Lutheran answer." On the few occasions when Sandy needed to be out of town, she relied on Joy to assume the role of facilitator. In late January of 2005, the week prior to Sandy leaving on a two-week mission trip to Tanzania, the following concluded a discussion of spiritual mentoring:

Joy: Jesus didn't whitewash it at all. The rich, young man left after Jesus told him to sell all he had. So, if somebody goes out our door because they don't like hearing about sin, let them go. That's better. Maybe they'll go and think about it. It's better than letting them think that sin is of no consequence and, "Yeah, salvation is a gift and that's the end of it. You can do what you want." Well, you can. But, they should know the work of the

Holy Spirit. That's why we have classes so that people know what the Bible says and that the Holy Spirit sanctifies us and it's gradual, you know, makes us more Christ-like as we study His Word.

Sandy: Aren't you glad she's going to be teaching us next week? Very good. I could have written that down.

Linda was reliable as well, but for a different reason. With a mix of seriousness and humor, her conservative political leanings added a shade of controversy and current interest to the discussions. In response to a study guide question asking for examples of when "God has freed you to leave [a situation] or freed you to stay," Linda was the first to respond:

"This isn't my experience, but I'm thinking of somebody like Dr. Bernard Masonson, who himself did or presided at tens of thousands of abortions. And, then, suddenly found that they were wrong when he came into a relationship with Jesus Christ. Now, you talk about guilt. He has to know that his guilt can be taken away, that it's absolutely covered by the Blood. But, can you imagine, as a human being, knowing you were responsible for tens of thousands of deaths of people? I mean, that would be a pretty tough thing to take. He's written books. He's probably pretty old, now. But, he was on television. He was on debating everybody. And, also, when the law was changed, he was part of the persons to make sure that it got changed. He did testifying to people so that the Supreme Court decision was what it was. So, he was responsible for that, as well as having blood on his hands. Totally changed."

Barbara also brought an unmistakable presence to the group. Her laugh was constant and infectious. She was often late and rarely failed to bring an apology and a mildly adventurous explanation of her delay. Behind this, though, was a proclivity for incisive questions, just as discussions seemed spent. Following a study guide question asking for an example of "God using you at a time when you felt weak," one of the women used the example of a confrontation with her Baptist sister-in-law. The use of Baptists as a theological Other was a commonplace occurrence within the group, but Barbara used the opportunity to pose a question: "Well, what does she, what do Baptists believe then if something should happen to that child before they are old enough to

decide for themselves?" This prompted a lengthy, stimulating discussion about the Biblical justification for baptism.

My role in the group was a (nearly) silent observer. I was included in the turn-taking style of reading assigned study texts and Biblical passages. Several women joked how they often forgot I was there. Sandy's response to a rare quip of mine gives a sense of this:

Sandy: Can you think of other leaders who have kind of a charisma, a way about them that draw people to themselves?

Jocelyn: Clinton.

Linda: Ronald Reagan. Or, even Abraham Lincoln.

Sandy: They say that Martin Luther was...

Margie: I heard he was a good teacher of young people. They flocked to his place. So, he must have had some special draw.

Sandy: He lived in a castle and filled it with people.

Jocelyn: Michael Jackson.

[[Laughs]]

James: I'm guessing that's the first, last and only time Martin Luther has been compared to Michael Jackson.

[[Laughs]]

Sandy: We FINALLY got a rise out of James. He has not said a WORD in months. I can't believe it!
[[Laughs]]

LCMS Men: The idea that ethnography is an unbiased, unprivileging endeavor is a taken-for-granted fiction in anthropology. The most we can hope for is to stay mindful of how we may be affecting our research, and do our best to temper it and factor it into our analysis. One of my ongoing struggles is with the LCMS Men. They were the group I enjoyed being with the most. Every Thursday I was out of bed at 6am, eager to be on time at 7am. In analyzing transcripts and deciding which interactions to use as representative examples, I am constantly drawn to this group.

My affinity for the group is not exceptional. The LCMS Men's Bible study was the largest group in my sample, with 27 regular members. About 20 of these men are there without fail nearly every week. The group has been meeting for ten years. They

gather every Thursday morning at a local restaurant, where they have a large room filled with circular tables reserved. When Pastor Daniel arrived in 2003, he took over the role of facilitator from Aaron, a lifelong LCMS member who still attends every week.

The group has grown steadily since Daniel took over. Daniel's acumen as a facilitator is one reason for this success. Daniel is hardly an imposing figure, standing five foot five with a bookish demeanor. He has an inviting manner, and is extremely well liked by his congregation. His wry sense of humor is equally a part of his affability. I liked him immediately. He plays the role of facilitator to near perfection. The men raise their hands to speak, and Daniel notifies them when the floor is theirs. He keeps primarily to open-ended questions. He tacks a brief comment onto the end of their contributions, offering a means for others to pick up the conversational thread. He has a way of softening dogmatic comments, and sharpening the more benign ones. He steers clear of long, preachy exhortations. And, he manages to raise potentially controversial and divisive issues without being controversial or divisive.

Daniel is cognizant of his skill in this area, and is most grateful for it. He thinks it important because of the group's make-up, and their ability to foster engaging discussions. He commented to me during an interview that he "eats breakfast with the leadership of the church every Thursday." Nearly the entire building committee, the youth minister, the church administrator, several ministry directors, and most of the church elders are faithful participants. They are a highly educated group. Among others, there is a University of Chicago lawyer, education consultants for the state, engineers, college professors, and physicians. The occupational diversity is mirrored by theological diversity. The resulting discourse is lively, and a certain reason why many incorporate the

group into their weekly routine. A steady stream of humor runs through discussions that are detailed, frank, and open for debate (see chapters 3 and 5 for examples of such interactions).

As the largest group, it was also the easiest for me to remain silent in. There were no objections to my presence, my use of a tape recorder, or my lack of participation. The men showed interest in the project, as well as my personal background. Throughout the research year, and still today, Daniel has been one of the biggest advocates of my work. His enthusiasm was apparent in our first meeting, and has not waned.

Restoration Movement

Prayer Circle: The Prayer Circle met in the home of Darren and Beverly. It is a modest, one-story house nestled in an affluent neighborhood. Beverly is a dental assistant at a local practice, and Darren is the executive director of an international ministry organization he founded in 1995. The group began meeting in 2003 when Darren volunteered to facilitate one of the 40 Days of Purpose home groups. The group bonded immediately and easily decided to continue meeting. They have capped the group at nine members to safeguard the cultivation of intimacy and trust. The Prayer Circle is the youngest of all the project groups. In addition to Darren and Beverly there is another married couple in their late 40s, a married couple in their late 20s, and three single women in their late 20s.

Darren is very much an authoritative voice in the group. His position in the ministry, as well as being a spiritual mentor to several of the younger group members, allows for a group dynamic that reveres his contributions. Darren is a soft-spoken

individual, and is not necessarily the first to respond to a question. But, when speaking, the other participants listen intently.

Darren also decides on the study format, a structure unique to this group within the project sample. After their 40 Days experience, Darren chose the topic for each meeting based on what "God had been speaking to [him] about" that week. More often than not, the topic was a spiritual lesson intended to increase one's "relationship with God." In addition to some thoughts of his own, Darren selected Biblical texts to demonstrate the lessons. This was usually a text Darren "felt led to" during the week.

Beginning with the fourth tape-recording, and continuing for the remainder of my fieldwork, Darren altered the format. Based on a book he read, Darren suggested a new approach where the group read a book of the Bible during the week then gathered to discuss "what God had been teaching [them]." The goal was to read the same book as many times as possible between meetings, ideally once a day. When they came together, the group went around the circle and individually "shared," pointing out specific verses that "spoke to" them. The group greatly enjoyed the method, continuing with it for 10 other Biblical texts (see chapter 3 for an extended discussion of this practice).

Following the last person's reflections on the reading, each individual shared his or her prayer requests and praise reports. They spent more time doing this than any of my other groups, devoting as much as an hour to prayer. Beginning with their ninth reading, Darren altered things again. Following the first person's sharing, he asked if they could connect what God had taught them in their Bible reading with their prayer requests. The group adapted effortlessly. Observing this prompted a minor epiphany about the group.

For the remainder of the night, as well as the subsequent meetings, each member spent

less and less time sharing reflections from the reading and more time on prayer requests. It was common for a single Bible verse and a few sentences to be followed by five minutes of prayer requests. This inclination toward prayer highlights the purpose of the group. Prayer requests are a way to invite others into your personal life. And, more so than any other group, extremely private information was spoken aloud in confident trust. Biblical analysis was clearly secondary to intensifying the atmosphere of shared intimacy.

Two groups accepted my presence as a researcher on a conditional basis. One women's Bible study requested that I not tape record. The other, the Prayer Circle, welcomed me into the group, but insisted that I be an active participant; silence was not an option. As the role of speaker moved around the circle to reflect on the Bible reading, I was expected to share, and did. Because of the nature of this format, I feel as though my contributions were of little consequence to what others shared. As explained above, the participation order was not interactive, but operated on a turn-by-turn basis. The time reserved for my sharing was no different than any of the other participants. And, there was no norm within the group that discouraged different individuals from talking about the same Biblical text(s) in their sharing.

"Mind of Christ": The room was affectionately called "the dungeon" by the RMC staff. The double doors to the side of the worship stage led to a small, rectangular room with concrete walls, slab floors, and no windows. Foldout chairs were arranged in several rows of five. A dry erase board covered the entire front wall, always bearing the remains of last week's lesson. The Sunday school class began at 9am, and by 8:55 most of the seven members were in their seats with their Bibles, notepads, and pens at the

ready. Rick had a knack for being punctual. He breezed in week after week, promptly at nine, excited and somewhat hurried. He carried his worn, leather briefcase in one hand, and his energy drink in the other.

Rick is a dynamic speaker. He commands the front of the room with an energetic, but not frenetic, presence. He stays mobile, but does not pace. He is clear, articulate, and forceful when emphasizing key points. He laughs, smiles, and gestures when enjoying a lighter moment. He makes proficient use of the board, jotting down words and phrases and sketching diagrams. He uses different color markers to signify a change in purpose. He stays faithful to the day's lesson, but clearly improvises; watchful for where existing ideas can be improved and new ideas inserted. He moves efficiently through each lesson, but does not rush.

The goal of "The Mind of Christ" is to "rid believers of the mindset they receive from the World, and equip them with a Christ-like way of thinking." He developed the course following a lecture he heard by an ex-Congressman on "understanding the mind of non-believers." He approached the speaker afterwards and requested a copy so he could develop a class around the ideas. He has been cultivating it ever since, incorporating new thoughts and adjusting the order and structure of the lessons. Currently, it is designed as an eight-week class. It progresses from describing the nature of the "Worldly mind" to describing the nature of "Christ's mind" to explaining the transition from the former to the latter. Every class member is given a 13-page course packet that includes an outline of each lesson, diagrams to fill in as Rick presents them on the board, and a list of his favorite quotations. The quotes come from a range of sources, including: Biblical texts,

Christian lecturers, Christian authors, sports figures, former U.S. Presidents, and literary figures.

The group consisted of little discussion because it is framed as a "class" that Rick "teaches." In each one-hour meeting, Rick talked the majority of the time. Participants' questions were brief and usually requests for further explication. This helped me as a silent observer. My use of a notepad, which was obvious in other groups, blended in with the class members' note taking.

Monday Group: Rick also served as the facilitator for the Monday evening Bible study hosted by he and his wife. Their group dynamic was much different than "The Mind of Christ." There were three regular members, and three other occasional participants. The regular participants were Rick, his wife Janice, and their close friend Tim. Each week the group gathered around the dining room table, everyone was within a few feet of each other. Our Bibles were spread out in front of us, and we alternated reading through the study text. Rick instructed us on how much to read, typically between five and ten verses, after which we stopped for discussion. Rick alternated between offering his comments immediately and beginning with asking what others thought of the text. Once the discussion had run its course – signified by silence or joking tangents – we moved on to the next reading.

Like the Prayer Circle, the Monday group began as one of the 40 Days home groups. Following the Warren study, the group has studied directly from the Bible. They began with the Gospels, and then moved to Paul's letters. During my research, they worked through Hebrews, 1, 2, and 3 John. The three regular members have been the

only ones attending for the entire two years. I still do not know why the remainder of the group membership has been fluid and unpredictable.

My roles as a participant and an observer were more crucial with the Monday Group than with any other group because of its size and intimate setting. My presence would have been far more obtrusive were I to remain silent, concentrating on note taking. As a result, I contributed more to this group's discussions than in any of the others. At times this meant I unwittingly shifted the direction of the conversation. On one occasion in particular, during a discussion of Gnosticism in the first century Church, I asked what they thought the biggest threat to the modern Church was. Fifteen minutes later we moved on to the next Bible reading. In the analyses that follow, I have restricted my use of examples from this group to occasions where I was marginal or absent in the discussion.

How Small Groups Work

Earlier in this chapter, I emphasized the groups' diversity. In the final section I describe sample-wide patterns. I focus on issues that are integral to questions of group discourse, particularly those integral to the remaining chapters: interactive frames, participation structure, text selection, and Bible use.

Interactive Frames

Small groups cultivate expectations about their central purpose. Participants join groups, remain in groups, and change groups partly depending on whether their expectations about the group are being met. This is not unique to small group Bible study, but a characteristic of any discourse event. Some of the earliest work in the ethnography of speaking focused on how speech events are defined, the norms of

interaction, and how social actors learn to function in them appropriately (e.g., Bauman and Sherzer 1974, eds.) Socio-linguists doing interaction analysis have focused on the way differing structures of expectations help distinguish between interactive types.

Ultimately, appropriate conduct and successful communication rely on participants correctly discerning what "interactive frames" are operative and predominant (e.g., Tannen and Wallet 1987). Any analysis of small group Bible study, then, requires an analysis of what participants' expectations are for group life.

Two distinct sets of expectations emerged for defining what groups perceive to be their primary purposes. First, there is the expectation of intimacy and personal "sharing." The emphasis here is on keeping the discussion connected to the lives of group members: emotional problems they are facing, efforts they are making at spiritual growth, work and relational situations they are struggling with. The hope is that as groups spend more time together, and as members come to know one another better, they will be increasingly willing to share the intimate details of their lives. For some, no matter the topic or text of study, this is the central purpose of small groups. The Prayer Circle best embodies this approach. Its strict adherence to this expectation became clear once Darren altered the format, and prayer and commentary occurred side-by-side.

The other set of expectations involves engaging in analytical discussions. The subject could be moral, historical, theological, political, Biblical, or virtually any other area. No matter the topic, the emphasis remains on ferreting out different ideas and their consequences surrounding a particular issue. At times, this consisted of a single individual asserting his position, the group agreeing, and the discussion moving on. And, in other cases, it was a lengthy exchange involving multiple participants and conflicting

opinions. The group that gravitated toward this approach most was the LCMS men. Their intricate exchanges made little reference to their personal lives, except as an occasional knowledge source invoked to pursue an analytical argument.

Very rarely did individual groups restrict themselves to one or the other of these interactive frames. More often, groups made the conscious attempt to balance these approaches, to devote discussion time to both. The case of the McGee men is an excellent example of how this can be a struggle. Participants stayed committed to and withdrew from this group based on the fulfillment of expectations. In the words of one member, "some left because they think we're too 'out there,' and some won't come back because we're 'not out there enough!" In other groups, satisfying both sets of expectations is precisely what members desire. The ability to move back and forth in the course of a single meeting can be a major source of attraction.

These two interactive frames, despite vastly different aims, are united by an underlying expectation. All groups – no matter the type, topic of study, demographic make-up, denominational affiliation, or size – share the common expectation of seeking to grow spiritually. A variety of terms are used to capture this – "being fed," "growing in relation with the Lord," "fellowship" – but the most common is "edification." Leaders, regular members, marginal participants all spoke of coming to small groups because they want to be "edified." They want to advance in their spiritual lives. They come to support each other, learn from each other, and share in a mutual maturation of faith. Equally important is what this expectation removes. If spiritual growth ranks above all else, then certain dynamics will be avoided at all cost; most notably, ongoing tension, hostile debate, and belligerence. And, indeed, groups work to eliminate these. This also has a

number of consequences, including: individuals choose groups not attended by people they are at odds with, controversial topics become taboo, and leaders work to achieve consensus rather than create division. All of these can be complicated discursive processes, but they all support the same goal – keep small groups edifying.

Participation Structure

One of the first things you realize when doing an ethnography of small groups is who talks and who does not. This may be partly reflexive. Your attention is drawn to those who are eager to speak. Then, you catch yourself ignoring those who are not. This may produce biases in the research, or at least the researcher, but it underlines a central topic for the study of discourse: participation structures. Accounting for available roles has been integral to the study of social institutions broadly, and the division of speaking roles has been particularly important for understanding speech events (cf., Goffman 1959). As a social institution, small group Bible study contains (at least) three definable role structures: facilitator, core participant, and reserved participant.

Facilitators – like Charlie, Ted, Sandy, Daniel and Rick – were the ones who organized the meetings. They decided when to begin, when to end, when to move on to the next reading, and when to pause to consider a question in depth. The central discursive task of the facilitator, though, is to generate participation. All of the project groups, besides the "Mind of Christ," understand themselves as discussion-based. Thus, all of those who "led" did so with the intent of involving as many members as possible as often as possible, and to avoid lecturing. Given the reticence of some to contribute, and the ever-present potential for divisive topics, this can be a difficult accomplishment. To fulfill this role structure, facilitators made use of numerous discursive and interactive

strategies. These men and women would often spark discussion by asserting their own opinion in concise terms, and then ask for reactions from members. Other possibilities include posing questions to the group as a whole, posing questions specifically to individuals, and using outside textual materials to assert an idea (e.g., Bible footnotes and Bible commentaries). Individuals invariably gravitated toward one or more of these strategies, but all were resources in the ongoing attempt to keep as many participants as possible active in the discussion.

Earlier in this chapter, when comparing the total number of members in each group, I distinguished between occasional and "regular" members. Regular members, those who attended at least half of the recorded meetings, are not always the most active participants. And, more important, they are not always the core participants: those whose contributions carry greater authority, and who other members look to when discussions go stale or especially difficult questions arise. Facilitators were always part of this category, but the overall number of core participants varied from three to six in each group. Core participants were very much responsible for carrying the discussion. They were not always the most talkative, but their talk reliably added a needed comment at the necessary time. This is true in the context of both interactive frames described above. In intimacy-oriented situations, core participants could be counted on to keep the discussion focused on personal detail. And, they were the first to increase the intimacy level by sharing deeper personal stories or inquiring further into what another member shares. The same is true for analytical situations. Core participants ensured discussions remained provocative and intriguing, moving the discussion along when it became frivolous.

Every group also had its more reserved members. The motivations for this were numerous. Some seldom spoke because they were new to the group. Others felt they learned more by listening to those with more "mature" faiths. And, still others told me they were "just shy" by nature. Complete silence was reserved for the larger groups. In the Monday Group, with only three regular members, everyone contributed if only sporadically. The LCMS men, with two dozen members present each week, had several participants who went several meetings in a row without contributing a word.

The resulting participation structure of facilitator-core participant-reserved participant was consistent throughout the entire sample. From the perspective of discourse and institutional analysis, this is predictable. Social institutions are comprised partly by the established roles that generate expectations (cf., Bourdieu 1977), and speech events are known for having definable participation structures (e.g., Philips 1974). Thus, the prevailing participation structure does not mark small group life as a particularly Christian practice. This participation structure does, however, buttress other interactive features of small groups as a discourse event; in particular, the maintenance of interactive frames and, as will be shown below, the process of text selection.

Text Selection

Its choice of study texts is an integral component to the life of every small group. Books, Biblical texts, videos, periodical articles, email testimonies, and audiotapes were all used as the basis for discussion. These texts were chosen for a variety of reasons. The Iconoclasts chose <u>The Jesus I Never Knew</u> because the facilitator and his wife had read it, enjoyed it, and thought it suited the group. The McGee Men chose J. Vernon McGee's study guide because they wanted a change from their previous commentator. They hoped

a new style would attract new members. They also liked the idea of using McGee's Fundamentalism as a ground for disagreement, which would ideally force them to make their own thoughts more explicit. Other groups chose texts based on national popularity, recommendations from other groups, familiarity with an author, and an author's affiliation with recognized institutions.

In her ethnography of Houston area book clubs, Elizabeth Long (2003) analyzed their text selection practices. Some preferred a democratic ethic, while others relied on the wisdom of the leader and still others circulated the responsibility among everyone. The process of text selection in small group Bible study was similar, but not identical. Decision-making responsibility virtually never moved from member to member. The Vineyard groups came closest to that dynamic, opening the role of facilitator to everyone, and encouraging different leadership each week. However, this did not always involve selecting a text. Individuals could devote the meeting to a topical discussion, or the sharing of a personal testimony. A method not applicable in Long's research on book clubs is the use of congregation-wide studies, where the pastor or small group director chose the text.

The democratic approach (usually an informal vote following a suggestion) and reliance on the facilitator were represented equally in my sample. Some groups, like the Iconoclasts, alternated between these two approaches from study to study. Sandy attempted a democratic approach with the LCMS Women, but the group quickly tired of the resulting indifference and was content to place the full responsibility on Sandy. For the LCMS men, Daniel takes suggestions toward the end of each study, and then makes an executive decision based on the possibilities he receives. For the Prayer Circle, though

the idea for Bible reading came from Darren, the choice of individual books was more democratic. In the meetings following Darren's initial choice, he would ask the group what they wanted to do next. There was little in the way of voting, though. The first suggestion put forth predictably became the one accepted by everyone else.

Text selection is not always such an agreeable process. It can also be divisive. Given how meaningful these groups are to their members, and the importance of the text for structuring discussion, individuals can hold strong opinions on what they wish to study. And, they can be equally vehement about what they do not want to study. Such divisiveness usually indexes larger points of divergence within groups. The clearest example of this was the DUMC women's Bible study.

The seven women, all regular members, took turns at the role of facilitator, and they chose texts by unanimous agreement. The interactive frame was more oriented toward analysis, but was not devoid of cultivating intimacy. Their discussion of issues often became contentious, at times threatening the goal of edification. The reason was a conflict between a theologically conservative member, Wanda, and the rest of the group. Wanda grew up in the United Methodist Church, but spent her adult life in the charismatic tradition. She returned several years ago because she missed the hymnal music, and she liked the community outreach at DUMC. The other women were decidedly liberal in theology: resisting masculine references to God, dismissing the doctrine of original sin, and freely questioning the authority of the Bible on a variety of matters.

Socially, the tension was not so readily apparent. Wanda was not excluded, and did not exclude herself, from fellowship before or after meetings. During my initial visits,

before the theological conflict escalated, everyone appeared to enjoy each other's company. As my research progressed, though, the tension between the women during discussions heightened from week to week. When I began data collection, the group had just finished its previous book study. There was no consensus on the next text, so the women spent a few weeks watching movies while they decided: one on Martin Luther, one on Joseph Campbell, and one criticizing <u>The Da Vinci Code</u>.

For weeks, Wanda had been advocating for Bart Ehrman's The Lost Christianities (2003), a scholarly study of the Gnostic scriptures. She was sensitive to her position as the "lone conservative." She thought this a good compromise, and was confident the historical and textual evidence would convince the others of the Gnostic scriptures' lack of authenticity. In turn, Wanda hoped this would move the women to reconsider the authority of the orthodox canon. The other women were leaning toward a (very) different text: Deepak Chopra's How to Know God (2001). Chopra is a well-known spiritual author well outside the orthodox Christian tradition. He does make reference to the Christian Bible in his books, but his conception of "God" draws on other faith traditions as well, including Eastern and Native American epistemologies. Despite Wanda's strong opposition, the other women decided on Chopra's book for their next study text. The week they announced this was the last week Wanda attended the group. Between this meeting and the next, Wanda sent a lengthy email to the group. It informed them of, among other things, her decision to leave, her disgust with the Chopra decision, and her concern for their salvation.

This example also illustrates how text selection provides a stage for communicating identity. Groups and individuals favor texts they identify with. Very

rarely did groups choose texts that were at odds with their theological position or texts that otherwise conflicted with their moral and spiritual sensibilities. This is not to say that groups (like the Iconoclasts) never chose challenging texts, or texts that called into question certain taken-for-granted beliefs. Groups often reflected at the end of a study how a text was "not what they expected" or how it "made them think." This element of surprise is part of the attraction of small groups, and the enjoyment of encountering new texts. But, still, their subsequent choices of texts remained faithful to their comfort zones of identity.

Using the Bible

All of these small groups are about Bible study to some extent. Five of the project groups read directly from a Biblical text: McGee Men, Tuesday Men, LCMS Men, Prayer Circle, and Monday Group. In numerous others, a Biblical theme or character was the topic of study: the life of Jesus, how God speaks in the Bible, prayer in the Old Testament, the life of Paul, how God shows love in the Bible, and Bible prophecy, to list a few. For those who read non-scriptural texts unconcerned with a single Biblical theme, the Bible was still incorporated, sometimes at length, throughout the book. Authors often asked readers to consult the Biblical texts being referenced. For groups like the Vineyard, where the topic may be a personal testimony, Biblical texts were woven into songs, prayers, testimonies, and responses. In a sense, then, all small groups are Bible-centered. It is a shared text, and a shared body of knowledge that groups rely on and return to in discourse.

Bibles were physically present as well. In certain groups, including some not directly studying from scriptural texts, participants arrived with Bibles in hand. On the

rare occasion that a Bible was forgotten or misplaced, facilitators were sure to have an extra copy. Some individuals arrived each week with numerous translations, privileging one and using the others for alternative readings (see chapter 2 for further discussion). In other cases, facilitators took it upon themselves to place a stack of Bibles in the study space.

When a book of scripture was not the topic of study, Biblical texts were still frequently consulted in cases of curiosity, clarification, and comparison. If the details of a particular story were fuzzy, it was located and read. If the discussion of one text prompted reference to another, it was read. Other Biblical texts were searched for their aesthetic appeal. The mere mention of certain Biblical texts, Psalm 23 being a popular one, resulted in reflections on the "beauty of scripture." These were not read and appreciated for interpretive purposes, but for their poetic qualities. Biblical texts were also read for their connection to specific memories, pointing out the dialogic quality of scripture (Bahktin 1981). Previous readings of the text for comfort or council, linked to times of joy and suffering, led to rereadings and recollections in the midst of group discourse.

Since the practice of small groups is, in some measure, always about the Bible, the remainder of this dissertation concentrates on this quality of group discourse. An analysis of how the Bible is perceived, used, read, interpreted, and incorporated is central to any understanding of small group discourse. The next chapter opens this analysis by exploring the system of assumptions that participants make about the Bible. These assumptions exist prior to any, and guide every, particular instance of Bible reading.

This chapter has described and analyzed small groups as a form of cultural practice. The following summarizes my major observations about my sample of small groups:

- ➤ All six project congregations emphasized the development of a small group program. Small groups were advertised as both a means of numerical growth in the church and of spiritual growth for individuals.
- ➤ Small groups gravitate toward or balance between two interactive frames, one emphasizing the cultivation of intimacy and one emphasizing analytical discussion. These two frames are united by an underlying expectation of small group life to ensure that small groups are a setting where individuals feel they are strengthening their faith.
- The participation structure of small group life is consistent among the project groups. It consists of three established roles and their accompanying expectations: facilitators, core participants, and reserved participants.
- The process of text selection provides an arena for the communication of group identity. Group chose texts that promised to affirm and/or reconsider their self-perceptions regarding who they are as Christian believers.
- All of the project groups stress Bible study. For groups that read directly from Biblical texts, and those who do not, a primary goal of meeting together in small groups is to "become stronger in the Word."

I present these claims of pattern and organization against the backdrop of a seemingly diverse sample of groups. Indeed, these groups appear to vary on every demographic and sociological dimension. They represent multiple Protestant traditions, different types of

study, men and women, working to upper middle class members, and a variety of ages. However, structurally and interactively, these groups prove to be very consistent. And, yet, I have also taken this chapter as an opportunity to demonstrate the uniqueness of individual group environments. Despite the consistency that exists amid diversity, one group always has the potential to 'feel' very different from another. The experience of being a participant observer, and of hearing members talk about their own experiences as group participants, is quite different in the Iconoclasts, the McGee Men, the Tuesday Men, the LCMS women and men, the Mind of Christ, and the Prayer Circle. This unique feeling combines with the other features presented in this chapter to make these groups meaningful to members. As I stated at the outset, members have deep emotional connections to their groups. They love them. And, they find time, week after week, in spite of all else they might be responsible for, to take part in the small group experience.

Chapter 2: Textual Ideology and the Bible

"The Bible is first of all...a mosaic: a pattern of commandments, aphorisms, epigrams, proverbs, parables, riddles, pericopes, parallel couplets, formulaic phrases, folktales, oracles, epiphanies, *Gattungen, Logia*, bits of occasional verse, marginal glosses, legens, snippets from historical documents, laws, letters, sermons, hymns, ecstatic visions, rituals, fables, genealogical lists, and so on almost indefinitely" (Frye 1981: 206).

This quotation from Northrup Frye, the eminent literary critic, aptly captures the complexity of the Judeo-Christian scriptures. If you consider it for a moment, the task of reading (not to mention understanding) the Bible is overwhelming. Take, for example, the problem of translation.

The vast majority of Bible readers in America engage the text in English. The 39 books of the Old Testament were originally written in Hebrew (except a few brief sections in Aramaic). The 27 books of the New Testament were originally written in "koine" Greek. These languages have little in common, structurally or semantically, with English (Dewey 2004). Aside from the purely linguistic problems, the process of translation involves a myriad of decisions, from the stylistic to the theological. We might also consider issues of manuscript transmission. Historical discoveries have never unearthed an autograph Biblical text²². Only a handful of copies exist in complete form. And, the oldest available manuscripts lack important, meaning-contingent features like capitalization, spaces between words, and punctuation (ibid).

Or, take the problem of historical and cultural distance. The social worlds of the Bible's authors were very different from those of modern American readers. Customs, habits, scenes, and locations are not the same. The prevailing knowledge structures differ substantially, from science to theology, competing philosophies, demography, and so on. These all find expression in language, posing further difficulties. The cultural worlds of

_

²² Autographs refer to the original written versions of Biblical texts.

the Bible lent themselves to idioms, metaphors, and analogies that are difficult to translate and often difficult to render intelligible to modern sensibilities (Dewey 2004).

Bible reading could be seen as the height of futility, or perhaps daring, reserved for those in love with the impossible. This conclusion is perfectly logical; yet millions of people do it everyday throughout the United States. Millions gather on a weekly basis to do it together. People with and without formal training in languages, hermeneutics, theology, and history read this text with confidence and great joy. In doing so they find meaning, comfort, security, inspiration, council, strength, and conviction. They are surprised and reassured, troubled and encouraged. All this begs an important question. How? How do readers interact with the Bible? In the chapters to follow, I analyze how the Bible is read, interpreted, and incorporated into small group discourse. This chapter provides a necessary first step en route to these other issues. Before asking how the Bible is used, we must ask how the Bible is understood. What do readers bring to their interaction with this text? And, what does this interaction look like when performed collectively? The answer is what I call the "textual ideology" surrounding the Bible.

What is Textual Ideology?

By textual ideology I mean the set of ideas, assumptions, and beliefs readers bring to their reading of a text²³. Textual ideologies exist prior to any given instance of reading, and they orchestrate how readings unfold. This concept resonates with three others in the

2:

²³ In his survey of the term, John Thompson observes two distinct conceptions of "ideology" in the social science tradition: descriptive and evaluative (1990). The latter understands ideology as meaning implicated in the reproduction of unequal power relations (cf., Fairclough 1989). The descriptive conception is "neutral," concerned with identifying systems of thought and belief. This is the sense of "ideology" I employ. However, the evaluative sense is not altogether disconnected from the Bible's textual ideology. As we will see, parts of its make-up are the inspired and absolute qualities of Biblical authority. This includes a closed scriptural canon. This is ideological in the evaluative sense when we consider attempts to add to or take away from what is understood as part of the Bible. For example, the rejection of Gnostic texts – most recently the "Gospel of Judas" – is ideological in this sense.

hermeneutic tradition of the social sciences: genre, conversational inference, and the hermeneutic circle. The concept of genre has raised similar issues among linguistic anthropologists and discourse analysts. Bauman defines a speech genre as a "style oriented to the production and reception of a particular kind of text" (2004: 3-4). This latter aspect of genre, its ability to impact how texts are received, is the closest area of overlap with what I am presenting as textual ideology. The key difference lies in the reason for impact. Genres are distinguished by stylistic, formal features of textual structure; whereas the concept I am developing here rests on ideological differences. The two do share the Bahktinian quality of being dialogical. Our interaction with a text is not isolated, but cumulative. No reception is fully detached from the experience of previous receptions or the expectations of future receptions (cf., Bahktin 1981).

Second, the field of conversation analysis has had an ongoing interest in the background knowledge that participants bring to face-to-face interactions. Grice (1975) developed the notion of "conversational implicature" to address how processes of implication are required for successful communication. Gumperz (1977) suggested that conversational interpretation is a situational and culturally bound process where participants gauge their response by their evaluation of others' intentions. Fairclough (1989) has contributed the significant observation that background knowledge – in spoken and written discourse – is also involved in the reproduction of dominant meanings and unequal social relations. This tradition informs the concept of textual ideology by its emphasis on what social actors bring to their experience with texts.

Lastly, those acquainted with principles of the hermeneutic circle will find textual ideology familiar. Part of this tradition is the recognition that readers approach texts with

biases and prejudices that condition how texts are understood. Included in these biases and prejudices are the "presuppositions about the nature and creative source of the text itself" (Bartkowski 1996: 262). These allow texts to generate multiple interpretations. The concept of textual ideology casts a broader net than deciphering textual polysemy, or from above, aiding effective communication and reproducing cultural forms. Textual ideologies can impact what readers expect from a text, how they evaluate different (and, divergent) interpretations, what they read, when they read, where they read, who has the access and legitimacy to read, and the method(s) of reading. The repetitiveness of a strict hermeneutic circle also belies the dialectic potential of textual ideologies. Readers' experience with a text, with cultural resources, and with social roles and relations can alter their understanding of a textual ideology, which can then shape future readings.

Textual ideology is an ecumenical concept, not restricted to any particular category of text. As readers, we are the bearers of textual ideologies for virtually every text imaginable. Even the most mundane, superfluous, or trivial text is not exempt. Indeed, once it has been judged as mundane, the work of generating expectations and assumptions has already begun. We then go about interacting with the text in a way that reflects this attitude and understanding. When I read a new cookbook recipe, I expect to be guided, perhaps very gently, from beginning to end. When I read a comic strip, I expect to be amused in some way. And, just as certain expectations are generated, others are eliminated. My cookbook will not be the source of financial guidance, and my comic strip will sorely disappoint if it causes melancholy. These processes become increasingly complex as we encounter texts held in greater esteem. Take the following as an example.

In his ethnography of American karaoke, Rob Drew (2001) observes that songs are texts that performers encounter. In the interaction between text and performer, Drew describes the faithfulness performers habitually ascribe to original texts. "The very absence of the familiar star's voice seems to summon you to fill it in. You find yourself instinctively reproducing the most convoluted cadenzas and subtle sighs of the original recordings without any prompting from the lyric monitor" (ibid: 39). He goes on to argue: "The songs that undergo the least modification in karaoke bars are, generally, the ones performers and audiences feel most passionate about" (ibid: 57). He uses a performance of his own in a working-class bar – Bon Jovi's "Livin' on a Prayer" – to demonstrate. "Many in the audience sing along with spirit. They know their Bon Jovi, as can be expected from the club's bridge-and-tunnel crowd. That means if I botch it, I'll be twice as culpable – I'll have sullied a canonical text" (ibid: 6). In terms of textual ideology, Drew identifies a responsibility, a precision, and a seriousness performers approach certain texts with – those deemed "canonical²⁴."

Cookbooks, comic strips, and karaoke are a clue to the inclusiveness of textual ideology as a concept. The karaoke example introduces the category of text I am concerned with: sacred texts. Drew's work suggests that 'sacredness' is partly constituted by the link forged between the text and the socially constructed value of its reception by an audience. Within this category of 'sacred,' texts tied to religious practice hold a

²⁴ A valid question to ask of textual ideology is why it is not an issue of habitus (Bourdieu 1977). After all, textual ideology and habitus seem to have several things in common. Both are structuring mechanisms. Both are comprised of predispositions. Both operate on individual and collective levels. Both carry a dialectic sense. Why, then, am I not presenting an analysis of 'textual habitus?' The primary reason why textual ideology is not a sub-category of habitus is its self-conscious quality. In Bourdieu's framework, habitus is a system of predispositions existing mostly outside the conscious reach of its inheritors. It is an issue of "learned ignorance" (ibid). This is not the case with textual ideology. Each of the individual principles I describe is well known and well articulated by believers. Indeed, their articulation in small group discourse is largely what enables this analysis.

special place in the genre because of their institutional basis (Kort 1996, Malley 2004). Few genres are as universally powerful in constituting belief and practice, or in garnering authority (Briggs and Bauman 1992). In turn, there is greater potency to textual ideologies surrounding religious texts. Given this significance of religious texts and the centrality of textual ideologies to the social life of texts, the relation between the two deserves scholarly attention.

Research in the anthropology of religion has identified the concept of "scripturalism," the epistemological basis for conceptualizing scriptures (Malley 2004). My analysis of textual ideology also emerges from this work. Through his ethnographic research on a Midwestern Baptist congregation, Brian Malley developed a working model of Evangelical Biblicism. This cognitive-oriented model emphasizes two aspects of how Evangelicals typically think about the Bible. First, the Bible is attributed a sense of absolute authority that derives from the doctrine of divine inspiration. And, secondly, this authority is maintained by an interpretive tradition: "a species of belief tradition in which a set of beliefs is transmitted along with the attribution of those beliefs to a text" (ibid: 146). Malley's observations are largely borne out in the chapters that follow, but I approach this phenomenon from a different angle.

My research shifts the focus of inquiry to the interaction between the epistemological structure surrounding the Bible and its evidence in discursive practice. In other words, I am interested in how attitudes towards the Bible interact with actual, collective practices of reading and using scripture. This requires formulating a concept of textual ideology that explicates the individual principles of assumption, expectation, and belief. John Bartkowski's (1996) work on Bible interpretation provides a second source

of insight for this research. He uses the example of divergent theological legitimations for child discipline in the work of two conservative Protestant writers. The authors differed in their understanding of the Bible's "overall message." One author emphasized the imperative of unconditional love, while the other focused on the theme of sin and punishment. Their interpretations of individual passages (e.g., "spare the rod") took shape in accordance with the respective emphases of love and punishment (ibid). Bartkowski concludes his article by stating the need for a similar inquiry to be done ethnographically. This dissertation provides a response to his call. I begin with an outline of the principles that comprise the textual ideology surrounding the Bible.

Textual Ideology and the Bible: six core principles

Textual ideologies can be as simple as a single presupposition, or as complex as an interconnected system of assumptions and beliefs. Through my ethnographic research, I have identified six principles that comprise the core of the textual ideology surrounding the Bible²⁵. All were present in group discourse, as well as in the less structured interactions before and after meetings. Multiple participants in each group, men and women, young and old, the spiritually mature and the young in faith, core and marginal participants all expressed these principles. In short, an adequate account of this textual ideology (as well as of group discourse generally) relies on explicating these six principles. They vary in kind, dealing with the authority of the text, the relevance of the

25

²⁵ I had a defined decision-making process for classing principles as either "core" or "non-core." This included two factors. First, meta-commentary statements about the Bible that appeared on only a few occasions and/or only among a specific group or individual were not included as "core" principles. Second, meta-commentary statements about the Bible that demonstrated no discernible manifestation in textual practice were not included as "core" principles. "Non-core" principles consisted of only a few types of statements (e.g., "You can tell the difference between Paul's writing and other New Testament writers").

text, and the Bible's textuality. Each raises a distinct set of expectations about Bible reading and possibilities for interacting with the text.

Biblical Authority

1) The Bible is the Word of God. The importance of this principle cannot be overstated. It encompasses many of the stock phrases and doctrines commonly heard about the Bible, such as: inerrancy, infallibility, inspiration, and literal truth. These terms are indexical, a discursive shorthand signifying a foundational (albeit, complicated and contested) understanding. They describe the absolute authority that is attributed to the Bible. It is absolute because it prevails over any other source of instruction, in matters ranging from the practical to the moral to the spiritual²⁶. Malley (2004) observes that this authority derives from the Bible's presumed inspired character. It is not simply a human product, but God's revelation to humanity. For these readers, the Bible is the only text that, in its entirety, bears the co-authorship of the divine. In this way, it is unique among all texts for them. In turn, they revere and read the Bible unlike any other text.

This principle guided the LCMS Men in their text selections. Before the group started Proverbs, I asked Pastor Daniel why this was their first study of the Bible in over a year. He explained that he preferred reading non-scriptural texts with the group because there is greater freedom to disagree with these books. He thought this resulted in better, more active discussions. The following exchange came at the end of the first meeting in the Proverbs study:

Daniel: As we read through this let's keep in mind that all this stuff [is] true. I think we should tell ourselves that every once in a while when we pick up the Bible. Now, I might not be able to figure out the whole Truth. I might not be able to figure out how to apply

²⁶ In his analysis of Biblicism, Malley observes that there is often a disjuncture between the principle of Biblical authority and the way believers act on that authority (2004). The practice of Biblical authority is a separate process that allows for "selectivity and ad hoc hermeneutics" (ibid: 144).

all of that Truth in all of the ways in all of my life. But, we should at least approach this from the standpoint, "This is right. And, this is more right than anything I would have in mind if I'm thinking differently." This is the wisdom of the ages that comes to us from on high, as opposed to the newspaper [where we] say, "Well, maybe that's right, maybe it's not."

Pierce: Yeah. As much as I've enjoyed the books you've picked out, and the great discussions, this is Truth. This is the highest form of Truth.

Daniel: I do think it's good for us to be doing just what Pierce would say, reading the text. One of the things I think happens when we read the books that's helpful is that I believe we're more free to disagree with the books because we know it's not God's word from beginning to end. And, we often feel intimidated understanding the Scripture and sharing the Scripture, and we're afraid we're going to get something wrong. But, when we're reading a book, I think we have a little greater freedom of, "Well, I think I can have an opinion on that."

The contrast between the textual ideology surrounding the Bible and that surrounding non-scriptural texts is clear: books are debated, whereas the Bible is discerned. This also illustrates the dissonance between the divine authorship of the Bible and the human audience. As fallible human beings, we are unable to always, in the fullest manner possible, understand the message of Biblical texts. What revelation does occur happens through the guidance of the Holy Spirit, and then only partially. This ensures that Bible reading is done continually, and also highlights the importance of reading as a social activity. Interacting with the Bible as a group, as part of a shared community of believers, makes this process of discernment more beneficial.

Recontextualizations of Biblical texts are used to testify to the Bible's absolute authority. II Timothy 3:16-17 is commonly used for this purpose: All Scripture is Godbreathed and is useful for teaching, rebuking, correcting, and training in righteousness, so that the man of God may be thoroughly equipped for every good work. The inspiration and authority of the Bible also results in the view that the Biblical canon is finalized, closed off from addition or subtraction. This too is understood as a scriptural mandate. Revelation 22, for example, warns about taking away or adding to prophetic words. For these believers, God's involvement in His Word extends beyond the writing of Biblical

texts to the permanent canonization of certain texts (i.e., the 66 books of the Protestant Bible) and the exclusion of others (e.g., Gnostic texts, Apocryphal texts).

As I mentioned above, this principle is both complicated and contested. Among other reasons, it is complicated for the nuances it entails. For example, anytime I inquired further about the Bible's "infallibility" or "inerrancy," group participants qualified that this is only true for the autograph texts. This addition tacitly recognizes the difficulties of translation and transmission, while still upholding the quality of absolute authority.

Insistence on a "literal" approach to the Bible presents a similar issue. As Malley (2004) argued, "literalism" is not proof of a hermeneutic tradition but a symbolic term denoting the veracity of the Bible. One might translate the claim "The Bible should be read literally" as something akin to "The Bible should be read as True, and human wisdom or reason should never be taken as proof against the Truth of God." What is being signified via "literalism" is a submission to and acceptance of authority, not a consistent interpretive style.

Like most of the principles I review here, understanding the Bible as "the Word of God" does not uniformly hold the same meaning for all groups and individuals. The LCMS Men provided an example of an inability to challenge the Bible. Or, as Daniel phrased it earlier in the same meeting, "If the Bible has one opinion and you have another, by definition, you're wrong." The McGee Men represent a different vantage point on this principle. On several occasions during their study of the book of James, Roland, the pastor of DUMC, stated that certain scriptures are "not always helpful" because scripture is "so absolute" and life is "all this in-between stuff." These two pastors are clearly making different pedagogical efforts in their groups regarding what freedom

there is to disagree with the Bible. However, participants like Roland still refer to the Bible as "the Word of God." And, more importantly, they still approach the Bible with the same expectations of being taught, counseled, convicted, comforted, and so on. In short, Biblical authority is conceived in different ways, but the consequences of expectation remain, as does the uniqueness of interacting with the Bible compared to other texts.

2) The Bible is the same today as it was when it was written. The authority of the Bible also derives from the constancy of its message. The Bible is understood to be eternal, just as true tomorrow as it is today as it always has been. One manifestation of this timelessness is the view that the Bible reveals the nature of humanity. Historically in Protestant theology, this has been captured as the doctrine of original sin. People are born into this world as sinful beings, separated from the righteousness found in a relationship with God. This nature has remained unchanged through time, and is only completely alterable in the individual life through the decision to follow Jesus.

The groups I studied emphasized how the sinfulness of humanity is revealed in the Bible through the moral depictions of characters and events. The Old and New Testaments are replete with portraits of individual lives, demonstrating the natural susceptibility to evil and the ability to overcome evil with God's help (e.g., David, Solomon, Peter, and Paul). This principle results in the discursive practice of constructing associations with Biblical characters. Susan Harding (2000) observed a particular form of this phenomenon in the discourse of the Fundamentalist preacher Jerry Falwell. His narrative practices invited audiences to listen to his words in the same way they might read the Bible, thus treating him in a typological manner. The participants in my research

followed this principle to a somewhat different discursive end. They identified themselves, their tendencies, and their actions with those of characters portrayed in the Bible. This was not done toward self-effacing or self-promoting ends, but rather to highlight an enduring characteristic of God: his desire to use imperfect people to accomplish divine purposes. This practice was particularly apparent in the Iconoclasts' discourse.

In their study of Philip Yancey's The Jesus I Never Knew they often returned to this principle of 'if God can use them, he can use us.' Charlie was fond of referring to the 12 disciples as the "Keystone Cops²⁷," calling attention to their failures to accept or apply Jesus' teachings. Yancey poses the following question about the disciples' triumphs as evangelists after making this same argument: "What else [besides the Resurrection of Jesus] explains the whiplash change in men known for their cowardice and instability?" (1995: 216). In response, Charlie reminds the group that the twelve men responsible for spreading Christianity were the same ones who "fell asleep in the garden," "argued about who was second in line," and "who were not the intellectuals or leaders of their day." They then drew the link between themselves and the disciples, and their own ability to "be disciples in today's world despite all [their] shortcomings." The same discursive process of constructing parallels occurred with various Old Testament characters revered for their weaknesses and failings, as much as their acts of righteousness.

Understanding the Bible as unchanging also results in the belief that "all truth is God's truth," quoting again from the LCMS pastor. Anything accepted as true because of human wisdom can ultimately be confirmed by the Bible. And, if not, its veracity should

²⁷ The <u>Keystone Cops</u> was a series of silent film comedies between 1912 and 1917 featuring an incompetent group of policemen.

be doubted. In short, every established truth is, in fact, a Biblical concept. The epistemology of Western science was a popular example for many of the group participants who engaged this topic. The assumption at work here is that whatever we discover through human agency has already been stated, in some measure, in the Bible. In discussing Proverbs 15:30 (A cheerful look brings joy to the heart, and good news gives health to the bones), Daniel offers the following:

"There might be more literal truth to that than we would guess. If you've listened to or read anything by Ken Hamm, a big creationist, one of his statements is that he'll stick with God's Word over science every time because science continues to change. That, as time goes on, in a lot of ways it corroborates what Scripture already says. I think it's interesting how there can be these little nuggets that, you read through this and this is a proverbial saying by a long dead king. And, that just might prove out to be true scientifically at a level the likes of which we might have never thought. That, this might be a much more profound or true statement, a statement true on more levels than we might have anticipated."

Finally, the ideological principle of the Bible's timelessness also prompts a view of the Bible as a sourcebook and guide for everyday life, and every situation. If sinfulness has not changed, neither have the predicaments people put themselves in. In equal measure, the wisdom of the Bible is the consummate guide for managing these problems. Aaron from the LCMS men and Rick from the Monday group both suggested – in the midst of discussions about experience being the best teacher – that the purpose of the Bible is for every generation to learn anew from the mistakes it portrays. A sizeable segment of the Evangelical literature industry includes books aligning Biblical texts as responses to life's 'most difficult questions.' Titles such as God's Answers to Life's Difficult Questions, The Greatest Questions of the Bible and of Life, and Checklist for Life for Men (and for Women) restate that Biblical answers persist in their validity because the message of the Bible is an unwavering Truth.

Biblical Relevance

3) The Bible speaks to you in new and different ways every time you read from it.

More so than the other five principles, this expectation provided a constant refrain throughout every group's discourse. There is a certainty that the Bible has the unique capacity to always be relevant and appropriate in the life of the individual, and in ways that keep pace with the uncertainty of life. There is the expectation when reading Biblical texts that the experience will not be the same as last time, nor will the consequences of doing so. The reader is assured of coming away with something new to consider or apply, something not present in previous readings.

No group embodied this more than the Prayer Circle. Indeed, this principle structured their method of organizing meetings. In his instructions, Darren challenged the group to read the same text every day during the intervening two weeks before the next meeting. Certainly, not everyone did this each meeting, while some did achieve the ideal. But, the reality of what individual readers did matters less than the expectation motivating the challenge itself. The significance lies in the assumption that the same Biblical text can be read every day for 14 days and, upon the final reading, still be the bearer of new insight. As the group began their discussion of Philippians, the third Biblical text they read together, Marcy prefaces her sharing:

Marcy: There's so much in each chapter, I feel like if I read straight through that I'm almost reading it too lightly.

Van: You think you're missing something.

Marcy: Oh yeah. Beverly: Mhmm.

Marcy: I tried to read one chapter a day, and I underlined and starred. And, then, you know, the next time you go through you find some more and you go, "Well, that's good too."

Jennifer: Yeah, I know. I found stuff over and over again too.

Prayer Circle members, and participants from other groups, sometimes explained this dynamic as a consequence of accumulated background knowledge and/or life experience. In short, the effect of the text changes because the reader has changed (cf., Fish 1980). The explanation heard more often, at times in tandem with the above, was the nature of the text itself as the Word of God. And, while the message and purpose of the Word is unchanging as described in the second principle, it carries this concomitant ability to speak in new ways.

This principle reinforces two aspects of Bible reading. First, it ensures continual use. If the Bible holds the ever-present potential to be new, then it can never be old, antiquated, or exhausted. No matter one's age, maturity in faith, or experience with the Bible he or she can always expect something new to be (re)discovered. This helps explain why the same group can meet for decades, returning in various ways to the same texts. This principle also highlights the primary purpose of engaging the Bible in the first place. Individuals read their Bibles every day, sometimes on multiple occasions, sometimes for hours at a time, with the expectation of being taught, inspired, and convicted. They do not do so for the strict purpose of deriving textual meaning. This does occur, but it is secondary to believers' hopes of being affected in some way. Chapter 3 deals at length with how this balance unfolds in group discourse.

4) The Bible speaks directly to lives and situations. Similar to the third principle, this is rooted in the ongoing relevance of the Bible to the lives of readers. Not only does the Bible hold the potential to be different each reading, it holds the potential to be precise in its application. There is no contradiction here for believers because the same process is at work: the Holy Spirit is revealing what is needed when it is needed from the

absolute truth that is the Word of God. Individuals and collectivities (e.g., small groups, congregations) expect Biblical texts to have application to their own, personal lives and particular circumstances. The application is not vague, but specific; not general, but amazingly exact in how it aligns with readers' lives. While discussing James 4:17 – Anyone, then, who knows the good he ought to do and doesn't do it, sins – Vincent from the McGee Men relates a story of his own:

"I went into an Office Max a couple of weeks ago. And, right there in the foyer there was a wooden filing cabinet for ten dollars. Normally, it's eight dollars. This is one of the things you put together. Well, it was already put together. It was the last one and they were getting rid of it. I said to myself, 'I don't need a cabinet, but maybe I could use it some time. Ten dollars is really good.' It was plain, but it looked decent. Weighed a ton. So, I bought it. When I was checking out this lady comes over and says, [[Southern accent]] 'Oh, where'd you get that. I would like one of those.' I was impressed to say, 'I don't need it, you can have it. You can buy it. Here. Take it off me.' But, I didn't. I was impressed to, but I didn't. I sinned. And, one reason I know I sinned is I had a chance to give this thing somebody who wanted to use it right now. And, what was I gonna do with it? I don't know. I was gonna leave it in a box. And, it's still sitting there in a box. It's a burden. It weighs a ton. I left that store. I was looking all over the parking lot for her."

Group participants also found Biblical evidence for this principle. Hebrews 4:12 is a commonly cited proof-text: For the word of God is living and active. Sharper than any double-edged sword, it penetrates even to dividing soul and spirit, joints and marrow; it judges the thoughts and attitudes of the heart. The Bible judges not only the thoughts and attitudes of some hearts, but of your heart, of my heart, of all our hearts. In his analysis of Evangelical Biblicism, Malley (2004) identifies this same principle at work in the Baptist congregation with whom he conducted research. He described this as a "dual context communication event" that ensures delivery of "the expected relevance" (ibid: 107). The initial context is a typical hermeneutic practice. Readers recognize the humanity of Biblical authors, "and that the human author's intent fixed the Bible's meaning once and for all" (ibid: 111). One of his informants used Song of Solomon as an

example. He considered it a love song because interpreters generally agree that was the author's original intention. However, the second context is where this principle emerges. In addition to the intended meaning, "God, in the person of the Holy Spirit, may speak to an individual, in the particular circumstances of that individual's life, through some passage of the Bible" (ibid: 111). Malley notes that this dual structure renders critiques of 'taking verses out of context' erroneous. It is quite impossible to read out of context when the life of the reader is the context.

The assumption that the Bible is suited to individual circumstances helps explain a variety of products and practices. One genre of Study Bibles – Life Application Bibles – provides an example. These appear in most every mass published translation. The aim of footnotes and study questions is to prompt readers to connect individual texts with the details of their life. Devotional materials, which are widespread in congregational life, are also a product of this principle. Organized around daily prayers and Bible readings, these materials are organized around the practice of personal, intimate applications of Biblical texts. "Bible dipping," the practice of opening the Bible randomly (though, prayerfully), hoping to be edified, is rooted in this principle as well (Malley 2004). Including Bible verses with gifts, and buying Christian greeting cards containing verses, provides further examples. All of these are made possible by this principle of direct application.

Biblical Textuality

5) The Bible tells a coherent story from beginning to end. The Bible is not understood as a group of disparate texts, or as a single book lacking a unifying theme.

The Bible is understood as a collection of texts that tell a coherent story through their intertextual relations. The story the Bible tells is about the nature of God and humanity,

and God's plan for the unfolding of time. Individual verses, chapters, and books are read within the context of this unifying narrative. Knowing what the Bible is about provides an interpretive frame in which to position and discuss any given Biblical text. In my first recording with the McGee Men, they pause for a moment on the meaning of James 1:22 – Do not merely listen to the word, and so deceive yourselves. Do what it says:

Joel: What are you asked to do? I'm reminded of the verse in Matthew where it says, Not all who say to me, 'Lord, Lord' will enter the Kingdom of Heaven, but only those who do the will of my Father. And he says, many will come and preach in my name and they will cast out demons and they will do mighty miracles, and I will say to them, 'Depart from me I never knew you'. So, maybe those aren't the actions that he wants.

Dennis: Yeah. And you wonder what he's saying? What's the basis of that? It's the motivation that counts. Where does your action come from? What's the basis for your action?

Joel: And, I always think of the so-called Great White Throne Judgment. Remember when the separation of the sheep and the goats come. What's the basis? Did you feed the hungry? Did you give water to the thirsty? Did you visit the prisoner? Did you help the sick? "Come on in." I mean, it's so simple in a way.

Ryan: So simple, yet so difficult.

Joel: But, it's like Jesus says, the loving the Lord your God with all your heart and all your mind, your neighbor as yourself is the fulfillment of all the Law and all the ages. So, love is, where does John say, God is Love. And I said, "Oh, come on." That's his definition?

Frederic: Well, he went so far as to say Love is the fulfilling of the Law, yeah.

Dennis: Maybe love is the basis out of which our action is supposed to come.

Joel: God, grace, and like Micah says, and Jesus, both, you neglect justice, mercy, and the love of God, walking humbly with God.

Frederic: Does it say, it says love kindness, or is it do kindness?

Joel: I think it's love, well I think *mercy* is in the Bible, but it can be kindness. But, it's *love mercy*. Don't just do it, *LOVE* to do it.

Frederic: Well, it's kind of the apex of the book, isn't? In other words, if you were to pick any one passage from the letter of James that's representative of what he's saying, this could be one of them.

Dennis: Sure. Joel: Yeah.

The men begin with a question about what is meant by doing what the word says. On Joel's cue, they use other scriptural texts to articulate the Biblical theme of love, and thus the proper way to read this verse from James. This particular interaction is reminiscent of the example John Bartkowski (1996) gives in his analysis of two conservative Christian

writers, where one author filtered his Bible interpretation through an understanding of the Bible's "overall message" as one of "unconditional love."

In the groups I studied, one of the most consistent strategies for succinctly characterizing the nature of God is to posit a division between the Old and New Testaments. In their final meeting for studying Proverbs, the LCMS Men identified having a "fear of the Lord" as a theme throughout the book. Daniel asked the group how they understood this concept of fear:

Bill: I think if I were to only read the Old Testament, I would interpret fear of the Lord as real fear. But, having read that and then read the New Testament. With that fear there comes more respect and understanding of love. The Old Testament, there's a lot of God lays the rules down and here's a story of how nobody could really stick with the rules. And, the New Testament is the story of Christ and His resurrection. So, it kind of changes your perspective on God of being dominating to a loving God.

Daniel: Do we need to remember more of the Old Testament than we do? **Bill:** I think if we can remember the Old Testament as putting the New Testament in perspective. This is why the New Testament occurred, because of what happened in the Old Testament.

Aaron: I happened to be reading the prophet Joel. He wrote about 780 or so BC, some time in there. He used the occasion of the locusts coming like an army, wiping out everything they had, saying this is God telling us to wake up. And, thinking about what happened in New Orleans. And, I know we foo-foo that kind of stuff. But, I think God's given us a wake-up call.

Daniel: Do we sometimes fall too much into all New Testament kind of thinking? Do we lose sight of, "Now, don't mess with God?" He's a God of grace and compassion, but don't mess with him. Can we even think in those terms? Ought we think in those terms? Andrew: I don't want to think in those terms. I don't want to think in these terms that Bill's describing, because I find a lot of grace in the Old Testament. I find a lot of Jesus. I just flipped back to Psalm 116, because it was close. Graciousness, glory and righteous, yes, our God is merciful. The Lord preserves the simple. I was brought low and he saved me. And, so, to somehow suggest that there must be a balance between Old and New, I don't think so. I think there's a balance, perhaps, between God's grace on one hand and his judgment on the other. And, there will be judgment, I understand. I also think that the fear aspect involves understanding what God's character is and what His capabilities are and what consequences might be. But, if we get too caught up in this terror business, is that works-righteousness? Are we, at that point, saying, "Oh, boy, I better be good because otherwise I'm gonna get hammered." And, that's not the way it works.

Aaron: I think I agree with both of you. But, I like what Bill said earlier. Christ shows

Aaron: I think I agree with both of you. But, I like what Bill said earlier. Christ shows God's love in a way that, even though the Old Testament is the love of God presented, it doesn't present itself like Christ dying on the cross.

Daniel: The greatest clarity of [God's love] is seen in Christ.

The Old Testament reveals God's justice; what happens when he deals with people in wrath and judgment. The New Testament reveals God's love in the person of Jesus Christ; what happens when he deals with people in grace and mercy. When this straw man gets articulated in group discourse, as Bill does above, a predictable dissent follows. Opponents, like Andrew, point to instances of love and grace in the Old Testament, and of judgment and stridency in the New. In the end, though, the notion of the two canons working together to present a complete picture of God – both wrath and mercy – is upheld.

Taken as a whole, the Bible tells a coherent story about the nature of God: loving and wrathful, creative and destructive, forgiving and judgmental. The publication of <u>The Story</u> (2005) exemplifies what can emerge from this principle. Based on the TNIV translation, <u>The Story</u> replaces the traditional book-chapter-verse organization of the Bible with a novel-like chapter structure. The book's description on the back cover reads like this:

"At its most basic level, Christianity is a story about God and the remarkable lengths he goes to in order to rescue lost and hurting people. *The Story* gives you just that – the story of Scripture. Condensed into thirty accessible chapters, it reads more like a novel than your typical religious text. And like any good story, *The Story* is filled with intrigue, drama, conflict, romance, and redemption."

The Bible is also understood to reveal God's plan for history. For Christians, this revolves around the person of Jesus Christ. Jesus is the central figure of the Bible, and the reason for his life is the focal point of the narrative. In this sense, the Bible is also read as a book of prophecy. Predictions, foreshadowings, and foretellings of Jesus are pointed to throughout the Old Testament. The birth of Jesus is understood as a turning point in history, a fundamental alteration in how God can be known by human beings. As a depiction of history's unfolding, Jesus' presence in the Old Testament is crucial to its

continuity. The New Testament is also part prophecy. The birth of Jesus signifies both a new beginning and the beginning of an end. It marks the beginning of the final age before the end of time. Different understandings of the nature of eschatology abound in Protestant theology, but there is agreement in the orthodox tradition that time will eventually have an end. New Testament prophecy, then, includes a description of how to recognize the arrival of the end. For some, the Bible's continuity is even more precise in its representation of history's unfolding. Genesis tells of creation, the beginning of all things, and Revelation describes the Second Coming, the end of all things. For others, the Bible is not as exact, but still contains the message of who God is and what this means for the purpose of the human life.

This sense of coherence and continuity extends to an appreciation of Biblical consistency for consistency's sake. The authority of the Bible is supported by its uniformity. The textual tool of a concordance is one manifestation of this. Biblical concordances offer readers lexical and thematic trails that run throughout scripture; usually in the form of page number listings beside individual names, words, and phrases. The LCMS women provide an ethnographic example of how this can play out in group discourse. The following interaction began with a reading of Colossians 2:6-7, which includes the phrase, *rooted and built up in [Jesus]*. Part of Sandy's preparation involved the use of a concordance:

Sandy: What psalm do you think of when you think of rootedness?

[[No response from group]]

Sandy: Psalm 1. I think it's kind of a special meaning. Beautiful imagery in this text.

Cindy, would you please read Psalm 1?

Cindy: Blessed is the man who does not walk in the counsel of the wicked or stand in the way of sinners or sit in the seat of mockers. But his delight is in the law of the Lord, and on his law he meditates day and night. He is like a tree planted by streams of water, which yields its fruit in season and whose leaf does not wither. Whatever he does prospers. Not so the wicked! They are like chaff that the wind blows away. Therefore the

wicked will not stand in the judgment, not sinners in the assembly of the righteous, but the way of the wicked will perish.

Sandy: Okay. What is this psalm teaching us about rootedness?

Jane: God always takes care of us.

Rebecca: Fullness of life.

Margie: That God is the living water.

Sandy: Yeah. It's interesting; sometimes you can almost encapsulate the entire meaning of the Bible by tracing certain elements of style, certain analogies, certain word pictures. How about Psalm 80:8-11? Margie.

Margie: You brought a vine out of Egypt; you drove out the nations and planted it. You cleared the ground for it, and it took root and filled the land. The mountains were covered with its shade, the mighty cedars with its branches. It sent out its boughs to the Sea, its shoots as far as the River.

Sandy: Okay. To what time in the life of the Hebrew nation is this referring?

Margie: The exodus out of Egypt.

Joy: God also brought Jesus, as a baby, out of Egypt. When they fled there because of Herod.

Sandy: Isn't the Bible an amazing thing? Just an amazing book. Proverbs 12:3 and 12:12. Frita.

Frita: A man cannot be established through wickedness, but the righteous cannot be uprooted. The wicked desire the plunder of evil men, but the root of the righteous flourishes.

Sandy: Very consistent imagery here. Rebecca, could you read Isaiah 5:24?

Rebecca: Therefore, as tongues of fire lick up straw and as dry grass sinks down in the flames, so

their roots will decay and their flowers blow away like dust; for they have rejected the Law of the Lord Almighty and spurned the word of the Holy One of Israel.

Sandy: Okay. *Decaying roots*, how does that speak to us?

Joy: It says you have rejected the instruction of the Lord. So, the decay depends upon us, if we stay in the Word. Because if you've rejected the instruction and despised the Word, you dry up.

Sandy: Well put. And, Joy, could you read Matthew 13:6 for us?

Joy: But when the sun came up, the plants were scorched, and they withered because they had no root.

Sandy: Very consistent. Revelation 22:16. Elinor.

Elinor: "I, Jesus, have sent my angel to give you this testimony for the churches. I am the Root and the Offspring of David, and the bright Morning Star.

Sandy: ...Okay. Let's look at Ezekial 47. Joy.

Joy: [[Reads Ezekial 47:1-12]] Sounds like heaven.

Sandy: Isn't that a beautiful passage? It truly is. And, one more, Revelation 22:1. James. Me: Then the angel showed me the river of the water of life, as clear as crystal, flowing

from the throne of God and of the Lamb down the middle of the great city street.

Sandy: What a beautiful picture. I guess the only point I've been trying to make here is the beauty itself of the scripture as it develops itself through this imagery; especially in this imagery of being rooted, being built up.

As Sandy says repeatedly, part of the Bible's power lies in its continuity. She uses the example of how a particular spiritual theme – rootedness – can reappear in various texts, demonstrating the coherence of the Bible.

This ideology around the Bible's textuality also places certain traits, such as contradiction, in opposition to the inherent nature of the Bible. By its very nature, the Bible cannot be at odds with itself. Its texts cannot be conflicting, or contrary with one another. Any suggestion of the sort is dismissed in favor of the Bible's uniformity. This continuity also gives rise to the interpretive principle that the Bible is 'its own best interpreter.' Because the entire Bible is the inspired Word of God, confusion around one text is best resolved by consulting another Biblical text. (See chapter 5, page 169, for an example of these two outcomes).

6) The Bible is written perfectly. How readers understand the language of the Bible is also woven into the textual ideology. It is viewed as amazingly precise and accurate in how it captures thoughts, emotions, and dialogue. This is less a commentary on the structural and formal qualities of Biblical texts as it is on their aesthetic potential and appeal. When participants allude to the Bible's "perfection," they are making a statement about the beauty of Biblical language and its close alignment with experience. This principle was articulated only a third as much as the others, but it was integrated occasionally in small group discourse. The LCMS men provide the clearest articulation of this in their reading of Proverbs.

The topic for the day's discussion was Proverbs 8, 9, and 10. In his transition to chapter ten, Daniel listed a few of his favorite verses. The last one he read was Proverbs 10:26: As vinegar to the teeth and smoke to the eyes, so is a sluggard to those who send

him. He then asked the group which proverbs they "underlined" or thought were "interesting." Andrew was among the first to respond to Daniel's question. He is a core participant in the group, best known for his tendency to introduce new layers of complexity to any topic. Andrew is a retired lawyer, a background made obvious by his habit of homing in on individual words and phrases for critique, clarification, and exposition. He said this about Proverbs 10:26:

"What always strikes me is the precision of the language. Just the way that things are framed. I mean, this is pretty well thought out. Just the verse you were talking about. Vinegar to the teeth. It's not something that's going to hurt you. But, it's just irritating. And, it's like smoke in your eyes. It's just irritating. It doesn't harm you. But, it's IRRITATING when that happens, when you send a lazy guy out and it doesn't get done. It's just IRRITATING. But, my point is that the language is just right, it's always on the mark. It's just perfectly descriptive of what's going on. And, I don't think that's an accident."

Andrew's appreciation for the language of scripture is indicative of a sentiment I observed in many of the project groups. Participants often remarked on the perfection of proverbs, psalms, parables, prayers, and a host of other Biblical genres. Biblical language is not approached with an attitude of 'how to say it better,' but one of 'it could not have been said better.'

This may strike the reader as an odd principle given that there are dozens of different English Bible translations readily available. Does the suggestion that the language of the Bible is perfect refer to a particular translation? If not, how can all translations be perfect, particularly when they often read very differently? The variety of translations does not pose a serious challenge to this principle because they are treated as resources, not as mutually exclusive alternatives. Multiple translations were consulted less for proof of one translation's accuracy over another, but more to find the best expression for particular verses. The McGee Men were particularly apt to do this, as

demonstrated in their discussion of James 4:11. Their initial reading came from McGee's study guide, which uses the King James Version²⁸. Dennis, the group facilitator, then read the text in the New English Translation²⁹. Following several minutes of discussion, Vincent reads from both The Message and the New Living Translation³⁰. Throughout this exchange, the validity of the translations was never broached. Instead, the focus remained on what the text of James 4:11 is stating, using the various translations to help this effort.

Translations are not placed in competition, but are used cooperatively to locate the best reading possible. Individuals will argue for the supremacy of certain translations, as well as for the illegitimacy of others. But, this was never the policy of groups. In the context of group discourse, translations are treated as resources to be drawn on as necessary, not as choices to be ranked. This view of translations allows the understanding of Biblical language as perfect to persist.

Textual Ideology: Responses to Four Probable Questions

1) Taken together, these six principles form the core of the textual ideology surrounding the Bible in the groups I studied. They are a system of ideas, beliefs, attitudes, assumptions, and expectations about the authority, relevance, and textuality of scripture. And, like any system, these principles are not islands unto themselves. Indeed,

judge.

29 Do not speak against one another, brothers and sisters. He who speaks against a fellow believer or judges a fellow believer speaks against the law and judges the law. But if you judge the law, you are not a doer of the law but its judge.

²⁸ Speak not evil one of another, brethren. He that speaketh evil of his brother, and judgeth his brother, speaketh evil of the law, and judgeth the law: but if thou judge the law, thou art not a doer of the law, but a judge.

Message: Don't bad-mouth each other, friends. It's God's Word, his Message, his Royal Rule, that takes a beating in that kind of talk. You're supposed to be honoring the Message, not writing graffiti all over it. New Living: Don't speak evil against each other, dear brothers and sisters. If you criticize and judge each other, then you are criticizing and judging God's law. But your job is to obey the law, not to judge whether it applies to you.

you may have already asked yourselves, don't these principles rely on each other to make sense? Aren't they interconnected?

These principles are indeed mutually dependent. For example, the authority of the Bible as the Word of God underleis the understanding of Biblical language as perfect; in fact, only the Word of God could attain such an expectation. It is because of its divinely inspired quality that the language is taken to be so accurate, so precise, and so descriptive. The two principles of relevance are closely linked. Because the Bible can speak in new ways with each reading, the Bible also speaks in ways specific to individual circumstances. If its relevance were restricted to a given period, it would cease to have the quality of ongoing personal application. The assumptions of relevance are also tied to the Bible's status as unchanging. The Bible continues to be relevant for modern believers, and promises to be for succeeding generations, because it contains eternally valid information; for instance, the nature of humanity and the answers to life's fundamental questions. The textual ideology surrounding the Bible is very much a system, with principles that are both distinct and inter-related. I suspect that any potential additions to this system could not be diametrically opposed to the existing principles. If the principles exist in a supportive relationship, there is no room for contrary notions that might threaten the stability of the relations.

2) The textual ideology surrounding the Bible is a historical phenomenon, like any other belief system. What, then, can be said about the genealogy of these six principles?

Aspects of this ideology extend to the Biblical period, as evidenced by the use of scripture to articulate various principles. Principles of this textual ideology also have

roots in the earliest expressions of Protestantism. Viewing the Bible as is its own best interpreter is a distinctly Reformation concept, most apparent in the writings of Martin Luther (Hagen, et al. 1985). Wesley Kort (1996), however, assigns the first defined theory of Bible reading to John Calvin. The current form of this textual ideology owes at least two debts to the way Calvin approached scripture. First, Calvin placed aims such as the accumulation of knowledge secondary to reading the Bible for its power for personal conviction. The ability of scripture to affect the life of the reader takes precedence over the strict search for textual meaning or becoming especially learned of Biblical texts.

Second, Calvin widened the sphere of the Bible's application. He insisted that the knowledge of salvation and an understanding of the texts themselves were only the start of the Bible's value. He emphasized the Bible's relation to the "whole of life" (ibid: 30). The Wesleyan tradition picks up this notion, emphasizing that the Bible must be confirmed in personal experience (Metts 1995). The modern practice of linking Biblical texts to every detail of one's life testifies to the lasting significance of Biblical relevancy.

The role of the Bible in the history of American Christianity also impacts the current form of this textual ideology. Assumptions of plenary inspiration and absolute authority were being championed by English Puritans when they arrived in the 17th century, and have been reproduced through Protestant institutions, including:

Anabaptists, Wesleyans, Lutherans, Calvinists, Pentecostals, Fundamentalists, and Evangelicals (Noll 1992). The publication of <u>The Fundamentals</u> between 1910 and 1915, a series of pamphlets defending the complete authority of the Bible, are one modern milepost in reproducing the centrality of these doctrines (ibid). The popularity of related

notions, such as Biblical inerrancy, were still in their infancy when the pamphlets were published, coming into vogue only after the Civil War (Noll 1982).

Understanding the Bible as a unified narrative has been a consistent dividing line among American Protestants. Peter Thuesen (1999) observes how this issue of narrative coherence posed continual translation difficulties for the ecumenical RSV translation committee. Language ideologies concerning the Bible's perfection are rich in the King James tradition (McGrath 2002), and have become increasingly widespread with the explosion of English language translations in America beginning with the RSV in 1952 (Thuesen 1999).

3) An essential question confronting any textual ideology, or any belief system, is the question of sharedness. Clearly, everyone I encountered in my fieldwork did not uphold all of the principles in identical ways. What, then, can be said about the extent to which this textual ideology is a shared set of expectations? After all, statements on the Bible's authority, relevance, and textuality have ignited schisms at the denominational, congregational, and small group levels for centuries (e.g., Noll 1992). Although some conflict did arise among group participants around particular principles, the overall level of agreement around these principles remained remarkably high.

Three sources of evidence support this claim: group discourse explicitly about the textual ideology, informal interviews with group members, and questionnaire results. I coded transcripts and field notes to identify when textual ideology statements were made, and who made them, in each group. Verbal and non-verbal reactions to these statements were also analyzed for agreement, disagreement, and ambiguity. I also coded fieldnotes of informal interviews conducted with group members before and after meetings for these

types of statements. These data were certainly not as numerous or rich as those from discussion transcripts. But, it did provide an opportunity for the voices of marginal participants, and of those not comfortable voicing dissent publicly, to be heard. Lastly, the final section of the research questionnaire (cf., Appendix 2) consisted of a 42-statement list about the Bible, 18 of which directly reflected selected principles³¹.

Triangulation of these data supports three claims about the sharedness of the textual ideology. 1) An overwhelming majority of participants upheld these six principles in all groups. And, in no case was the same individual the dissenter for more than two principles. 2) The six principles are unequal in both their presence and the extent to which they are disputed. Principles of authority and relevance were far more likely to be articulated. However, Biblical relevance and textuality were rarely the location of disagreement; unlike Biblical authority, which marshaled a wider range of opinions. 3) Most important, the textual ideology was prominent enough (on both the individual and group levels) to create a climate of recognition. Everyone, whether they agreed completely or not, was aware of the principles and aware that their participation in discourse took place against this ideological backdrop. And, in several cases, where there was little or no disagreement, the textual ideology functioned as a taken-for-granted reality within the group. This was the case in Malley's (2004) Baptist congregation, where the principle of Biblical authority was required knowledge for appropriate participation in the group.

Disagreement, dispute, and tension were all still observed despite this high level of sharedness. Three of the project groups had recurring disagreements in their

³¹ These statements were presented to group participants on a Likert-type scale: strongly disagree, disagree, no opinion, agree, and strongly agree.

discussions. Disputes arose over various principles of the textual ideology in the McGee Men, the LCMS Women, and the LCMS Men. As described in the previous chapter, these disagreements were most prominent with the McGee Men. In the following chapters, I present analyses from all three groups.

There was almost uniform agreement on the principles of relevance and textuality. One of two things happened when those principles arose in group discourse. The notion was affirmed and discussion moved to the next topic. Or, the notion was affirmed at length by numerous examples from numerous participants. The two principles of Biblical authority presented a different scenario. Differences arose over the degree to which the Bible is authoritative. For example, the questionnaire statements receiving the highest levels of agreement and the highest levels of dissent addressed principles of authority. Consider the responses to these six statements:

Principle	<u>Statement</u>	Disagree	No Opinion	Agree
1	I believe the Bible is	0%	0%	100%
	divinely inspired.			
1	I believe God guided the	0%	6%	94%
	formation of the Bible	ĺ		
	as it now appears.			
1	I believe the Bible is	0%	3%	97%
	authoritative over any			
	other text.			
1	I believe the Bible is the	32%	10%	58%
	only text inspired by			
	God.			
1	I believe the Bible	22%	13%	65%
	contains NO			
	inconsistencies.			
2	I believe parts of the	61%	6%	33%
	Bible are NOT			
	applicable in today's]		
	world.		<u> </u>	

Table 5: Questionnaire data for textual ideology (1)

The agreement around the first three statements is astounding. However, the second set of statements reveals a marked tension surrounding the nuances of Biblical authority. This contrast suggests that participants disagree with each other most over the principles of authority.

Finally, individuals expressed ambiguity toward certain principles. During informal interviews, individuals described an assurance of the inerrancy of original texts, but also an acceptance of possible alterations via translation and manuscript transmission. In group discourse, individuals found occasion to express support for the Bible as a coherent narrative, but also confusion as to why some events seemed at odds with the storyline. In certain questionnaire responses, individuals agreed and disagreed with statements upholding principles of authority and relevance. In short, just as group discourse can be the site of conflicting ideologies, so can the mind and heart of an individual.

4) One final question may be raised about my use of the category of "the Bible." Despite the use of this gloss by participants, the historical fact remains that the Bible is not one text, but a collection of many texts. Moreover, it is not a collection of equally used texts. For instance, any given reader in my sample would be more familiar with the Sermon on the Mount than they are with the genealogies of Leviticus. In regard to the textual ideology, do these principles apply to the whole of scripture or only selected texts?

Unlike the other questions, the answer to this one is quite simple. The textual ideology surrounds the entire Bible. It is not restricted to commonly cited, commonly debated, or well-known texts. Kort's analysis of Calvin's theory of reading is again

useful: "Although the knowledge of who God is and that God is one who saves as well as one who creates is clearer in some parts of the Bible than in others, it is available anywhere and everywhere [in the Bible]" (1996: 30). This describes well the sentiment among the project groups. Selected Biblical texts were associated with special memories, admired for poetic qualities, seen as indexical for whole doctrines, and favored for use in different settings. Nevertheless, no Biblical text was exempt from inclusion in the textual ideology.

One clear point of separation was between the Old and New Testaments. I described above how they are understood as a composite portrait of God's character. However, readers did react differently to the two canons in some other ways. The Old Testament was considered "more distant," "harder to understand," and "less significant for evangelizing." In regard to Biblical authority, questionnaire responses reveal a dissonance between the two canons in terms of a willingness to disagree. Table 6 shows the results for three statements presented in sequence:

<u>Statement</u>	Disagree	No Opinion	Agree
I feel free to disagree with things I read in the Old	68%	13%	19%
Testament.			
I feel free to disagree with things I read in the Gospels.	78%	19%	3%
I feel free to disagree with things I read in the rest of the	78%	19%	3%
New Testament.			

Table 6: Questionnaire data for textual ideology (2)

The difference needs little explanation. Participants do not feel the same burden of adherence when reading the Old Testament. Even still, specific texts in the Old Testament (e.g., Job, Psalms, Proverbs, and Isaiah) are revered and treated on par with the whole of the New Testament.

This chapter has analyzed the textual ideology surrounding the Bible. This ideology is a constellation of beliefs, ideas, assumptions, and expectations that guide readers' interaction with Biblical texts. My goal has been to sketch the outlines of the ideology itself, focusing only on the first half of the interaction. This textual ideology does indeed exist in relationship with a variety of textual and discursive practices. The remainder of this dissertation is devoted to completing a sketch of this interaction. In the next three chapters, I demonstrate how this ideology works in various forms of practice. Chapter 3 considers practices of reading and interpretation. Chapters 4 and 5 consider the practice of incorporating the Bible into group discourse. In chapter 4, this is observed in the interpretive resources members invoke to engage in discussion. In chapter 5, it is observed in the practice of recontextualizing Biblical texts in the midst of group discourse. Through these analyses, I argue for an ongoing, dynamic and mutually reinforcing relationship between textual ideology and discursive practice.

Chapter 3: Interpreting the Bible in Small Group Discourse

"If you really keep the royal law found in Scripture, 'Love your neighbor as yourself,' you are doing right. But if you show favoritism, you sin and are convicted by the law as lawbreakers. For whoever keeps the whole law and yet stumbles at just one point is guilty of breaking all of it." (James 2:8-10, NIV)

The McGee Men began a mid-February meeting by reading these verses. Five of the group participants were there that morning. The facilitator for the day, Vincent, started the discussion: "Joel. If you show favoritism, you sin. Do you have a sense of what that is?" Joel responds: "Yeah, that if you judge people in terms of, 'he's okay, you're not okay." Vincent agrees, and elaborates. Frederic is next to join the discussion: "Well, my reaction is that James is very harsh here. I think it's not so much a metaphor as much as a, you call it an exaggeration. It's a literary device. You exaggerate in your writing to make a point." He continues, explaining this reaction to the text, and several others join in considering the possibility. Frederic then moves the discussion forward: "I think the key thought in this passage is two words, whole law. I think that's the one maybe we should debate." And, they do, at length.

Roland makes his first entrée into the discussion following a series of back-andforth comments about the meaning of whole law. He returns to Frederic's original
assertion: "I want to push, Frederic, do you think it's a good technique to do it the way

James is doing it here? Do you think it misleads people?" This prompts Fredric to discuss
at greater length the choice of rhetorical technique, as well as the veracity of the text. Joel
changes the direction of the discussion by posing a new line of questioning: "Wasn't

James kind of in between the Gospel and the Law? He goes back and forth, because he's
blended. They had a sense of grace, but not in Paul's fullness." Joel attempts to present a

historical explanation for why James reads like it does. The group considers this briefly, but returns to debating the meaning of whole law.

Roland disrupts this again by turning the question of meaning into a question of application, summarizing the position of several participants: "So, the conclusion is that in God's eyesight, if you commit adultery, you don't have to be as uptight as you would if you had murdered somebody. Is that what you mean?" Joel follows Roland's attempt to apply the verse with his own, cross-cultural, example: "I wonder how they look at this where somebody has a number of wives? You know, like some Arab and African countries, and even here in America with the Mormons." Ultimately, Frederic concludes the discussion, and turns to the next reading: "Well, the rest of the chapter kind of bails us out, I think." Vincent takes the cue and asks Frederic to begin reading the next set of verses.

This chapter is about how the Bible is read and discussed in small groups. I explore the styles participants use to interpret Biblical texts, how these styles unfold discursively, and how they are positioned in relation to one another. The interaction I described above illustrates how groups go about reading and interpreting Biblical texts. This particular exchange gives a glimpse of how complex and how dynamic this aspect of group discourse can be.

My descriptions and analyses are rooted in two explanatory frames: textual ideology and interpretive community. The interpretive styles participants employ carry the imprint of the textual ideology principles explained in chapter 2. Moreover, the ways in which interpretive styles are employed demonstrate differing orientations to these

principles. Throughout this chapter I show how the role of Bible interpretation in group discourse is inextricably linked to the set of assumptions and expectations believers hold about the Bible.

I conclude that small groups can be understood as "interpretive communities." This concept was developed in literary analysis, but there is great potential for its ethnographic application (cf., Forstorp 1990, Radway 1984, Svensson 1990). It emerged from reader response theory, particularly the work of Stanley Fish (1980), who stressed the active role of the reader in establishing meaning, as opposed to an emphasis on meaning being already present in the textual artifact. Fish insisted that this was not a completely subjective phenomenon, and that interpretations arise from communities of readers who share assumptions about texts, methods of reading, and strategies for interpretation (ibid). I argue that the shared interpretive styles, reliance on textual ideology principles, and the remarkably similar interpretive processes of Bible study discourse identify small groups as a form of interpretive community. First, however, I will demonstrate how the discourse surrounding Bible interpretation unfolds in small groups.

Five Case Study Groups

The findings of this chapter are based on the five project groups who consistently read and studied Biblical texts. The remaining project groups focused on non-scriptural texts or topical studies. (See Table 7 for the five case studies and their study topics.)

Group	Biblical Texts Studied	Recordings	Study Guide
McGee Men	James	12	McGee (1955)
Tuesday Men	Acts 1-8	23	LifeGuide (InterVarsity Press)
LCMS Men	Proverbs	12	None
Prayer Circle	James, Ephesians,	14	None
	Philippians, Colossians, 2		

	Timothy, Habakkuk, Psalm		
	119, Hebrews, Acts 1-15		
Monday	Hebrews, 1 John, 2 John, 3	8	None
Group	John		

Table 7: Study topics for five project groups

The first three sections of this chapter are devoted to the three interpretive styles I identified as operative in group discourse: application, establishing meaning, and explanation. I emphasize the hermeneutic activity accomplished by each style; the interpretive process of each style; the relation of each style to group interactive features; and the interaction of each style with principles of the textual ideology surrounding the Bible.

Applying Biblical Texts

Participants employed the application style to establish a connection between a Biblical text and some aspect of their life or surrounding world. This style raises the question, 'what does this Biblical text mean to...?' instead of 'what does this Biblical text mean?' This style assumes that the Bible is applicable, relevant, and effective.

Application was by far the most predominant style in small group discourse. Participants read the Bible primarily for comfort, conviction, instruction, devotion, worship, and a host of other motivations not connected with an attempt to attribute meaning to a text.

This phenomenon has been observed in other ethnographic and historical contexts. Jeff Todd Titon (1988) described this style as "tropology" in his ethnographic study of a Southern Baptist congregation in Appalachia. Tropology views the events and personages of the Bible as timeless, always germane to new generations of believers. Per-Anders Forstorp (1990) identified application as the primary form of interpretive act performed among congregants of a Swedish Fundamentalist church. Jody Davie, in her

ethnography of a New England Presbyterian women's Bible study, observed a personalization of the Bible in the individual spiritualities of group members: "[It] is used to facilitate examination of the personal journey, rather than being a closed entity, a mysterious storybook, left idolized and undisturbed in the religious process" (1995: 54). Brian Malley (2004) argued that Evangelical Bible reading is driven by the continued search for relevance, as opposed to the search for semantic meaning. And, in interviews with readers of the Left Behind series, Amy Frykholm saw the reading of apocalyptic fiction as influenced by models for Bible reading. In particular, she identified the "life-application method," where "scripture must be directly and immediately applied to one's daily life" (2004: 111).

The common thread in these cases is the way the Bible is read for its meaning in the lives of readers, as opposed to its semantic meaning as a textual artifact. Wesley Kort traces this style to the Middle Ages, where an intimate relationship with the Bible was emphasized. Kort identified this as a Calvinistic practice, given John Calvin's insistence on lay Bible reading and the "diligent application of scripture to the whole of life" (1996: 23). Historian Richard Mouw includes this interpretive style as one of the four "minds" characterizing 20th century American Protestantism. He terms it the "pietistic mind," and describes it as geared toward "pious living," "devotions," and "fostering a sense of divine presence" (1982: 144). The interpretive style of application directly engages the textual ideology principles of Biblical relevance. Forstorp describes this relation aptly: "The premise for this interpretive approach is that the Bible is true, that all of the Bible is relevant for living a Christian life in any time" (1990: 164). The widely shared nature of

this ideological principle among the project groups helps makes sense of the uniform predominance of this style in group discourse.

My research confirms the prevelance of application; however, my discourse-centered framework reveals two additional characteristics, which I discuss later in this chapter. In the following, I argue that application appears in three forms in group discourse: personal, collective, and global. I use one case study to demonstrate how these iterations of the application style unfold discursively. Personal application, the most common style, is clearly illustrated by the Prayer Circle.

Personal Application

In chapter 1 I described the unique meeting format of the Prayer Circle. They were the only group to assign continuous reading of the same Biblical text, then come together and "share what God had been teaching them" through this reading. The group's facilitator, Darren, suggested this change of format. The following quote was part of Darren's initial presentation of the idea during the final minutes of the third meeting I recorded with the group:

"Okay. I had this thought. I've been reading this book called <u>Cultivating A Life for God</u>³², and the author talks about getting together in triads and holding one another accountable and encouraging one another. It involves prayer, confession, reading the Bible, and outreach. Then, what they do is they come together. They don't have a leader. Each one tells what God's been teaching them through what they've been reading. And, I thought, 'we ought to try that some time.' So, here was the thought. Read James. It's got five chapters. I think you can read it in about 20 minutes. And, what if we attempted, between now and two weeks from tonight, to read James everyday. And, let's say the minimum is that we do it ten times. And, that's not sitting down and reading James ten times in one day. It's reading James ten days or more. Now, my guess is if we spend that much time in James, God's gonna teach us something."

Darren emphasizes three things, all of which emerge from principles of the textual ideology: constant Bible reading, continued reading of the same text, and the confident

7

³² Cole (1999)

expectation that God will teach something to each participant through his or her reading. Two weeks later, nine of us met with Bibles in hand and teachings to share. Darren went first, and we moved around the circle clockwise. The following are a few examples of what was shared that evening. Three women in their early 30s shared a verse that "convicted" them. These examples also illustrate elements of the ideological principle of relevance:

Melinda: "For me, it's kind of going along with my life right now, of course. And, that happens a lot. About the trials and temptations, having perseverance, and then having the faith. I'm having a lot of trials come up, at least in the last couple of weeks...And, when I pray about it, this section kind of correlated with a verse in Matthew too, about praying as if you've already received what you've been praying for. So, praying to God with enough faith that it's almost as if you've already overcome this trial. So, I guess that's what's sticking out to me; just having that complete faith."

Jessica: "Yeah. The wages you failed to pay the workman who mowed your fields are crying out against you. I was home by myself and I was reading and I got to that. And, I had read it, however many times before, and it was right after work. And, I thought of my boss right away. As soon as I read that, I started writing him a letter. It just popped into my mind, 'what am I supposed to do with this?' Because you know me, I'm not confrontational. I can't go to him and say, "Read that" [[pointing to open Bible]]."

Jennifer: "It says we give blessings to the Lord and then we curse from the same mouth. It really convicted me on my interactions with other people that I show grace and compassion and that I not be judgmental. And, one example was this snowstorm we got within the last two weeks. I woke up the next morning and there was all this snow. And, my sister and I live in a duplex. I have this shovel that I bought two years ago. I leave it out front and prop it against my door. I don't care if she uses it. But, oftentimes when it snows she uses it and then she leaves it way on the other side of her house. So, in the morning when I'm trying to get ready for work and I've got my work clothes on I have to trek through a ton of snow and go searching for it. I've asked her time and time again, "Will you please put the shovel back?" And, I've been really thinking about my sister and wanting to show her love, but also kind of convicted in some ways that I was being an enabler. So, that morning when I woke up and there was all this snow and I was running late for work, the shovel wasn't there. So, I had to go get it from her side of the house. My pants were wet. My shoes were soaked. So, I shoveled my drive. And, I put my shovel in the house and left for work. Well, she called me at work and said, "Do you have the shovel?" I said, "Yes. It's in my house." She was very upset and she hung up. I was really convicted. I realized I didn't handle that in the best way. I was really praying about, you know, "You need to humble yourself here and show her grace and show her compassion."

This sequence of reading, sharing, and moving on to the next participant continued until everyone had shared something. Each participant read at least one, but more often several, verses from James. Following the verse, they shared an example from their life – sometimes generic, but usually specific – to demonstrate the text's veracity and applicability. On some occasions, the truth was one they were living out, showing how the text confirmed them. At other times, like the examples above, the truth was one they were failing to live out, showing how the text convicted them. As a collective reading event, these processes of confirmation and conviction became shared. The principle of relevance extends the reaction beyond the speaker, as listeners imagine their own experience of confirmation or conviction, bringing them closer to one another and to the Bible. Discussions rarely ensued about the shared nature of the application. Instead, participants listened intently and allowed one another to speak uninterrupted 33. This structure was followed in each of the remaining eleven meetings.

The Prayer Circle most clearly demonstrated the link between textual ideology and discursive practice. The organization of their meetings and the order of participation proceed directly from their expectations of Bible reading, particularly its relevance and its precise application to individuals. The variety of Biblical texts they chose reinforced the understanding of the Bible as a unified whole. They read from both the Old and New Testaments. They read a variety of genres, including: letters, prophecy, poetry, and history. These texts had purposes as diverse as theological instruction, prayer, and encouragement. Yet, the type of text they read never altered their approach.

-

³³ During the whole fieldwork period, I never once observed a participant challenge the appropriateness of another participant's application of a Biblical text during group discussion.

The mode of reading the same text repeatedly over a two-week period was also unique. This confirms that they should not treat the Bible like they do other texts, and it underscores the ability to always find new application. The personal application style also alleviates requirements of adhering to textual progression and sequence. These processes, such as following paragraphs from sentence-to-sentence, are taken-for-granted in most of our everyday reading activities. Yet, participants moved effortlessly among Biblical texts. They connected passages separated by multiple verses. They combined verses from different chapters of the same book to articulate common themes, as though they occurred back-to-back. This is not problematic for them in the case of Bible reading because these texts are bound together by a common search: how is God speaking to me through His Word?

Collective Application

This style of applying the Bible extends beyond the personal to other levels of social involvement; for example, situations of collective belonging. The primary target of this in group discourse is the life of their local congregation. The same expectation of relevance applied to individuals is equally present when framed for a community of believers. The Tuesday Men provide an exemplary case of how this can shape the nature of a group's discourse.

The following example comes from their first meeting in studying the book of Acts. The study guide question asked them to, "Put yourself in the shoes of the apostles. How would you feel if you were the first to be given the task described in verse eight³⁴?"

³⁴ "You will receive power when the Holy Spirit comes on you; and you will be my witnesses in Jerusalem, and in all Judea and Samaria, and to the ends of the earth." (NIV)

Following a few brief contributions by other participants, Ted proceeds to contrast the ideal presented in scripture and the life of their local congregation:

"Well, certainly a part of our Jewish apostles is a question of, "What in the world are you talking about? What do you mean? I mean, the Jerusalem and Judea are okay. But, when you get to Samaria and all the world, what are you talking about? Why in the world would we go there? Those are not God's people." I was talking with someone in some setting the other day about outreach. It was at church, but I don't remember what the setting was. I was actually sharing that I really felt like I needed to find ways to get out of the church more. One of the problems for us, those that have been around the church for a long time and for me, as somebody who works here, is that we don't run into people who are not a part of the church. We don't spend time, I mean, our friends are here. Our connections are here. And, so we're not in places where we're influencing for the Gospel, and we have to be. And, so, we need to find ways to do that intentionally. And, someone's response was, kind of joking, "Well, I hope we don't find you in the bars." And, my response to that was, "I hope you do." See, I think some of us have ideas of Samaria and the ends of the earth still that are off limits to us.

This case illustrates their interpretive preoccupation to continually engage in this form of collective application. They repeatedly established specific points where their congregation (as well as the modern Church in America) falls short of Church life as presented in the opening chapters of Acts. Given its position in scripture, its proximity to Jesus, and its involvement of Jesus' disciples, the first century Church is held up as the pinnacle and the standard of what the Church could and should be. Over the course of the remaining 22 sessions, their reading of Acts became a running comparison between the two states of the Church. Even when study guide questions prompted the group to consider features internal to the text, as in the above example, they managed to return to the collective application style³⁵.

As with the personalized style, collective application involves principles of the textual ideology surrounding the Bible. In this example from the Tuesday Men, assumptions of Biblical relevance are clearly at work. The historical and cultural distance between the Church of Acts and congregational life in contemporary America is

³⁵ For more on the Tuesday Men and their reliance on collective application, see chapter 5.

subordinate to the Bible's (divinely sanctioned) ability to offer guidance, insight, and exemplars. *Jerusalem, Judea, Samaria, and the ends of the earth* are not restricted to the first century. They are read as symbols that signify locations in the world of the reader; in this case, bars. The assumption of Biblical authority is also embedded in this style. Texts are applicable because they carry the weight of being the Word of God. They are not simply useful and instructive, but prescriptive and convicting.

Global Application

The Bible also is deemed relevant to societal and global affairs. This was the least common form of application, but it did appear occasionally in each of the case study groups. The McGee Men provide a brief example. During my third recording they paused on James 2:6: But you have insulted the poor. Is it not the rich who are exploiting you?

Are they not the ones who are dragging you into Court? Joel, a core participant, connects this to a recent corporate scandal:

"I couldn't help but think of Enron, WorldCom, places [where] fabulously rich people ripped off thousands of shareholders, thousands of pensioners. And, boy, when I read that that's the first thing, it said all the perfect stuff. [That] is still going on and people are still walking around."

The Bible's ability to remain relevant in the modern world is foregrounded again. The ability to apply the verse testifies to the Bible's continued and context-independent veracity. Joel's application of the text to corporate corruption also carries a hint of the ideological principle of textuality. The words from James are not simply fitting, approximate, or 'kind of the same thing.' They say "all the perfect stuff."

Taken together, these three forms of application constitute the predominant style of interpreting the Bible in group discourse. They all assume that the Bible applies to the situations of modern living in the same way it did when it was written. As I have

indicated, this style is upheld by several principles of the textual ideology, most notably: a relevance that transcends time, place, and people; a veracity that transcends similarly; and an authority that compels believers to consult the text and take seriously the application they find.

This style of application may be having a "colonizing" effect among American Protestants. Fairclough (1989) describes how certain discourse types, such as the discourse of therapy, have become models on which other, unrelated discourses pattern themselves (ibid: 198). Such 'colonization' may be at work with the interpretive style of application. First of all, we know this style traverses some theological and social divisions within Protestantism, as it does in this sample of three denominational settings. However, the question remains: does this style dominate within less mainstream Christianities (e.g., Native Christians, Jehovah's Witnesses, the Mormon Church)? Secondly, Amy Frykholm's (2004) analysis of Left Behind readers showed how this style has colonized the reading of Christian fiction genres.³⁶ A second question is: what other genres this style of reading extends? Is it restricted to religious reading, or has it colonized other forms of reading as well? Lastly, has this style of reading colonized other modes of "reading?" Crapanzano (2000) and Harding (2000) both argue that Fundamentalist hermeneutics bleed into their interpretation of everyday and current events. For example, Middle East conflicts are initially framed in apocalyptic terms, and only subsequently in political terms, if at all. Is the interpretive style of application also colonizing the understanding of everyday and current events?

³⁶ Frykholm focuses on "apocalyptic fiction" as a specialized sub-genre within the broader category of Christian fiction.

Establishing Textual Meaning

The Bible is not treated primarily as a textual artifact containing a semantic meaning, nor is it read to decipher such meaning. However, individuals and groups do interact with the Bible in an attempt to establish meaning; though, significantly less frequently than they do for application. And, my analysis of these searches for meaning provides significant lessons regarding textual ideology, interactive norms, and the interpretive process.

Four groups sought to establish the meaning of Biblical texts: McGee Men, Tuesday Men, LC-MS Men, and Monday Group. These groups used several means of identifying a lack of clarity in textual meaning. At times, this revolved around the definition of a single word or phrase. The McGee Men puzzled over the semantic ranges of total law (James 2:8-10); double-minded (James 4:8); and speak evil of the law (James 4:11). Establishing meaning also involved questions of genre and intention. The LC-MS Men had several extended debates about whether a text should be understood as a divine promise or simply a general statement of truth. Their lengthy debate over Proverbs 31:6-7³⁷ included framings of satire, hyperbole, and historiography. Several groups referenced the emic category of "literal," and its alternative possibilities. For example, the Monday Group's discussion of Hebrews 12:12-13³⁸ wrestled with the opposition between literal and spiritual lameness. Ambiguity sometimes involved situating a text as a theme running

³⁷ "Give beer to those who are perishing, wine to those who are in anguish; let them drink and forget their poverty and remember their misery no more" (NIV).

38 "Therefore, strengthen your feeble arms and weak knees. 'Make level paths for your feet,' (Proverbs

^{4:26)} so that the lame may not be disabled, but rather healed" (NIV).

throughout the Bible. For the LC-MS Men, an attempt to resolve Proverbs 28:2-3³⁹ meant discussing what other Biblical texts said about the concept of democracy.

However, regardless of what textual feature caused the ambiguity or disagreement, the interpretive process by which this style unfolded was consistent. In 16 of 19 sample interactions, the attempt to clarify the meaning of a Biblical text was characterized by a lingering uncertainty within the group about the most accurate reading. Group facilitators and participants were continually willing to accept this lack of consensus. The interpretive process typically progressed as following: an assertion of confusion -> an offering of multiple interpretive possibilities by multiple participants -> the likelihood of some, if not all, of these possibilities was considered -> a coda that evaded settling on one of the possibilities was stated -> a segue to the next conversational topic was given. The LC-MS Men employed this style most often, and the following example was among the most entertaining to observe.

Whoever flatters his neighbor is spreading a net for his feet (Proverbs 29:5).

When Pastor Daniel opened the discussion for questions and comments, Jeff raised his doubt about what exactly this pithy proverb meant. Others echoed Jeff's sentiment. The interaction lasted just shy of ten minutes, but the group managed to consider five different possibilities of the verse's meaning. Daniel began by asserting that the net must belong to the neighbor. Dan raised the first possibility: "[the verse means] you will catch more flies with honey." Jeff challenged him on the premise that spreading a net does not sound positive. Andrew affirms Jeff, adding: "flattery suggests a little bit of color, a little deceit, a little hyperbole." Andrew proceeded to challenge Daniel's attribution of the net

_

³⁹ "When a country is rebellious, it has many rulers, but a man of understanding and knowledge maintains order. A ruler who oppresses the poor is like a driving rain that leaves no crops" (NIV).

to the *neighbor*, asserting that the *net* more likely belongs to the one doing the flattering. Daniel rejoins the discussion, directing the group to Proverbs 29:18⁴⁰. His intent was to show how agency in the Bible is not always clear due to differences between Hebrew and English grammar. Bill joins the exchange next, affirming Dan's original assertion that the verse is positive. Daniel uses this as an opportunity to suggest that the positive/negative juxtaposition is a common strategy in Proverbs. Jeff then articulates a fourth possibility: the verse is meant as a warning against dishonesty and luring people into false self-perceptions. Daniel adds the related possibility that the verse is equivalent to "the old one-minute manager...you say something positive to someone, then you nail them, and then you end with something positive." This conjured a round of laughter. Dan then provides an example from his business interactions to demonstrate his original assertion. The exchange closes with Daniel restating the strategy of juxtaposition, using it as a link to the next proverb to be discussed.

I remember enjoying how the men worked through this interpretive process. They appeared so intent on 'figuring it out.' Yet, in the analysis of this transcript, as well as the examples of this style in other groups, the determination to establish meaning was not the most striking feature of the interaction. That was, instead, the seeming acceptance of polysemy, and the lack of a definite resolution. Several implications arise from this distinctive interpretive process. In regard to the norms of group discourse, it aids the expectation of mutual edification. The purpose of small groups would be challenged if the interpretive process were defined by either a stubborn insistence on one's correctness, or a derision of others' suggestions. Allowing multiple possibilities to linger also relieves the group of settling on one meaning (and, perhaps, the subsequent pressure of defending

⁴⁰ "Where there is no revelation, the people cast off restraint; but blessed is he who keeps the law" (NIV).

it). The textual ideology is also at work in this process. The operative principle is the commitment and necessity of continually reading the Bible. Leaving the meaning unclear encourages this, and promises future debate. Lastly, this interpretive style supports the predominant style of application. If meaning is understood as flexible, one is reassured that imagining forms of relevance is not only acceptable, but perfectly appropriate.

Explaining Biblical Texts

The third style of interpreting the Bible is also involved in the search for meaning. However, its explanatory style is distinct from the strict establishment of meaning in its aim and requirements. When groups attempt to explain a Biblical text they are identifying why a text says what it does. It is not a question of 'what,' as with establishing meaning, but of 'why' and 'how.' This style encompasses a variety of hermeneutic activities, including: specifying the intentions of the author; attributing motivations and emotions to Biblical characters; describing details absent from the text; and appealing to issues of historical and cultural context.

The key process in explaining Biblical texts is textual inferencing. The interpretive work accomplished in this style requires participants to assert meaning not explicitly stated in the text. Groups encounter questions, from study guides and of their own formulation, that ask them to fill in information and rationales not provided by Biblical texts. This process of inferencing also accomplishes epistemological work by requiring readers to access background knowledge, predispositions, and fundamental beliefs (Fairclough 1989). In performing this function of knowledge production, explanation is central to the interpretive work of group discourse. In the following example, I use the McGee Men to demonstrate the importance of this style.

The second chapter of James closes with two Old Testament examples of faithfulness to God⁴¹. The first is Abraham, the iconic figure of Genesis who earned the title of *God's friend* by his willingness to sacrifice his own son on God's command. The second example is the prostitute Rahab, who James singled out for aiding Joshua's spies and deceiving the king of Jericho. The McGee Men paused on this second example. After reading the story from Joshua 2:1-21, Dennis objects to its use as a demonstration of faithfulness. He suggests that Rahab's actions resulted from an altogether different motivation:

"I don't think it was as clear cut a case as the Abraham thing. To me, Abraham's thing was really faith. He would have sacrificed his son if God hadn't intervened. I think with Rahab there was a whole lot of self-preservation involved there. She could see where the bread was buttered and what was going to happen, and she wanted to be on the right side of them when they came in."

Vincent defends the example, arguing at length that Rahab did indeed act out of faith. In an attempt to support his position, Vincent draws a parallel between Rahab and the well-known example of Daniel and the lion's den:

"Ultimately, what's the difference between [Rahab] and Daniel, who was supposed to have quit praying to God, didn't, went into the lion's den and decided that was his better recourse than giving up his relationship to God?"

Dennis persisted in his assertion:

"For me, that's a better illustration of faith than the Rahab thing because Daniel, despite knowing that the king might kill him or do whatever, he was going to pray to God anyway."

The inference at work in this example comes from the Bible. The passage from Joshua never equates Rahab's reaction with faithfulness. This motivation is presented in the book of James. The inference of faithfulness is dismissed by Dennis and supported by Vincent. This example exemplifies how the style of explanation is involved in the

⁴¹ James 2:20-26.

production of knowledge. In particular, the two men are expressing different orientations to the textual ideology principle of authority. By replacing faithfulness with self-preservation, Dennis is asserting that the book of James is, at best, misguided, and at worst, wrong. At one point, he accuses James of "stretching a point." Dennis's position implies a freedom to disagree with the Bible, and thus a challenge to the absolute authority of scripture. In defending James, Vincent is insisting on the authority of the Bible, and refuses to challenge it. The group knows that Vincent adheres to the Bible's absolute authority, and that Dennis eschews such doctrines. These differing relationships to the textual ideology become enacted through the style of explanation and the process of inferencing. Their disagreement, in fact, has much more to do with the nature of Biblical authority than it does the faith of a prostitute.

Inter-Relation among Interpretive Styles

The three styles of interpreting the Bible employed by group participants – application, establishing meaning, and explanation – exist in dynamic relationship to one another. Rarely, if ever, do groups construct this interpretive process as a tightly bound sequence progressing from explanation to interpretation to application. These styles exist more as discursive resources utilized by participants and drawn into group discourse on a text-by-text basis. For some, like the Prayer Circle, application was the only style employed, primarily in the personal form. For others, like the McGee Men, attention was divided fairly evenly among the three styles.

This balance of styles among the McGee Men might appear to be a foundation for accord within the group. It is reasonable to suspect that a shared willingness to move among the three interpretive styles implies a shared orientation toward Biblical texts. But,

this is not the case. As chapter 1 described (and, as I argue in chapter 5) the group's discourse is defined by opposing orientations to the Bible. The example in this chapter of Dennis and Vincent sparring over scriptural authority provided a glimpse into this disjuncture. The use of these styles is clearly a locale for exercising such ideological tensions. What, then, can we say about the co-mingling of styles? How does their interrelation intersect with these divisions? The participant roles of Vincent and Roland are instructive.

Recall the adherence to Biblical authority displayed by Vincent in his reaction to the Rahab example. He sustained his loyalty to this principle of the textual ideology throughout the group's study of James. With Vincent's loyalty in mind, consider the following statements made by Roland, both of which followed lengthy debates about textual meaning:

"James [is] a little difficult for us because he sounds so absolute. And, I think if you put the love ethic on it, you don't want to be so absolute in everything you say. I don't know whether he's trying to make everybody feel guilty or what."

"People think [the Bible] is a yes or no, black or white, kind of thing. And, it isn't. It's all this in between and we've gotta make sense out of it... I can go to scripture, but it won't help me with this because I've got all this in between stuff that's going on."

Roland's statements reveal a fundamentally different attitude toward Biblical authority than that of Vincent. The freedom to disagree, challenge, and criticize Biblical texts diverges from Vincent's unwavering commitment to confirm scripture. Vincent's fidelity to Biblical authority, however, was on par with his commitment to the style of personal application. In every meeting, he found at least one verse from James to prompt a detailed retelling of a recent event exhibiting his own personal triumphs or failures at "living Biblically." Consider, then, the following statements made by Roland:

"I think we want to believe that the Word of God is new everyday. It's not lost in history,

but it's new everyday. So, when you read, it's like reading it for the first time."

"I bring this up not to expose the church. I'm really bringing it up because we really got to keep thinking this thing through, and see how it applies in our real, actual lives."

Two principles of the textual ideology are being articulated in these examples: authority and relevance. Clearly, Vincent and Roland are much closer in their understanding of relevance. As the group's study of James progressed, the primary emphasis shifted from establishing meaning to finding application. The lengthy debates over establishing meaning gave way to lengthy exchanges where participants shared the mirroring of their own lives in Biblical texts. This shift in group discourse can partially be explained by the greater capital of application as an interpretive style (cf., pp:135-6). However, other factors contribute to the group's distinct shift from one to the other. Roland's position as pastor of the congregation is part of this dynamic. His recognized authority made his appeals to application more convincing for others to do the same. Also, for Vincent and others like him in the group, it was far better to exploit the shared understanding of relevance and personal application with the pastor than to continually struggle over textual meaning and Biblical authority. And, because these two interpretive styles foreground different ideological principles, the shared ground of relevance won the day.

Compared to the McGee Men, the Tuesday Men's study guide played a much larger role in orchestrating how they managed the co-mingling of interpretive styles. The McGee Men's use of a Fundamentalist commentator had more to do with providing a foil to spark discussion than it did with aiding their understanding and appreciation for the book of James. In contrast, the Tuesday Men appreciated the direct and open-ended style of the Life Guide study questions. These questions fit the categorization of interpretive

styles I identified in group discourse. Table 8 shows the distribution of these questions for the sheets they used during my fieldwork⁴².

	Personal Application	Collective Application	Meaning	Explanation
Acts 1	6	-	-	3
Acts 2	1	3	1	4
Acts 3	5	-	-	3
Acts 4:1-31	2	2	-	4
Acts 4:32-5:16	3	3	1	4
Acts 5:17-6:7	3	3	-	3
Acts 6:8-7:60	4	•	1	3
Acts 8:1-40	3	•	-	4
Total	27	11	3	28

Table 8: Tuesday Men Study Guide Analysis

The group addressed each study question in sequence. Therefore, Table 8 also provides a cursory view of the group's interpretive styles. Consistent with the other project groups, the style of application, particularly personal application, was predominant. The few attempts to establish meaning, as well as the relatively high number of attempts at explanation, are both striking. How the group handled the explanation questions is even more striking. As illustrated earlier in this chapter, Pastor Ted and the other participants often used questions internal to the text as opportunities to contrast the life of the Church in Acts with the faith and church life of modern believers. For the Tuesday Men, then, the co-mingling of styles led to collective application and the accompanying principles of relevance and authority.

In the case of the LCMS Men, the inter-relation of styles takes place within a defined participation structure. Daniel began each meeting with a summary of the day's chapters, usually identifying themes that characterized the majority of verses in each

⁴² An example of these study guides can be found in Appendix 3. In addition to the styles described in this chapter, the study guide questions ask two other types of questions. "Role playing" questions asked the group members to put themselves in the position of a Biblical character and discuss how they would have reacted. This is distinct from explanation because it is not asking the group to explain why the characters acted as they did. "Descriptive" questions were very straightforward, asking the group to recount from a text what happened, i.e., to paraphrase and summarize the events.

chapter (e.g., truthfulness, laziness, adultery, foolishness, social justice). Daniel then opened the discussion to the group, asking which verses "stuck out to them" or that they "[had] questions on." Midway through the meeting (~30 minutes) Daniel read off the Proverbs that "struck him." Depending on how much these overlapped with the verses already discussed, Daniel sometimes returned to asking the group which verses they highlighted. Daniel was always prompt in transitioning to the prayer time at or near the eight o'clock hour.

The co-mingling of styles in the men's discourse centered on personal application and establishing meaning. As I described above, the LC-MS Men attempted to establish meaning more than any other group. Therefore, the disjuncture between application and meaning was less apparent with them. The distribution of these two styles amongst the participants, however, was far less balanced. When Daniel opened the discussion to the entire group, the men raised issues of meaning or veracity, questioning the verse or pointing out just how true a particular verse is. Daniel assumed a much different role. He was virtually the only participant to raise questions of personal application. This happened at every stage of the meeting structure, from his introduction to his sharing what verses he highlighted.

One explanation for this is that the other participants do not accept textual ideology principles regarding Biblical relevance. This is quite wrong, as evidenced by the results of the questionnaire. Five different statements were posed about this aspect of the textual ideology. In each case, there was nearly 100% agreement affirming the ongoing relevance and veracity of the Bible for every generation. Additionally, the willingness,

ease, and detail of their responses to Daniel's eliciting of personal application belie any separation of men from this principle of relevance.

Instead, I understand this distribution of interpretive styles to be a product of Daniel's attempt to balance the interactive frames of 'sharing' and 'analytical discussion' described in chapter 1. Much of the church growth literature posits a link between intimate small group environments and expansive small group programs. This everpresent goal of bolstering one's congregational activities and members has prompted Daniel to make personal application a more explicit component of group discourse. The attempt to increase 'sharing' is also apparent in Daniel's alteration of the group's prayer time. Prior to his arrival, the group members prayed silently following the discussion. Now, the men gather in groups of three to five and pray aloud individually. Thus, Daniel's attempt to place more emphasis on personal application is part of a broader goal of creating more intimacy in the life of the group. Interestingly, he is ever cognizant of his progress in this effort. Just prior to their second book study, Daniel shared a text with me that he was considering suggesting to the group: Every Man's Battle⁴³. The book deals with sexual temptations, such as pornography, and promised to raise topics and levels of intimacy foreign to the group's inclination toward analytical discourse. I asked him during an interview why he never suggested the book. He replied, "I don't think they're quite ready for that yet."

These three examples demonstrate how the inter-relation among the three interpretive styles unfolds in different group settings, often intersecting with other aspects of group dynamics. In each case, the application style predominates, though groups achieve this in unique ways. In introducing application, I hinted to an argument about the

130

_

⁴³ (Arterburn, et al. 2000)

relation between application and the other styles. The co-mingling of styles is not simply a matter of numeric comparison. As I demonstrated above, these styles exist in substantive relations with one another. My second argument about these relationships is raised by this question: how does the nature of appeals to the style of application demonstrate its predominance in group discourse? The following two interactions help provide an answer.

The first comes from the McGee Men. It occurs directly after Vincent and Dennis's debate about the use of Rahab as an example of faithfulness. Following the back-and-forth disagreement, Dennis abruptly changes direction:

Dennis: A current illustration of just what we're talking about, to me, was this woman in Atlanta who was involved with this murderer⁴⁴.

Jason: Oh, yeah.

Dennis: What a dramatic story. This woman took this guy in and instead of just taking the first crack she could to get out of there and turn him in she went with him, took this truck somewhere, came back to the house, cooked him breakfast, talked with him for hours. That was an act of God. He said, "God has sent you to me," or something to that effect, and, "I am your brother and you are my sister." I thought, 'Wow. What a powerful witness.' And, I don't know what her faith situation is, but to me, God was active there. Saved from many more killing and, ultimately, the guy gave up without a struggle. She was one cool lady, I thought, who went through a lot and handled it in an amazing way. Vincent: I haven't heard that story, although I've heard just a little bit of an allusion to it. I heard that she made breakfast for him and stuff, and that he said, "I don't want to hurt vou."

Dennis: Mhmm.

Vincent: Of course, you hear that a lot, from people like that. But, I'm very interested in reading that story because I agree with you. I read a first person story from a woman who was kidnapped, pushed into her car. She was driving. But, basically, kidnapped by this guy who was running from the law and had just killed a bunch of people, and had a long record of it. And, she didn't know who he was. She had to drive around a roadblock and do all this kind of stuff, and many hours later she dropped him off somewhere. Her husband, you know, didn't know where she was. The police were looking for her, because she didn't come home when she was supposed to and they concluded that the worst probably happened. And, he was shortly thereafter caught. But, he had a string of killings and it made absolutely no sense that he did not kill this woman. And, she talked

⁴⁴ Dennis' illustration refers to an event in March of 2005. A man, Brian Nichols, killed several people before taking a woman, Ashley Smith, hostage. Smith, a self-avowed Evangelical, was able to persuade Nichols to turn himself over to the police by talking to him about her faith and reading excerpts from The Purpose Driven Life (2002). The story received widespread media coverage in the subsequent months, and ultimately led to Smith co-authoring her spiritual memoir.

to him like a regular person. And, she kept praying, "Keep me calm. Keep me focused." She didn't know who he was. It was only afterwards did she, and she said, "I'm really glad I didn't, because he was such a horrendous killer." That could have frightened her so much that she would have acted irrationally or out of frustration, [and he] would have killed her. And, she consoled him and she listened to him and, step-by-step, his fear and his rage just calmed right down. So, he had no desire to harm her.

The second interaction is from the Tuesday Men. It begins as a case of explanation, as raised by the style of the study question. The group had just read Acts 8:9-25 – the story of the sorcerer, Simon, who was converted by Peter and John after he witnessed their power through the Holy Spirit:

Ted: How does he attempt to get spiritual power and what is his motivation for wanting the power? Pretty straightforward. What's he trying to do? {2.0}

Larry: Tried to buy it with money.

Ted: Yeah. That's what he wants. He sees this healing that Peter and John bring about as they pray. And, what's his motivation?

Larry: So he could lay hands on the people.

Ted: Right. And, I think probably by implication, given what he had done before, he liked being important. He liked having power. He liked having people coming to him. He had a following of folks. He liked people to be impressed by him.

Eddie: Maybe just to take the other side, I can see adding to his repertoire. But, for myself, I'm seeing that maybe he saw what he was doing before was more trickery and now he saw he could really do something good. You know, "I'm really gonna help God now." And, he had this picture. He had really been affected by the disciples and now he wanted to be top man at being a good disciple.

Ted: Well, that's a possibility if we only read through 19. But, if we keep going and read what Peter says about him, I think Peter saw something else in his motivation. He seems to be all about himself. He seems to be all about looking good and impressing people and those kinds of things.

Eddie: I was thinking we can just kind of push him to the side, 'He's a bad guy. We're good guys. This doesn't have an application to us.'

Ted: Oh. okav.

Eddie: This is what I'm getting at. And, I think my conversion involved realizing that my job wasn't to help God, that I needed his help. I always wanted to help God out and I thought that's what it meant to be a Christian is to help God out. Well, no, it's something entirely different.

Ted: Sure. Yeah, our need is deep and great isn't it?

Eddie: Yeah. So, I sort of think we need to look at Simon in a different way than as the bad guy.

Ted: Yeah, I think you're right. I think we understand Simon. I wasn't suggesting at all that we just cast him aside as the bad guy. Yeah, I think it's very applicable to our lives. **Eddie**: And, I think the other application would be the very idea that the condemnation that Peter had about buying is our temptation to find short cuts.

Ted: Sure. 'We're gonna get this done. Oh, I can just do this,' not really looking that God is to do it and we have to do it God's way. We flip the switch and then get on with other

things. That happens all throughout the scriptures. The classic case that comes immediately to mind is Abraham [and God telling him he would have a child.] He [went right to Hagar and] didn't check with God about it. He just figured out what was the quickest way to get God's work done rather than waiting and listening and following what God had in mind. So, sure. I think we do that all the time.

These two interactions share three important features. First, they involve competing explanations. Vincent and Dennis disagreed about how to account for Rahab's action and motivation. Ted and Eddie disagreed about how to account for Simon the Sorcerer's action and motivation. Secondly, the disagreement is defused, but it is not resolved. And, lastly, this defusing is made possible by an appeal to the style of application. In the McGee Men, after Vincent's insistent attempt to convince Dennis of his assertion, he transitions without objection to a recent example of faithfulness he recalled from the news. In the Tuesday Men, Ted dismisses Eddie's inference until he frames it as an application of the text to their own everyday spiritual life. Ted immediately accepts this, supporting Eddie with the scriptural example of Abraham.

This sequence of application superseding another style was present throughout these case studies. This can be understood in terms of "cultural capital" (Bourdieu 1977). In any given cultural situation different resources, strategies, and actions will differ in how they are valued by social actors and institutions. In short, they will have unequal worth, and their worth will always be contextually dependent. In the case of group Bible study, application is the interpretive style with the most capital. As the above examples demonstrate, appeals to how a text applies in the life of an individual or group often came at the expense of struggles over meaning and explanation. Application worked to defuse, mediate, circumvent, and even ignore interpretive dilemmas arising from the other styles.

Small Groups as "Interpretive Communities"

This chapter has presented several arguments about how the Bible is interpreted within small groups:

- ➤ Bible interpretation in small group discourse is characterized by three styles: application, establishing meaning, and explanation.
- > Application occurs in three forms: personal, congregation, and global.
- > These three forms of application are united by textual ideology principles emphasizing the ever-relevant nature of scripture, its atemporal veracity, and its consistency as a coherent narrative.
- > Application may be exerting a colonizing influence within American Protestant culture, affecting how believers experience other texts and everyday events.
- The primary characteristic of establishing meaning is the willingness of groups to offer multiple interpretive possibilities without any need to achieve consensus.

 This interpretive process reinforces group expectations to be edified, textual ideology principles, and the predominance of application.
- > The key feature of explanation is the process of textual inferencing. This process contributes to the production of knowledge, particularly as it relates to the textual ideology surrounding the Bible.
- > These styles are mutually dependent. These relations are upheld by interactive features operative within individual groups.
- > The greater capital of the application style is evident in its numerical prevalence, but also in its ability to defuse conflict arising from competing possibilities over meaning and explanation.

The consistency of these characteristics throughout the project sample are sufficient reason to view small groups as a form of "interpretive community" (Fish 1980). Stanley Fish theorized that just as speakers of a language share a structural system, readers internalize similar systems that make possible shared methods of interacting with texts. Interpretive strategies are not to be found in the internal structure of texts, or in the subjective preoccupations of readers. Rather, interpretive processes are collective products of a community of interpreters, and their shared system of strategies, styles, and assumptions. In concluding this chapter, I present five propositions for viewing small groups as interpretive communities.

Proposition #1: The unfolding of these styles in group discourse is inextricably tied to the textual ideology surrounding the Bible. The notion that one's beliefs about a text will impact his or her reading of that text takes on new life in the context of small groups. For the Prayer Circle, the structure of the group's meeting and the interactive order are both rooted in a strict adherence to the ideological principles of relevance. In regard to the role of establishing meaning, the interpretive process of leaving meaning unresolved reflects an emphasis on continual interaction with the Bible. In cases of inferencing, assumptions about the nature of the text, such as its authority, are a primary form of background knowledge. Therefore, the use of a particular style necessarily assumes principles of the textual ideology. This helps ensure that the textual ideology is an everpresent figure in the backdrop of small group discourse, and often the hidden ground of confusion, agreement, and disagreement.

Proposition #2: The use of these three styles is central to the production of knowledge. Primarily, the Bible is approached as a text useful for making sense of the

individual life and the surrounding world. Bible reading is open to questions of meaning and explanation, but these are subordinate to the style of application. Foregrounding application also ensures a dialogical quality of the Bible (cf., Bahktin 1981). If a verse or passage can be approached 'new everyday,' and is expected to speak directly and individually, then interpreting the Bible accrues the weight of past experiences with the text. Certain Biblical texts are regarded as particularly meaningful because of their impact at an earlier period in one's life, be it joyful or tragic. Yet, they are not restricted to that specific time. The epistemological work involved in the style of explanation has already been discussed. The process of textual inferencing requires readers to access forms of background knowledge to fill in gaps of meaning left open by the text. In the case of Bible reading, and the examples used in this chapter, participants often draw on textual ideology principles. However, as groups have differing interpretive preoccupations and emphasize different ideological principles, this process is open to other forms of epistemological work.

Proposition #3: In addition to being an epistemological tool, the Bible is also a tool for communication. For the style of personal application, the placement of the Bible in one's own life provides participants an opportunity to talk about and share their experiences. The Prayer Circle, again, is the consummate example. Participants are able to share their lives with the group through linking personal struggles and victories to the authoritative text of the Bible. The Bible is a means to present their life for affirmation and advice. In regard to meaning, participants communicate their desire to 'know the Bible better,' their preferred interpretive styles, and their willingness to posit an interpretive possibility. For explanation, the Bible provides a ground for communicating

one's understanding of the nature of scripture, as well as familiarity with Biblical narratives via the imagining of characters' emotions, intentions, and rationales. In short, the Bible is used as a device for communicating notions of personal, interpretive, and group identity.

Proposition #4: The reliance on only these three interpretive styles highlights the boundaries of Bible interpretation in group discourse. Application, establishing meaning, and explanation are not the only styles we might reasonably expect to find in Christian communities. For example, typological⁴⁵ readings have been pervasive in historical and contemporary African-American Bible use (Noll 1982, Wimbush 2000). Native American Christians have often deployed the Exodus narrative as a type for their own removal and return to native lands (Treaty 1996). In her study of 7th Day Adventists in Madagascar, Eva Keller (2005) observes this as a regularly employed interpretive style. And, typological interpretation has been observed as a rhetorical strategy used by preachers to connect themselves, and their audience, closer to Biblical characters (Harding 2000). Yet, this historically significant style is absent in these Bible study groups.

This quality of downplaying and foregrounding particular styles is consistent with Fish's understanding of what interpretive communities accomplish. The use of certain interpretive strategies and not others must be traced to the community, not to the text itself or to the individual reader. This uniformity in Bible interpretation reflects the

-

⁴⁵ A typological approach to the Bible envisions it as consisting of a series of 'types' and 'anti-types.' 'Types' prefigure 'anti-types,' so that the anti-type can be read as a fulfillment of the type. Historically, Old Testament characters have been understood as 'types' of Christ, such as Adam, Noah, or Moses. They partially reveal the nature of Christ, and point toward the need for Christ, but they are not Christ. Christ, as the anti-type, fulfills what was only anticipated by the Old Testament types (cf., Frye 1981). As Harding (2000) notes, what I have termed "application" is a form of typological reading. However, it is distinct from the historically-oriented style of typological reading that is evidenced among African and Native Americans.

uniformity of this interpretive community, primarily around the textual ideology surrounding the Bible. While alternative styles may be known by participants, they are avoided because they are outside the bounds of the interpretive community, and therefore outside the bounds of group discourse.

Proposition #5: The importance of the application style has been discussed at length. There are other implications, however, for the model of the interpretive community. Reading the Bible for its relevance allows groups to circumvent divisive theological issues raised by scripture. Moving from 'what does the Bible mean' to 'what does the Bible mean to...' enables groups to avoid positioning themselves within a doctrinal camp that could be at odds with other participants and/or the denomination.

Gravitating toward the application style also allows group discourse to be more inclusive. The socialization of newcomers into the life of the group is not as daunting in an environment where application is the preferred style. New participants can arrive and immediately contribute when the textual ideology of the Bible's relevance is being emphasized. A much less inviting situation would be constant and, at times, contentious attempts to establish meaning.

This chapter has demonstrated how the Bible is interpreted in group discourse. In addition to describing the individual styles, the chapter defined the discursive features of these styles, their distribution within the groups, and their inter-relation among each other. Ultimately, this portrait of interpreting the Bible is simultaneously a portrait of textual ideology-in-practice, and the nature of small groups as interpretive communities. Chapter 4 continues this effort by shifting the attention from the interpretive style

employed in small groups to the interpretive resources participants rely on to engage in group discourse.

Chapter 4: The Interpretive Matrix of Small Group Discourse

I picked Rick up from his house at 7:30 in the evening. The event began at 8, but Rick wanted to be early so he could talk to the featured speaker – a personal friend of his. The event was a "business seminar," the type Rick attended at least once a month to gain insights for bolstering his own "independent business."

Attending the event with Rick was a case of ethnographic misunderstanding, and ethnographic faith. I was four months into the fieldwork, and Rick was the only person who facilitated two different small groups: the Mind of Christ and the Monday Group. Rick consistently wove experiences from his relationship with a network of "Christian businessmen" into his participation with each group. The link between faith and financial success is an established part of Evangelical culture⁴⁶. This prompted me to arrange an interview with Rick. When I arrived at his house for lunch, I had to rearrange my expectations.

As the interview progressed, I realized two things. First, the "Christian businessmen" of Rick's discourse were not part of any nominally Christian organization, but simply businessmen that were Christians. Secondly, Rick understood my interest as an expression of interest in wanting to become involved in the business. I was familiar with the pyramid scheme approach that was the basis of his business's operation, but had no interest in it. The interview eventually turned into Rick presenting his practiced pitch for the business. Eventually, he invited me to join him at a business seminar the

⁴⁶ The primary example is John Maxwell, a best-selling Christian author. His books deal almost exclusively with strategies for improving "leadership," "financial success," and "business growth" through an increased emphasis on Christian faith. His best sellers include <u>The 21 Irrefutable Laws of Leadership</u> (1998) and <u>Today Matters: 12 Daily Practices to Guarantee Tomorrow's Success</u> (2004). He also has an annotated Bible emphasizing leadership, <u>The Maxwell Leadership Bible: Lessons in Leadership from the Word of God</u> (2003).

following night at the Holiday Inn. I was determined to find application where Rick saw application, so I promised to accompany him.

The hotel conference room was arranged simply: chairs lined ten-deep in five rows, an aisle dividing two sets of seats, and a podium at the front. When Rick and I arrived the room was nearly full. I followed him to the front of the room where we took two front row seats. Rick then directed me to the back of the room where Jay, the night's featured speaker, was greeting people. Rick's start as an "independent business owner" came via Taylor, another member of the Monday Group. Jay recruited Taylor into the business, which made Rick and Jay part of the same "network." Jay's success in the business has made him a model for success, as well as a sought after speaker for seminars. Jay greeted me enthusiastically, pleased to know that Rick had brought someone new.

The talk was lively, and Jay maintained an air of spontaneity throughout. After the first 30 minutes, though, my faith in seeing application to Bible study was fading. Up to that point, he talked only of business details. Following his lecture, Jay moved into a period of recognizing individual sales achievements. My patience was nearly exhausted as Jay concluded:

"My advice to any of you who want to achieve these kinds of goals is simple. Go home and read The Prayer of Jabez⁴⁷. You can't do anything without divine help. You can do anything you're doing ten times more if you're praying about it. My advice is to get God involved. The reason I'm successful, the reason these other guys are successful, is because we have a powerful partner. The reason I went diamond⁴⁸ is the Lord. If I have Jesus sitting next to me you think I'm afraid to pick up the phone? The guy on the other end doesn't stand a chance!"

⁴⁷ The Prayer of Jabez (2000) was authored by the popular Evangelical author Bruce Wilkerson, and spent considerable time on Christian and secular best-seller lists. Wilkerson bases the book on an Old Testament prayer, and emphasizes how Christians ought to pray for security and prosperity, claiming at times that if the prayer is prayed faithfully and regularly, then success will be achieved.

⁴⁸ As one achieves certain levels of income in the business, they progress through different demarcations. The "diamond club" is the second highest level.

The remaining hour of Jay's talk included sporadic, but sustained testimonies like the one above. Though not explicitly Christian, the seminar bared the imprint of Evangelicalism. This mingling of discourses helped contextualize Rick's small group participation, as well as his blended identities of "independent business owner" and "Bible-believing Christian."

My evening at the Holiday Inn indexes a central issue in small group discourse. The discussion of any text, topic, subject, or question takes place against a backdrop of textual and experiential resources⁴⁹. Just as Rick incorporated his life as a businessman into his role as group facilitator, other Bible study members found other sources of application. Participants marshal these resources in the service of clarity and understanding, for getting their point across, for making sense of others' contributions, and for interpreting texts. This chapter addresses three questions: what resources do group participants rely on, and toward what ends do they use them? How do these resources differ in terms of the degree to which they are shared within and across groups? And, how does the use of resources contribute to the work of textual ideology that underlies small group discourse?

Chapter 3 argued that small groups are interpretive communities, with a focus on common styles of Bible reading. However, even those groups who studied directly from the Bible talked about more than just scripture. In turn, a portrait of small groups as interpretive communities requires closer attention. And, in any interpretive community, the role of discursive resources is an integral component.

_

⁴⁹ I owe my original interest in the role of knowledge sources in Bible study discourse to Dr. Fredric Roberts, who included such questions in his four-year ethnographic study of 12 churches (Roberts 2005).

The work of William Hanks (1989) on how texts are defined and understood provides a useful framework for this task. Hanks suggests that for any text to move beyond potential meaning(s) to concrete meaning(s) it must be "centered" in an "interpretive matrix" (ibid: 106). Hanks identifies this matrix as the "social context" of the discourse event, and calls for anthropological approaches to text that account for "norms of interpretation, institutions, and habitual orientations to discourse" (ibid: 107). The bodies of knowledge participants draw on as resources are a significant element in all three categories identified by Hanks. Jody Davie (1995) recognized the importance of the interpretive matrix in her ethnography of a Presbyterian women's Bible study group. Davie demonstrated how extensively the background in Jungian psychology of the group leader informed the group's conversations. The assumptions and insights of Jung's theory were a filter through which the group's discourse continually passed.

The goal of this chapter is to describe the resources that define the interpretive matrix for these groups, and identify how they are integrated with the textual ideology surrounding the Bible. The entire interpretive matrix is much broader than what will be identified in this chapter. Certainly, participants rely on their experiences as parents, employers, employees, volunteers, siblings, leaders, conscientious citizens, and so forth in group discourse⁵⁰. However, I will concentrate on the resources that define and exert a persistent influence in specific groups. My discussion of the interpretive matrix is divided into three sections: resources present throughout the entire sample of groups, resources with a heightened emphasis in particular groups, and resources with a heightened emphasis for individuals.

_

⁵⁰ Cf., Roberts 2005:45-48 for a discussion of how non-religious knowledge sources impact congregational life.

Shared Resources

Few resources were present throughout the entire sample. I will discuss the two that were most influential, using one group for each to demonstrate how the resource was used. Both are involved in the work of identity formation, but at different levels of belonging. One body of knowledge groups relied on was the evolving situation of their local congregation. Each of the project groups was affiliated with a church body, which provided a shared filter for engaging texts and topics of discussion. The second resource was denominational identity, where groups identified Selves and Others within and outside the Protestant tradition⁵¹. To begin, I use the Tuesday Men as a case study for how local congregations are used as interpretive resources.

Congregations as Resources

I described in chapter 1 a recent change in the demography of ICUMC – a growing population of foreign refugees from Africa and the Middle East that began in the fall of 2003. The first refugee, Muhammad, came from Afghanistan. He was a middle-aged man, already a Christian, and delighted to find a church near the home where he was placed. The warm welcome from the congregation prompted Muhammad to become an active evangelist for ICUMC. His frequent visits to the local Refugee Services office provided an opportunity for him to invite other refugees and their families to Sunday worship services. At present, there are nearly 30 refugees (including children) who attend the church regularly.

In the late spring of 2004, the refugee presence was large, and unorganized – enough that Eddie volunteered to act as an unofficial conduit between the refugees and

⁵¹ Throughout this chapter I employ the nomenclature of "Self" and "Other" to signify conceptualizations of collective identity. For an extended discussion of how congregations employ denominational identity in this manner, see Roberts 2005:59-77.

the church. Pastor Ted was excited about their presence, but unsure of how to make the best adjustments in the life of the congregation. He eagerly accepted the offer, primarily because Eddie had spent three decades as a Christian missionary in India. Eddie's job was to introduce the local church to the life of the refugees, introduce the refugees to the life of the local church, and provide instruction for those unacquainted with Christian (read: United Methodist) theology. At present, Eddie is the personal contact for the refugees, ICUMC's representative at Refugee Services, and the leader of a Sunday school class designed specifically for the refugees.

Given Eddie and Ted's positions as core participants in the Tuesday Men, it makes sense for the congregation's ever-evolving relationship with the refugee population to be a key feature in the group's interpretive matrix. The group's use of this resource coincides with their typical style of interpreting the Bible – finding examples of collective application. Consider the following interaction. The group had just finished reading Acts 2:42-47, which describes the fellowship among the early church believers:

Ted: How does the life and purpose of your church or Christian fellowship group compare to this group?

{11.0}

George: Peter just kept preaching the same thing, it seems like here. You read throughout, he kept preaching salvation, which I'm very thankful for. But, I look at the group, the church, I talked a little bit about this last week that they weren't, this was a new group. They're just starting up. They weren't focused on what the carpet was [and] didn't have the distractions that we get into today.

Ted: Yeah. Excellent word. I've been thinking a lot about that as we're working on starting the new service and kind of looking at it from the concept of a new church. I wonder about the average of how long it takes for a new church to move from the level of excitement that you see in many new churches to, and it's a bad word but I'm gonna use it anyway, to the institutional church? I don't think it takes long. And, I think you have to constantly fight against it. I think you have to talk about it all the time, because it's SO easy for us, all of us. And, my thing might not be the carpet, but we've all got our things that are part of, and it takes us back to that Wesleyan statement⁵². You know, what are the

-

⁵² The statement Ted is referring to here is a quotation he attributes to John Wesley and cites often: "In essentials unity, in non-essentials freedom, and in all things love." The statement actually originated with St. Augustine.

essentials? What are the things that we must have? And, what are the non-essentials that are negotiable for the sake of Christ, and for the sake of ministry?

Eddie: I could tell a story. Muhammad called me last night. And, he was disturbed by this. He said, he wanted to talk to me about something that happened on Sunday. He said that, he's bringing people to the church. And, he always tells them, "If you have needs, the church will help you." And, "Do you need anything for your house?" And, a lot of them do because they're just trying to get settled. So, he's been taking people down to the basement and giving them things out of the household bank. And, he said to me, look, "If I give away a pillow and someone comes to Christ. Isn't that a cheap thing to do? But, Robert [[the church maintenance engineer]] saw me giving away a dollar pillow and he scolded me in front of these other people." We can say these aren't essentials and we shouldn't think about them and we should keep focused. The fact is there are conflicting things that are concerns here, that the things that we are concerned about may be significant. And, maybe they aren't.

Ted: Well, and, I think it's an important one. And, I think there just needs to be ways in which that gets done, just needs to be communicated. But, yeah, that's great.

Eddie: Tied in with this was the cultural thing of not showing respect to someone. He felt Robert was disrespectful because he told him he shouldn't do that in front of other people who were new in the church, coming for the first time.

There are several things about this interaction that typify the way the Tuesday Men incorporated the refugees into their discussions. 1) Eddie's closing statement is one that appeared in various forms in his comments and in those of others: an incident involving the refugees is explained by "cultural" differences. In this case, Muhammad's annoyance is attributed to Robert's unwitting violation of the cultural norm of respect. 2) The seemingly transgressive acts of the refugees are not presented negatively, but in a productive fashion. The refugees were viewed as demonstrating a valuable lesson to the rest of the church. The above example is framed by Eddie as one of valuing faith conversion over mundane items (e.g., a pillow), placing Muhammad in fulfillment of Wesley's guiding dictum to focus on the "essentials" of the Christian faith. 3) The lessons being taught by the refugees parallel the ideal presented in the book of Acts. In this example, the conflict between Muhammad and Robert is positioned as a mirror image of the contrast between the early Church of Acts and the modern church. As presented in chapter 3, this group's discourse continually asserted a dissonance between

the Church in Acts and contemporary congregations. The use of the refugee experience as a resource, then, is done in alignment and support of the group's primary interpretive relation with the Bible.

From the group's opening session, Ted, Eddie, and the other participants found ways to incorporate their experiences with refugee in this manner. Table 9 indicates the number of occasions during the 23 recorded meetings the refugees were invoked as a resource compared to the most common alternatives:

Refugees	Denomination	History	Current Events	Business
9	6	5	6	4

Table 9: Tuesday Men Interpretive Resources

As well as being the most numerous, ICUMC's relation with the refugees was the only resource to be invoked by four different group participants. The individual refugee being used as an example changed, as did the foil - a role often occupied by the speaker. On one occasion, Eddie lamented about a prayer of his that focused on solving a problem instead of how God might use the problem. He presented this in contrast to a prayer by Muhammad that focused on heavenly rewards, which paralleled a prayer from Acts that they had just read.

The local congregation was a shared resource in each of the project groups. In the McGee Men, for example, the changing financial fortunes of the church – from being notably affluent to representing a wide range of socio-economic tiers – proved a valuable resource for reading texts regarding "riches" and "favoritism" in the book of James. In the LC-MS Men, the impending start of a new church building project was a target for sayings of wisdom in the book of Proverbs. This involvement of the congregational situation into the interpretive matrix testifies to the notion of congregations as "local"

cultures," providing shared and readily accessible stocks of knowledge to members (cf., Ammerman 1997, Becker 1999). Viewing congregations as "local cultures" was originally done to demonstrate how churches provide resources for everyday and interpersonal needs. My argument here is that the local congregation is also an interpretive resource, a body of knowledge that serves as a filter for discursive interests.

Denominations as Resources

A second resource shared across the entire sample was denominational identity.

Table 10 shows how many times this resource was invoked in each of the project groups⁵³:

<u>Iconoclasts</u>	McGee	Tuesday	LCMS	LCMS	Prayer	Monday
	Men	Men	Women	Men	Circle	Group
17	6	6	24	10	2	2

Table 10: Interpretive Resources of seven project groups

I use two contrasting cases of this resource to illustrate its variable function in the interpretive matrix. The first example comes from the Iconoclasts, where the facilitator's Baptist background played a prominent role in establishing what constitutes a Christian Other. The second example comes from the LC-MS Women, where the texture of a Lutheran Self was constantly re-evaluated. For both, the proper relationship with the Bible anchors their conceptualization of denominational identity.

Denomination as Other

The Iconoclasts were the very first group whose meetings I began to attend and record. Janet, the small group director at SUMC, connected me to them. In our first meeting together, Janet and I reviewed all of the small groups in the church. Immediately, she highlighted the name of the Iconoclasts in yellow, guaranteeing their excitement

⁵³ The lack of references among Prayer Circle and the Monday Group are primarily a reflection of the deemphasis placed on "denominations" as a category of identification in the Restoration Movement tradition.

about being included in the research. She was quite right, though perhaps not for the reasons she suspected. During my initial visits in the fall of 2004 they were finishing a book-video study that was a church-wide initiative. Aside from ridiculing the "corny" setting, the group found little enjoyment in the text. This was the second consecutive time they had disliked Janet's choice of study material. The group members unanimously decided that they would choose the next book themselves. Charlie, the group facilitator, already had a suggestion in mind. His wife, Diane, had read and recommended Philip Yancey's The Jesus I Never Knew, which I henceforth refer to as Jesus. Charlie had read the book twice and was convinced the group would enjoy it. In early November they began Jesus, a study that lasted until the end of May.

Yancey is a best-selling Evangelical author, and <u>Jesus</u> won the Gold Medallion Christian Book of the Year Award in 1996. The following quote from the book's introduction provides a sense for its premise:

"That...is the problem with most of our writing and thinking about Jesus. We read the Gospels through the flash-forward lenses of church councils like Nicea and Chalcedon, through the church's studied attempts to make sense of [Jesus]...In this book I hope to go back beyond those formulations. I hope, as far as is possible, to look at Jesus' life 'from below,' as a spectator, one of the many who followed him around. If I were a Japanese filmmaker, given \$50 million and no script but the Gospels' text, what kind of film would I make?" (1996: 24)

Yancey divides the book into 14 chapters and three sections: "Who He Was," "Why He Came," and "What He Left Behind." Yancey presents a highly intertextual work, incorporating excerpts from the four canonical Gospels and lengthy quotations from Christian writers and thinkers.

The group loved the book, much as Charlie and Diane predicted. They met weekly on Sunday evenings, eager to discuss the chapter they had read during the week.

On several occasions, they decided to extend a single chapter over several weeks. To

organize the meetings, Charlie passed out a list of (6-14) questions for each chapter. Each question was accompanied by a (1-3 paragraph) reading from <u>Jesus</u>. The group was never strict about covering all of the questions, and Charlie took the lead in deciding which ones they addressed.

The group's discourse revolved around one theme, one very reminiscent of the Tuesday Men's focus: the Jesus of the Gospels is very different from our contemporary understanding. This is an intended theme of the book, but not the only theme. Yet, it is the only theme the group latched itself to. They observed this separation at several locations: the modern Church in the United States, the local congregation the group belonged to, and the religious background of the participants. This latter category is where Charlie's denominational history became a resource for himself, and eventually for the rest of the group.

Compared to many of the project groups, the religious and social identity of their denomination was of little concern to the Iconoclasts. But, from the very first meeting, Charlie's youth and early adult years in an Independent Baptist church became a baseline for 'how not to be Christian.' This example emerges from a reading out of <u>Jesus</u> discussing the unfathomable extent of God's grace:

Charlie: I know in my own personal case I think the Lord has really worked on me over the last ten to twenty years and every shred of self-righteousness that I have, one piece by one piece, has been taken away. And, fortunately, for me, I think I now see me for what I am; good, bad, and ugly. The piece that I'm working on now is that God actually accepts who I am, regardless of the shape that we come in. It's a revolutionary thought. Didn't you find that interesting where it says the two principles of Jesus' life, the first thing he does is he shows us all our sin in glorious Technicolor, but yet, he freely forgives us. This is a very tough thing for a Baptist guy like me to understand because I want to get out my little Boy Scout notches and do the right thing by the Lord. I'm much more comfortable with a performance-oriented God. That's kind of where I would like to live, actually. Because I think, at least in terms of a show, I could do as good a job as anybody. When I was in high school I was in my self-aware, self-righteous phase. And, I used to drag a Bible to school. Now, I never read it. But, I would drag it to school and I would put it on top of my books.

Lauren: As an accessory?

Charlie: As an accessory, a fashion accessory. And I'd drag it around and everybody would say, "What's that?" And I'd say, [[switch to deep, resonating tone]] "That's a Bible."

[[Laughs]]

Diane: Nobody ever said to you, "Open it and read it?"

Charlie: No. You didn't worry about that. So, I find this whole concept absolutely revolutionary, that God would look at me with all my sins and, you ever sat down and done a sin inventory in your life?

[[No response]]

Charlie: Exactly. I could, five minutes; I think I could fill up a spreadsheet. I really could. I'll never forget this, there's a verse in the Bible that says, and the Lord takes our sins and places them as far as the East is from the West. One of the greatest, the reason I wanted to stop and talk about this for a while is, at least in my church growing up, here's one of the greatest tools that I've seen pastors really use, pastors and other people including self-righteous Nazi Christian types, is the whole thing of guilt.

No Sunday evening passed without a few exchanges like this, initiated by Charlie or by someone else. The Baptist tradition served as a reliable representative of a Christian Other – the type of Christian not to be aspired to. The face of this antagonist was prone to shift. The above Baptist Other emphasized performance and guilt over grace. On other occasions, "Baptists" were used to illustrate a stress on fear over love, determinism over free will⁵⁴, and denial over forgiveness. Charlie drew on a seemingly endless set of anecdotes and idioms, and the rest of the group appropriated Charlie's experience when searching for their own Christian Other. The effectiveness of this resource intensified once the group realized that Yancey, like Charlie, grew up in the Baptist tradition and used his experiences as a foil for presenting Jesus in the book⁵⁵.

The fundamental assertion at work in the group's discourse is that the Jesus (and, by extension, the Christian life) advertised in the modern era is becoming further adrift

⁵⁴ This is a historic and ongoing theological difference, and one that often assumes new nuances from the discursive environment it is carried out in. In this group, theological determinism was associated with the popular notion that God has a plan and a purpose for everything that happens in the lives of believers and in the world at large. They often linked this to the best-selling book *The Purpose Driven Life* (2003), which the group studied (and disliked) as part of a church-wide curriculum.

⁵⁵ It is worth noting here that "Baptists" encompasses a large, heterogeneous group of Protestants in the United States, with multiple sub-denominational affiliations. Charlie's point of reference is an Independent Baptist Church, not affiliated with any of the sub-groupings of "Baptists."

from the Jesus of the Gospels. The underlying claim, then, is an issue of dissonance with the Bible as a historical text, not just the personality of the historical Jesus. This casts the group's reliance on denominational identity as a resource not only for engaging Yancey, but also for engaging the Bible. "Baptists" serve as an embodiment of a Christian Other, which in turn serves as a Biblical Other. "Baptists" were often used as a narrative equivalent for Jesus' encounters with the religious Others of the canonical Gospels, as in the following:

Charlie: Tell you another story from my days in the Baptist Church. And, this was actually taught. I listened to this in sermons. "Well, you know, behavior ultimately doesn't matter because you get your fire insurance policy going. And, you get into heaven. But, in heaven, there are degrees of rewards." This is taught. So, I thought, years later, I thought, okay. You don't need it to get in. But, you really need it because you don't want to be living in the shack down by the railroad_tracks.

Lauren: And, you want those rewards.

Charlie: That's right.

[[Jokes about Charlie's competitive streak]]

Charlie: But, you know, so I, in these latter years I've chalked that up to a propensity to want to rank everything and put ranks and social levels on everybody and everything. It's just something we like to do as humans. And, indeed, this stuff made, this whole grace thing, if you read the Scriptures, made the Pharisees and the Sadducees absolutely nuts. They went nuts, because it threatened their religious order and threatened their little religious house of cards. And, indeed, for a person who's, what I call, hyper-religious, someone who's completely hung up on how you ought to behave, that is their chief measure.

Just as the Pharisees and Sadducees are seen as failing to grasp Jesus' teachings,
Charlie's "days in the Baptist Church" reveal a similar failure. Clearly, then, the position
of this resource in the interpretive matrix is involved in the work of textual ideology.

Much like the Tuesday Men, the task for the Iconoclasts is to construct a Biblical Other
in order to establish what is separate from the ideal presented in scripture. This contrast
relies on both the absolute authority of the Bible, and its timeless relevance.

Denomination as Self

The LC-MS women met every Thursday morning at 10am in the church library. The regular group of 10-15 women fit snugly around the large rectangular table in the middle of the small room. Shelves packed with books, videos and Bibles sat only a few feet behind our heads, giving the room a cozy and somewhat scholarly feel. The youngest woman was in her early 50s, but most were in their mid-60s and older. The facilitator of the group, Sandy, was a life-long LC-MS member (like most in the group). She had a marvelous rapport with the other women. And, despite her self-deprecating manner, she was thought of as a highly competent facilitator.

When I made my initial visits to the group in the late fall of 2004, they had just begun a new study. To Live is Christ: the life and ministry of Paul is an interactive, "indepth women's Bible study" by the popular Evangelical author and speaker Beth Moore. The study guides the reader through Paul's ministry in the book of Acts, while splicing in readings from the Pauline Epistles. The study is designed to last 10 weeks. In between weekly group meetings, there are five days of reading and questions. For each group session, there is a video that supplements the week's teachings. However, Sandy worried that assigning "homework" might discourage some from attending. So, they adopted a different method of working through one day's reading and questions at each meeting, turning the 10-week study into a 50+-week study. When I began recording in January of 2005, they were into the third week of the study (Acts 14).

The women met at 10am, but never started until 10:30. The first half hour was reserved for socializing, coffee and tea, sampling culinary creations, and (much to Sandy's chagrin) women arriving late with an apology and a story. One morning, early in

the fieldwork, the interpretive matrix allowed for a bit of laughter. Sandy confirmed with several women that they were enjoying the study. wryly pointed out that she liked the study despite Moore being a Baptist, and not Lutheran. A few women laughed off the comment, but several others turned to Sandy for a response. Ever aware of her leadership position, Sandy turned to me and quipped, "I won't tell if James won't." Trying to match her sharp wit, I replied, "They won't like hearing about this in St. Louis." I without missing a beat, Sandy rejoined, "I didn't know you knew about St. Louis." I wondered afterwards if our laughter disturbed the office managers across the hall.

My jovial exchange with Sandy indexes the central feature of the interpretive matrix for the LC-MS Women: a reliance on 'being Lutheran' as a discursive resource. Unlike Charlie of the Iconoclasts, the group members did not employ denomination as an Other, but as a resource for an ongoing negotiation of their religious Self. The marking of textual materials alone was remarkable. The denominational affiliation of Bible commentators, authors, speakers, hymnals, and Bible translations were constantly being identified; a quality unique to this group. The group members also frequently defined their denominational Self in terms of how it contrasted or conflicted with the traditions of Christian Others. Invoking denominational identity most often preceded or followed mention of how "Lutheran" contrasts with "Catholics," "Reformed churches," "New Churches," "megachurches," and a host of other Protestant categories. The following is a typical example from the group. They had just finished reading Beth Moore's explication of Paul and Barnabas's trip from Derbe to Antioch (Acts 14:21-28):

"Before we penetrate the heart of our lesson today, note the obedience of Paul and Barnabas to the great commission (see Mat. 28:19-20). Acts 14:21 tells us they ministered to the people in Derbe in two vital ways. They preached, which means 'to

-

⁵⁶ St. Louis, Missouri is the denominational headquarters for the Lutheran Church-Missouri Synod.

evangelize,' and they taught or *matheteuo*, which means "to instruct with the purpose of making a disciple.' Paul and Barnabas were careful not to neglect evangelism or discipleship. Both are vital, life-giving elements in any New Testament church." (1997: 64-5)

Sandy stopped the reading there and opened the topic for discussion:

Sandy: One commentator I read suggested that Matthew 28 had the sense of not just teaching, but also discipling. Setting up churches, discipling, setting up an order in the church, a way of continuing.

Linda: See, I don't think that order is here today either. I think a lot of churches, new churches that are popping up, I don't think they're orderly. You know, they're fun, but I don't think there's order.

Sandy: Do people get baptized? Evangelized?

Jov: Adults do.

Sandy: Taught in a relatively stabilized fashion.

Lynn: But, taught, [[sighs]] I mean you would oft, oft, oft times never hear a word of Law in months. It's much more Gospel and feel good, and how to be good people.

Frita: Yeah. And, sometimes I don't think they, I mean we call it structure. Just to learn what our church stands for when you're first going. I think some churches, people just join and they don't really know what the basic beliefs are in that denomination.

Lynn: Well, the word of Gospel is not nearly so sweet unless you realize the, the Law and the sin, and then you realize the sweetness of Gospel.

Sandy: It's interesting that the kinds of churches you might be talking about have very, kind of fairly strict lifestyles, or covenants. If you want to join that Assembly church across town, you'll covenant with them not to do certain things.

Linda: Yeah. I'm thinking of the much more casual churches.

Sandy: Like the Vineyard churches?

Linda: Yeah. Mhmm. And, we have one, in fact I went to a travelogue and I looked down and there was a bulletin. So, I was, during the intermission I was looking at the bulletin for this [local church]. And, you know, it was a lot of nice. And, grace is important, oh my. But, that word of Law should still be there.

Sandy: Remember when members of Zion came over, we had the green hymnal, that's the ELCA hymnal...The order of service was such that the confession of sins is separate from the service. And, you can have it or not. A number of times our pastors, they didn't choose to have a confession of sins.

Frita: Well, I have a friend who will not say the confession of sins because she says she is not a poor, miserable sinner. She said, [[defiant tone]] "I'm not." [[Laughs]]

The group's reaction to Moore's explanation immediately centered on the resource of denominational identity. Linda begins with the juxtaposition of "order" and "fun," with "new churches" occupying the latter designation. Frita joins in, noting the lack of "structure" in "new churches" and the centrality of this to the LC-MS tradition.

Sandy and Frita both highlight the aversion to the "confession of sins" in the ELCA⁵⁷ liturgy. The use of local representatives for these categories, such as Linda's reference to the "[local church]" and Sandy's reference to "Vineyard churches," was common in other interactions of this type. Of all the oppositional categories to "Lutheran," "new churches" (or, the semantic equivalent of "megachurches") was presented most often. Judging from the group's discourse surrounding this category, its recurrence is a result of new churches' success in the American religious economy. The implicit question is twofold: how different are we from growing churches? How much of that difference is key to Lutheran identity, and how much is negotiable for the sake of membership growth?

A second example crystallizes this Self/Other relationship, and introduces the link with textual ideology. This interaction follows the group's reading of 1 Corinthians

16:19⁵⁸, which Moore suggests to contextualize Paul's ministry in the city of Ephesus:

Sandy: So, what does that tell you is going on there? What are Aquila and Priscilla doing?

Mary: They were hosting meetings.

Sandy: House church, just like home groups.

Mary: They would probably have had communion at theirs, wouldn't they?

Sandy: Mhmm.

Mary: Didn't they always sit down and break bread together?

Sandy: Almost never without. We don't do that here because of our rule about ordination, and the way we treat the sacrament. It has to be blessed by an ordained pastor. At our previous church, the elders were allowed to take the communion around to the shut-ins. But, the pastor would consecrate the elements first, and then they could take them out. We'd have communion with these shut-ins, which was really very nice.

Mary: Mhmm. It is.

Sandy: So, here we are, people of house churches, just like our home groups. And, in fact, in the home group concept, because it started in an evangelical background, it's not as sacramental as we are. Home groups and say, the Assembly church, I imagine that they have them. Or, Trinity, they would have communion in the home groups, because they don't consider that you need to have that sacrament. It is a sacrament, but it's not as regulated.

⁵⁷ Evangelical Lutheran Church in America. See chapter 1 for a discussion of the relation between the LC-MS and the ELCA.

⁵⁸ "The churches in the province of Asia send you greetings. Aquila and Priscilla greet you warmly in the Lord, and so does the church that meets at their house." (NIV)

Linda: I don't think it is a sacrament. They call it "public communion." But, they don't call it, like, we talk about "Word and Sacrament."

Sandy: Mhmm.

Linda: They do have baptism, but they're adults. But, I don't think they call it the sacrament of baptism.

Cindy: I think the big thing with those churches, and I speak for the Assembly church, is having people profess that they believe Christ is their Lord and Savior, and that they are born-again or whatever. You have to profess this. I know with the motorcycle group that my daughter is involved with now. They get them to come to church and they take off their patches that say all these horrible things. But, to be part of this group you have to do this. You have to believe that Christ died for your sins. This is what they are really wanting these people to do. Whether they just say they do and then they ride off into the sunset, I don't know.

Linda: But, it's "what I did." "I came to you." And, we say, "We were called BY the Holy Spirit." And, I think that's the difference. It wasn't what I did to come. It was that "the Holy Spirit called me by the Gospel, enlightened me with His gifts, sanctified in"... [[overlapping]]

Sandy: "In the one true faith."

Again they draw the comparison with "evangelical" churches, using the local representatives of "the Assembly church" and "Trinity," the two largest congregations in the area. Here, the juxtaposition centers on the "Lutheran" concept of "sacrament⁵⁹" and the lack thereof in these Other traditions. Denominational identity is used to make sense of a Biblical text, affirming the legitimacy of the Lutheran position. What I find most striking about this interaction compared to the first one I presented are the contrasting means by which they construe the proper relation to the Bible.

In the first example, the group's use of denominational identity automatically places them in accordance with the Biblical text: the "order" evident in Matthew 28 and Acts is comparable to the "structure" of Lutheran theological training. This is a notably different approach to the Bible than that of the Iconoclasts and the Tuesday Men, which emphasizes dissonance. The textual practice here is to identify where the LC-MS

59 The LC-MS tradition considers itself sacramental because it views particular rituals, such as baptism and

communion, as mediations of God's grace. Other Protestant traditions consider these rituals as symbolic acts that merely signify God's grace. The sacramental quality of rituals is a point of contention in the Lutheran traditions in America. See Roberts (2005) for a discussion of how these interpretations in two different Lutheran churches from the same denomination are largely a reflection of each congregation's local history.

tradition aligns with the Bible, not where there are inconsistencies. In the second example the conclusion is the same, but it begins quite differently. The Biblical example suggests that first century home groups practiced communion, which Sandy identifies as different from standard Lutheran practice. In addition to the dissonance on the Lutheran side, the "evangelical" Other appears consistent with the Biblical example. This seemingly favorable position of the "evangelical" approach is quickly dismissed. Cindy and Linda emphasize, respectively, the tenuous and misplaced motivation of the "evangelical" approach. The closing recital of Lutheran creed by Linda and Sandy restores the Lutheran approach to its elevated position, reasserting the parallel between the Bible and their denomination.

These two examples illustrate the ideal relationship with the Bible articulated by the group through their use of denominational identity: establishing consistency between scripture and the LC-MS tradition. However, to present such a uniform portrait would be incomplete. The humor behind the opening anecdote, and the repair work in the second example, belie the ongoing and negotiable nature of the group's conceptualization of LC-MS identity. On several occasions, the ambivalence surrounding their sense of Self was drawn into sharp focus. Consider the following exchange, which led to a lengthy discussion:

Barbara: Beth Moore asks us, upon whom did God say He would pour out His spirit in Acts 2:18?

Jane: Well, women.

Barbara: If that happened so long ago, why does, for example the Lutheran Church-

Missouri Synod, not let women read or let women do anything?

Rebecca: Well, there is another verse, though, that says the women should not be teaching over the men. It doesn't mean, and personally I always thought there's plenty for the women to do.

Barbara: Oh yeah.

Rebecca: We have so much we can do we don't have to lead men's Bible studies. But, I think men and women are equal, God made us differently and I think the men think and process in different ways, most of them. And, it's an advantage.

Barbara: I love that description, "They process differently."

Jane: They do.

Rebecca: And, we do too. Dr. Dobson's⁶⁰ thing, you know, when you're shopping. Have you heard that one? "Women shop." [[elongated stress on the [o]]] [[Laughs]]

Rebecca: And, he describes going shopping with his wife and he says, "When are we gonna find this blouse?" She wanted to stop and have lunch. He couldn't have lunch until they got the blouse. He was on a mission.

[[Laughs]]

Rebecca: You gotta have a goal and go right to it, and you can just stay focused on like one thing. Where, we're made differently because we might have lots of children around or grandchildren or we have 20 things to do at one time.

Barbara: That's right.

Judith: Men are able to compartmentalize their thoughts.

Rebecca: I'm more than halfway through life and I think they're different.

Sherry: I think in ways they're certain things. But, I think sometimes that that's just the way our life, whatever we've chosen, that we are like this. So, you agree with the Missouri Synod, that women shouldn't be involved in, you agree with that? I mean, you really think that that's what God meant, that women are not to be preachers?

This discussion continued on amiably, back and forth, primarily between Rebecca and Sherry, for nearly ten minutes. Barbara is quick to place Acts 2:18⁶¹ into the interpretive matrix of denominational identity. Unlike the other examples, the LC-MS relation with the Bible is not agreed upon. The proof-texting continued briefly, but the discussion soon turned from discerning "what God meant" to what the women "feel comfortable with," relieving them of any resolution or consensus. The lesson of this example is twofold. First, the resource of denominational identity is invoked at times with a certain ambivalence, indexing the contestable nature of the interpretive matrix.

And, secondly, sustaining a consistent relationship to the Bible (in this case, establishing

⁶⁰ Dr. James Dobson is an iconic figure of the conservative Christian movement in America. He is best known for his institutional structure "Focus on the Family."

⁶¹ "Even on my servants, both men and women, I will pour out my Spirit in those days, and they will prophesy." (NIV) This particular text is a recontextualization from the Old Testament prophet Joel (Joel 2:28-29).

parallels) takes precedence over resolving particular debates or extra-Biblical epistemologies (e.g., women as preachers and accompanying gender ideologies).

This opening section presented two resources that were part of the interpretive matrix in each of the project groups. Conceptualizing identity in terms of the local congregation and the denomination played an important role in how groups evaluated texts and engaged topics of discussion. In both case studies, I showed how these resources unfolded in group discourse, as well as how they were implicated in the core issue of textual ideology. This theme continues in the second and third examples of how the interpretive matrix operates.

Group-Specific Resources

In each of the project groups certain resources took precedence, and played a more active role in their interpretive discourse. Here I focus on one case study, the Iconoclasts. Charlie frequently appealed to his Baptist background as a resource, and the other members appropriated it. However, denominational identity was not their most consistent, widely shared resource. Table 11 shows the variety of interpretive resources the group relied on, the number of occasions they were invoked during the 16 recorded meetings, and (in parentheses) how many of the group participants made use of each resource⁶²:

Congregation	Denomination	History	Televangelists	Christian Authors
5 (2)	17 (5)	30 (7)	10 (3)	9 (2)

Table 11: Iconoclasts' Interpretive Resources

As Table 11 indicates, the interpretive matrix for this group is best characterized as historically oriented. The members shared a common passion for employing examples

⁶² Table 11 excludes a number of resources that were invoked only once or twice, such as "business experience" and "current events."

from modern and ancient history; though, the specific topics and periods varied. This is not a total anomaly in the project sample, and certainly not in American society generally. A walk through any Barnes&Noble Bookstore shows that the "History" section is always among the best stocked and up-to-date⁶³. Prior to the start of each Sunday's meeting, the group often traded titles of the latest books they had read from this genre. Charlie and Danny often joked that heaven would be akin to "perpetual History Channel and Band of Brothers movie night." Their choice of study after Yancey speaks to this inclination as well: a video study based on archeological sites from the Old Testament. Contributions to group discussion were more likely to include reference to the latest history book they read than a Christian motivational or theological text.

The influence of history as a knowledge source was most evident when the group faced ambiguous questions, or cases where an answer was unclear. In such cases, the reasoning and explanatory mechanism used was the epistemological category of history. The following example comes from the group's first meeting. Prior to reading the first chapter of Jesus, Charlie devoted the initial meeting to a series of questions meant to elicit members' conceptions of "who Jesus is." One question dealt with the physical appearance of Jesus:

Dwight: One thing that always seemed to bother my mom was all the pictures of Jesus, he's never smiling he was always serious.

Charlie: That's the sixties Jesus, though. That was what I called the heroine, drug addict Jesus, you know.

[[Laughs]]

Charlie: You know, like that popular image of movies during that time. Jesus always looks like he hadn't had sleep for about ten days and he's got this kind of half-crazed look. And that's kind of a popular image that you don't see except in that genre, like the whole issue of Jesus being in "The Passion" as we see him. If you think back to the movies in the sixties and the seventies that portrayed Jesus like, you wouldn't have seen that, you know.

⁶³ I owe this observation to Dr. Fredric Roberts, my advisor. He pointed this out to me several years prior to this project, and I continue to find it accurate.

Lisa: "Jesus Christ Superstar."

Charlie: Yeah, yeah. So, you know, it's really interesting how our pictures of Jesus, Yancey will say that, our pictures of Jesus are so colored by our culture and the world that we live in, kind of stage that we live in, it's really tough to see the actual Jesus, you know.

Terry: I always thought of him as being very muscular, strong, and athletic with the idea that he would have been a carpenter at a time when there wouldn't have been... [[overlapping]]

Christine: Right.

Terry: You know, the tools that are available now. So, I thought he had to have been very strong to do that, plus with all the walking that they would do from place to place and as much traveling as he did. I just always pictured him as being, I'll say almost an athletic type of build in order to be able to do the things that he did.

Charlie and Terry both gravitate toward history in their responses. Terry's especially is typical for the group. His way of contextualizing and piecing together a scenario based on historical probabilities was very much the norm when members were faced with such questions. A second example demonstrates this strategy as shared amongst the other group members. The following interaction comes from the second meeting of the Yancey study. It is particularly interesting because it is the only case where the group expressed a collective disagreement with Yancey, perhaps indexing the authoritative character of this resource:

Charlie: Nine months of awkward explanations, the lingering scent of scandal – it seems that God arranged the most humiliating circumstances possible for his entrance, as if to avoid any charge of favoritism. I am impressed that when the Son of God became a human being he played by the rules, harsh rules: small towns do not treat kindly young boys who grow up with questionable paternity... Often a work of God comes with two edges, great joy and great pain, and in that matter-of-fact response Mary embraced both. She was the first person to accept Jesus on his own terms, regardless of the personal cost. Lisa: I think he's losing a lot of Joseph's role in this whole dynamic because Joseph was pledged to Mary, at least Mary was betrothed to Joseph. So, it was already pre-planned that they were to be married. And, he stood up for her and said, "I will accept you as you are," once she told him she was pregnant.

Charlie: Yeah, once he was told in a dream by the Lord. He was actually going to put her away quietly.

Lisa: Okay, so he had the intervention from the Lord to say, "Hey, she's on the up and up level, she is carrying my child." So, then he had that extra role of standing up to the community to say, "This is my betrothed, this is my wife. This is my child. He is my son."

Charlie: Yeah, it's real interesting, Joseph and Mary, these were not people of high social status.

Lisa: And when did he, when did they finally get back to their home community. Because, okay they go off to Bethlehem because of the census, so he's leaving his home community then. And, then they go off to Egypt for a couple of years. So, I mean, so it's got time for the scandal to really die down by the time they would have circulated back to their home territory.

Lauren: Yeah. And, not only that, they're moving back to Nazareth. And Nazareth was like the South side of Chicago, like the slums.

Lisa: But, he was a skilled craftsman. So, at least he could provide for his family. He had the means; I mean it wasn't that it was all poverty. He had a good income for the community that he lived in.

Jill: He had to move so people wouldn't kill him.

Lisa: And actually, to have the baby born in the stable was the cleanest place he could have been born. Because it was clean straw and I mean, like with all this ingathering of all these people that have been spread out for this census purpose. It would have been a filthy dirty place inside the inn. The stable was actually the cleanest place to be.

Lauren: But, what was the society like then? Would it have really been that big a deal if, I mean, you kind of assume that everyone thinks this is Joseph's kid.

Terry: Right.

Lauren: This is something so, the only two people that have to know anything different about it are Mary and Joseph. Everyone else is going to think it's Joseph's. So, I mean, you really only had two people that had to really know what it was. So, I'm not sure it was even all that big a deal. I mean, how many young girls became pregnant back then? I imagine it was very similar like it is today.

Charlie: Yeah.

Lauren: So, they were poor and they were a lot of things. But, whether it was, she was going to get put off in a loony bin I guess would depend on, you know, did she go around saying this or was it just, other than Elizabeth?

Lisa: And, even betrothal at that time was a much deeper commitment than just being engaged is in today's society.

Here, Lisa and Lauren lead the historical critique of Yancey. Much like Terry's reliance on historical context to fill in a portrait of Jesus, they do so to understand the circumstances surrounding the birth of Jesus. They cite historical explanations of betrothal, being a carpenter, stable sanitation, and teen pregnancy as proof against Yancey's assertions. This constant reference to history made this resource a current that ran through the group's discourse. This second example is also instructive in regard to the relationship between history as a resource and its implication in the textual ideology surrounding the Bible. The link is embedded in Lauren's statement towards the end of the interaction: "I imagine it was very similar like it is today."

The group's interest in history works in conjunction with the belief that human nature – and, by extension, human action and motivation – is constant across time. This also was a refrain in their discussions. In this respect, the use of history as an interpretive resource incorporates textual ideology principles regarding the authority and relevance of scripture: the Bible reveals the moral nature of humanity, and so is continually instructive for making sense of human problems. The following are two brief examples from the group's reading of Yancey. The first deals with the issue of purity, and the second with miracles:

#1

Charlie: Impurity is not only sexual. Purity is other things. Purity is our attitude towards money. It's our attitude towards others. There's lots of ways to be impure. We just think about sex all the time because that's what we do in the United States. But, it's many ways, having a pure heart. It's like if you obsess on anything else. Maybe this is a good working definition, if you have an obsession you can't see God clearly because you're thinking about that other thing.

Lauren: But, part of the problem with that is mankind was built with that innate urge to continue to multiply the whole thing.

Charlie: Mhmm.

Lauren: It's not something that's just now. If you look back in history to the things that went on, we maybe didn't have the media, but there were a lot of other things that went...[[interrupted]]

Charlie: That's Diane's point. Diane tells me that all the time.

Lauren: You go back and you read, or even, what did we watch, the Vikings or something when they came in. What were they going to do? Rape and pillage.

Christine: It's nothing new under the sun.

Diane: Man and evil and temptation and all those things that have been around forever.

#2

Charlie: Jesus consistently refused to tell people, "You gotta do this or I'll," you know. And he could have. He was God. If God really wanted people to behave, all he'd have to do is line up a demonstration.

Dwight: What would it have mattered? I mean, what did, people got freed from Egypt.

Lauren: Wouldn't have mattered.

Dwight: Out in the middle of nowhere, they got food out of nothing. And, they are still casting rams and stuff to worship.

Lauren: Right. And, the miracles that Christ did. Some people it was odd and moved them, and others just wanted to know what kind of trickery and who, you know, it was done by the devil.

Charlie: Yeah. Ravi Zacharias, on the radio the other day, said exactly the same thing. He quoted a Scripture of, I think it was John, where Jesus said, I sent you the prophets. You didn't listen to them. I sent you John the Baptist. You wouldn't listen to them. So,

what makes you think you're going to listen to me? Because we always want to do what we always want to do, and we always want to get what we always want to get. We want to believe what we want to believe.

Lauren: Even after having proof right in front of you. I mean, how many times have you tried to prove to someone that what, whether it's wrong, and you have undeniable, it's proof, and they still won't believe you. It doesn't matter to some.

Charlie: Years ago we had a family member staying with us and they had a little boy named Jack. Jack was at the house, and he was not obeying his mom. And, his mom said, [[angered tone]] "Okay. That's it. I'm taking you up the stairs." And, all the way up the stairs he goes, [[defiant, obnoxious tone]] "No. You're not in charge of me. I'm in charge of myself. You're not in charge of me." And, I looked at him going up the stairs and said, "That's us. That's all of us."

[[Laughs]]

Charlie: I had just finished reading something in the Old Testament about God calling the Israelites a stiff-necked person. If you ever want to read what you are like as a human being, what we're all like as a human being, read the Old Testament. Read the New Testament. It's all there. Every little trick, every little devious evil thing, we can all do it.

The textual ideology regarding the nature of humanity is evident in both of these interactions. Lauren introduces the notion that the problem of impurity is not an 'American' phenomenon, but a result of mankind's "innate urge." Charlie and Diane affirm her assertion, as does Christine via the proof-text from the book of Ecclesiastes. This particular verse was a common, shorthand way of getting this notion across. Interestingly, given this group's unique reliance on history, it was also the only group where this particular Biblical text was recontextualized. In the second example, Dwight posits a commonality between the Old Testament exodus and the modern reaction to miracles. Lauren agrees, and Charlie suggests a parallel between "stiff-necked Israelites" and a ranting adolescent. Charlie concludes the discussion by making an explicit statement about the mirror image of Biblical characters and human nature. These discursive constructions reveal the group's orientation to history as tightly bound to the ongoing work of textual ideology. In this case, history as a resource embeds the ideological principle that the Bible exposes the moral nature of humanity. Ultimately, this

makes historical reference an authoritative means for framing discussion because it echoes the lessons of scripture.

In the course of their time together, groups come to rely on a set of resources that comprise their interpretive matrix. Some are predictable no matter the type of group, text of study, or make-up of participants, such as congregational or denominational life. In other cases, resources that are marginal or absent in some groups are an essential lens through which discourse passes in other groups. For the Iconoclasts, that resource is the use of examples from various historical periods. To complete this sketch of the interpretive matrix, one type of resource remains to be discussed.

Individual Resources

Interpretive matrices also include resources that are unique to individuals. Such resources are not shared throughout groups, but participants remain willing to invoke the resources, other members are willing to receive the resources, and (most importantly) the connection between the resource and the ongoing work of textual ideology persists.

Here, we return to Rick and his experience as an "independent business owner." Since 1999, Rick has been involved in the business: buying and selling products, recruiting new members, and attending seminars. As I stated in the opening of the chapter, Rick was the only person to facilitate two different classes that I observed and recorded. He designed and taught one as part of RMC's Sunday school curriculum. The second was a home group he and his wife hosted on Monday evenings. During my research tenure, this latter group worked through Hebrews, 1 John, 2 John, and 3 John. Rick incorporated his business experience in different fashions with each group.

Rick's Sunday school class is entitled "The Mind of Christ." As described in chapter 1, Rick created the class following a speech on "understanding the mind of non-believers" by an ex-congressman. Rick also cites an audiotape series by Earl Nightingale, "The Strangest Secret," as a major influence. Nightingale was an ex-insurance agent who became a leading figure in the self-help movement of the 1940s and 50s. The strangest secret is that "you become what you think about." Rick identified this as a Biblical concept and, by the end of the research, had heard the "secret" 153 times. Currently, the class is designed for eight weeks. Rick has developed a 13-page course packet that class members are expected to complete.

Rick has incorporated his business experience into the class as a resource through various means. I focus here on the two most extended linkages. The first can be found on several pages in the course packet. To help demonstrate various lessons, Rick uses his favorite quotes from business speakers. Rick matches each quote with a Bible verse that he understands as stating the same principle. The following are only a few examples:

"The only reason people fail is Broken Focus!" – (name of business associate) *Philippians 3:20*

Respectively, these quotes lined up with teachings on the disciples' failing Jesus, the decision of salvation, and evangelism. Rick posits a relationship of semantic equivalence, where the quotes and the Biblical texts share a common meaning. The textual ideology work being accomplished here is discussed below.

The second example from the class involves the appropriation of a model for financial earning. The model is called "The Cash Flow Quadrant" and it comes from Robert Kiyoski's (1998) book of the same title. When presenting this lesson, Rick drew a

[&]quot;Success is just a decision." – (name of business associate) Ephesians 5:8-17

[&]quot;Anything worth doing well is worth doing poorly, until you can do it well." (name of business associate) *Philippians 4:9*

circular diagram on the board and divided into four sections. In each "quadrant" he wrote a principle that dictates how people earn money. The model is progressive, with the ideal of moving through the four quadrants as stages. After a brief explanation Rick erased the title and the principles, but not the diagram. He then proceeded to fill in the model again.

The revamped model is titled the "Christ Flow Quadrant." Rick replaces each of the financial principles with one of his four principles for "achieving the mind of Christ:" "receiving" the Gospel; "growing" in knowledge of the Bible; "discipling" other believers by leading a Christ-like life; and "equipping." This final principle is positioned as an ideal of faith, where you guide others through this same process of spiritual growth. Throughout his description, Rick wove in other discursive resources. To introduce the model he used Nightingale's "strangest secret" to describe how this evolution of Christian identity takes place. The lesson was also brimming with Biblical proof-texts, as well as doctrinal requirements of "immersion⁶⁴" and being "born-again⁶⁵." These examples illustrate Rick's intricate blending of the discourses of business and Evangelical Christianity. Rick's participation in the Monday Group demonstrates an alternative means of combining these two discourses.

The Monday Group used a unique method for Bible reading. Rick would ask participants to read a selected number of verses, usually between five and ten. Following each reading was a period of discussion, followed by the next set of texts. In the course of these discussions Rick drew on his business as an interpretive resource. Table 12 shows

⁶⁴ This refers to the baptismal practice of immersing someone completely underwater, as opposed to simply sprinkling him or her with water. Rick emphasized this as a Biblical necessity, as does RMC and the Restoration Movement.

⁶⁵ This refers to the spiritual change that takes place when someone professes a belief in Christ. It is based on the text of John 3 where this phrase appears as Jesus is speaking to the doubting Pharisee Nicodemus. For an extended discussion of "born-again" as a discursive construction see Bielo (2004).

how this resource compares to others invoked during my eight recorded meetings with the group⁶⁶:

Bible Footnote	Business	<u>History</u>	Visions/Dreams	Denomination
15	8	4	3	2

Table 12: Monday Group Interpretive Resources

The following interaction comes from my first recording of the Monday Group, and is a representative example of how Rick incorporated his business experience. It begins with Jonathan reading Hebrews 4:6-8:

Jonathan: It still remains that some will enter that rest, and those who formerly had the gospel preached to them did not go in, because of their obedience. Therefore God again set a certain day, calling it Today, when a long time later he spoke through David as was said before: "Today, if you hear his voice, do not harden your hearts." For if Joshua had given them rest, God would not have spoken later about another day.

Rick: Yeah, he capitalized "Today" there because it encompasses until the time he returns again.

Taylor: The beginning of the kingdom, basically.

Me: Sort of, "Today" is every day?

Rick: Yeah.

Taylor: The beginning of the kingdom.

Rick: Yeah. The beginning of the kingdom until he comes again. So, that really denotes his authority over all history, too, that he still calls this day today. He'll come tomorrow today if tomorrow comes.

Jonathan: Yesterday's today. Tomorrow's today.

Rick: Yeah. And, so, I heard an example this past weekend at a function. I thought about this scripture. Frank Feather is a leading futurist in the world who was talking about where business is going. But, this scripture just popped out for some reason. And, somebody asked Frank Feather why he decided to study the future. And, he said, "A professor of mine told me something that stuck in my head. He said the past is the past and there's nothing you can do about it. You blew it, so you can't manage the past. Today, the present, is only here for a nanosecond and then it's the past. So, you can't manage the present. The only thing you can manage is the future." So, once he learned that the future is the only thing you could manage, he focused on the future. And, every decision he made was based upon his future. And, as you look at this, it's like, we have a future. Today is here and half of it's gone, and what I just said is gone. And, so, the idea of "Today," is SO awesome because it, if tomorrow does come there are still people who can come into the kingdom. And, his grace is sufficient for those people to obey.

⁶⁶ Even though Bible footnotes were used most often, the participants' use of this resource was not very significant. Typically, footnotes were consulted briefly prior to or following the group's discussion of Biblical texts. They did not treat these notes as authoritative or absolute in any fashion.

The group homes in on one particular aspect of Hebrews 4:6-8, the reference to "Today." The group establishes quickly what the capitalization of "Today" means. But, to expound on the significance of the concept, Rick recontextualizes Frank Feather's statement on the future. Rick connects the verse to the immanence of Christ's Second Coming ("if tomorrow does come..."), the practice of evangelism ("there are still people who can come into the kingdom..."), and the expansiveness of God's grace ("his grace is sufficient for those people to obey"). Rick uses the resource of business to extract multiple forms of resonance and relevance from the Biblical text.

Rick's reliance on his business experience comes across somewhat differently in these two groups, but both are equally implicated in textual ideology work. Two particular linkages are most evident. First, the ideological principle that the Bible is as true today as when it was written runs throughout Rick's blending of discourses. This happens most clearly when he equates the meaning of Bible verses to quotes from contemporary business gurus. In doing this, he asserts both the authority of the speakers and the ongoing evidence of Biblical messages in the modern age. This also has an openended quality, as the list of quotes Rick includes in the course packet is always up for revision. Rick finds new examples of Biblical messages as he attends new seminars and listens to new speakers. The ability to continually find new examples re-iterates the timeless quality of scripture.

Secondly, the textual ideology work of is present in Rick's affinity for quotes.

During one Sunday school class, Rick was asked why he liked quotes so much. He responded:

"Because they are easy to remember. Just look at scripture. It's organized into chapters and verses. Translators did that. I believe God worked through history to do that. But, it's like that because that's how we remember things."

For Rick, quotes are a divine mandate. This notion is rooted in the textual ideology that the Bible is the authoritative Word of God. It begins with the actual writing of Biblical texts. It extends to everything that has happened to the Bible through history: canonization, translation, and even the division into chapter and verse. God's control over his book dates to the earliest writings and continues through "Today." And, for Rick, designing the lessons for the Mind of Christ occurs in step with textual ideology principles.

The Significance of the Interpretive Matrix

In this chapter I have sketched the composition of the interpretive matrix in small group discourse. I have outlined the three types of resources used within groups, and showcased a variety of specific resources that are invoked. I have also analyzed how these resources function in discourse, how resources impact the unfolding of discussions, and how the same resource can be employed for different purposes. In conclusion, I now turn to broader questions. What is the theoretical significance of the interpretive matrix? Why devote so much attention to this issue of resources?

The first point of significance is the theme that has extended across all of the resources discussed: their role in the ongoing work of textual ideology. As participants find ways to integrate their favored resources into discussions, they also enact their relationship with the Bible. Different cases demonstrated how the use of a given resource carried with it the imprint of one or more principles of the textual ideology. In some cases, like the Tuesday Men and the Iconoclasts, resources were invoked in a way that aligned with the group's primary interpretive orientation to the Bible. In other cases, like Rick's groups, the resource indexed the continuing goal of demonstrating the relevance

of the Bible. In short, the resources of the interpretive matrix are closely connected to the work of affirming ideologies and establishing the proper relationship with the Bible.

A focus on the interpretive matrix also highlights a mechanism that I have not emphasized in other chapters. It ensures that Bible study is not treated in some form of experiential vacuum. Earlier chapters emphasized what unfolds and what is cultivated during Bible study, whereas this chapter shifted the emphasis to what participants bring to Bible study. The lives of members outside of Bible study, and the incorporation of those lives into group discourse, are integral to this chapter. Thus, attention to the interpretive matrix provides a crucial perspective for the analysis of small group Bible study, recognizing the dialogical relation between group discourse and experiences outside of the group. This is what Roberts (2005) calls, "lived Christianity: the types of knowledge, beliefs, practices, experiences, and feelings that members of Christian congregations bring to their religious lives" (4).

Chapter 3 argued that small groups are interpretive communities. Every act of scripture reading in the small group setting relies on a common set of styles for interpreting the Bible, and a common textual ideology. This chapter's depiction of the interpretive matrix further supports my claim that small groups function as interpretive communities by delineating the types of resources participants rely on to participate in group discourse. This chapter also qualifies the unity and sharedness implicit in chapter three's presentation of interpretive communities. Indeed, the basis there was defined by a commonality of method and textual ideology. However, the interpretive matrix introduces a sense of variety and idiosyncrasy amidst this commonality. Interpretive communities, much like speech or linguistic communities, can be expected to have

important axes of diversity. Illustrating the range of resources used, and the differing strategies for employing them, ensures the notion of small groups as interpretive communities encompasses this variation.

Lastly, understanding the interpretive matrix of small groups allows for comparative studies. Any type of discourse event will rely on a set of interpretive resources through which its participants filter their contributions. This chapter has shown that small groups are a fine example of how complex any given matrix might be. Perhaps my analyses here will provoke others to ask what constellation of resources defines other discursive situations. How do participants weave these resources into conversation? Are the resources shared or idiosyncratic? Are there any larger structures of meaning or practice that are attached to the use of resources? This type of comparative questioning can lead to better understandings of interpretive matrix as an analytical tool, and the nature of group discourse as a social activity.

The central themes of this analysis – interpretive practice and textual ideology – are continued in the next chapter. At various points in my discussion of the interpretive matrix I identified instances where participants "recontextualized" words of speakers and units of text. Hanks also recognized the intrinsic link between interpretive matrices and the recontextualization of text (1989: 107). In chapter 5 I argue that the central forms of recontextualization in small group discourse – recontextualizations of Biblical texts – are also implicated in the work of Bible interpretation and textual ideology.

Chapter 5: Recontextualizing the Bible in Small Group Discourse

"All of us have become like one who is unclean, and all our righteous acts are like filthy rags; we all shrivel up like a leaf, and like the wind our sins sweep us away."

(Isaiah 64:6, NIV)

Charlie: Jesus talked about personal responsibility, and that is SO Christian. But, there's also a set of weightier matters: justice, love, seeing people as God sees them. We work so hard to do the opposite.

Christine: Who are we to say somebody is going to hell?

Charlie: Well, you know, I use to think I knew. I use to think I knew exactly what the score was. I've discovered in these latter years that that was self-righteousness, and it was wrong. I've discovered that, like the prophet Isaiah said, my righteousness is as filthy rags. At this point in my life, I'm feeling like I'm lucky just to get in the door. If the castle door's going to be closing, I'm going to be sliding underneath. I'll settle for the cheap seats, the nosebleeds, as long as I get in. I don't necessarily need to live in the mansion. I can live in a pup tent. I'll be fine.

Vincent: If I choose to do something in my own volition, from my head I guess, to make it more understandable, it's good. But, it may not be what God wants me to do. If my heart's not engaged, my spirit's not engaged, and if He's not talking to me, or I'm not listening to Him, then, I'm going "I think this ought to be done." In a sense, I'm playing God. And, so, here's a little god. There's a little god. And, it says, your own works are rags to God.

Daniel: The ninth and tenth commandments are not disconnected with some of the 40 Days of Community stuff, in terms of the accent of not looking after our own affairs only, but also those of others. Maybe take a quick look at Philippians 2, where Paul says: If you have any encouragement from being united with Christ, if any comfort from his love, if any fellowship with the Spirit, if any tenderness and compassion. As we think about living a better life, which is what the 10 Commandments would be speaking to us about, that always only grows out from his love for us. In and of ourselves, all of our righteousness is like filthy rags, but in Christ and through Christ and because of Christ our hearts are changed. Our lives are made strong and we're better able to live this out.

In the three excerpts above, three speakers from three different groups used the same Biblical text to aid their respective explanations. In this chapter I examine this practice of invoking Biblical texts. In particular, this chapter considers the interpretive, textual ideological, and interactive implications of incorporating the Bible into small group discourse. To frame this analysis I begin with a theoretical and methodological introduction.

Recontextualization as a Discourse Strategy

The practice of invoking Biblical texts exemplifies what Bauman and Briggs call decontextualization and recontextualization: "the process of rendering discourse extractable, of making a stretch of linguistic production into a unit – a text – that can be lifted out of its interactional setting" (1990: 73). In the examples that opened this chapter, my righteousness is as filthy rags was extracted from its original contexts of production – a verse, a chapter, and a book in the Bible – and placed into a new discursive setting – group discourse. At first glance, the concept of recontextualization may seem like academic hubris for what is simply a particular form of citation or quotation. However, that claim misunderstands this intertextual process. Acts of recontextualization ultimately provide a meta-commentary on the original context, the extracted textual unit, and the new context. The text is treated as a discrete, objectifiable unit. Its potential for circulation and applicability in new settings is recognized. And, its semantic range and properties are positioned for change (Bauman 2004).

Recontextualization is a concept that was originally developed for the analysis of spoken discourse, particularly performative contexts. But, scholars have also recognized the utility of extending its application across semiotic mediums (cf., Davie 1995; Silverstein and Urban 1996). In the present case, written texts from the Bible are recontextualized as spoken texts in group discourse. Recontextualization is a particularly potent strategy in religious discourse generally because of its transcendent and fluid character (Keane 2004). Others have observed the practice of Biblical recontextualization outside the small group setting. Anna Meigs (1995) argued that the everyday language of conservative Christians was imbued with scriptural references, embedding notions of

obedience and temporal relations between the Biblical past and the present. Robin Shoaps (2002) argued that the verbal genres of prayer and praise songs in Pentecostal church worship draw on Biblical texts in a way that highlights spontaneity and speaker creativity, both valued ideals in the charismatic tradition⁶⁷.

In this chapter I emphasize the significant impact of Biblical recontextualization in the interpretive and interactive qualities of group discourse. I argue that Biblical recontextualization is patterned by the interpretive orientation within the group. I also argue that the style and pattern of Biblical recontextualization indexes principles of the textual ideology, larger discourses operative in American Protestantism, and key interactive norms of the small group setting. I organize these claims in relation to two questions. What does Bible study discourse tell us about Biblical recontextualization? And, what does Biblical recontextualization tell us about Bible study discourse?

To answer these questions, I need to address three issues. The first is methodological in nature: why did I choose the practices of particular speakers as the subjects of my analysis of recontextualization? The second issue is analytical: toward what functional and interpretive ends did participants recontextualize the Bible? The third issue is theoretical: are there any relevant formal properties of recontextualization?

Sampling the Group Participants

Participants from seven of the eight Bible study groups are represented in this chapter⁶⁸. I draw from the recontextualization work of 14 participants. Each was chosen on the basis of two criteria: the number of Biblical recontextualizations made and the

⁶⁷ Cf., Crapanzano (2000), Csordas (1997), Engelke (2004), Harding (2000), Malley (2004), Muse (2005), and Stromberg (1993) for related discussions of how Christians employ Biblical texts.

⁶⁸ The Mind of Christ was not included because multiple meetings were not tape-recorded. However, Rick, who is represented by his participation in the Monday Group, would have been the only participant to be included in the analysis.

individual's role in group discourse. For groups like the Iconoclasts and the Tuesday Men only one the facilitator recontextualized the Bible enough for me to perform an adequate analysis. In the remaining five groups, multiple participants are included. This allows the variety of roles occupied in group discourse to be represented. For example, with the LC-MS women, Joy is not the facilitator, but she was among the most active members, and her contributions were highly valued within the group.

Any form of reference to the Bible was included as an instance of recontextualization. This includes direct quotation, paraphrasing, and invoking the name of a Biblical character, story or event (see below for a discussion of these distinctions).

Table 13 identifies the participants I chose from each group, the number of meetings they attended, and their total number of Biblical recontextualizations.

Group	Speaker(s)	Meetin	Recontext.	Avg. Recontext. Per Meeting
Iconoclasts	Charlie	16	120	7.5
McGee Men	Joel, Roland,	12	25, 19, 18	2, 1.6, 1.5
Tuesday Men	Ted	23	79	3.4
LC-MS	Sandy, Joy	26	24, 25	1, 1
LC-MS Men	Daniel, Aaron, Ned	18	73, 57, 20	4, 3.1, 1
Prayer Circle	Darren, Jennifer	14	30, 24	2, 1.7
Monday	Rick, Taylor	8	42, 18	5.2, 2.2

Table 13: Recontextualizations for 14 Bible study participants

Functional and Interpretive Qualities of Recontextualization

Each instance of Biblical recontextualization was analyzed and coded for its usage in group discourse. I developed six codes, which can be divided into three mutually exclusive categories. The table below presents these codes and a brief definition of their function.

Recontext. Style	<u>Definition</u>		
Interactive Support	Bible text used to support the contributions of other participants.		
Interactive Challenge	Bible text used to challenge the contributions of other participants.		
Text Linking	Bible text posited as connected to another Bible text.		

Text Interpretation	Bible text posited as informative for interpreting another Bible text.
Critique	Bible text used as the basis for critiquing individuals and groups.
Parallel	Bible text used to establish similarity with individuals or groups.

Table 14: Definitions for recontextualization styles

Recontextualization and Interactive Function

Group members recontextualized the Bible toward two interactive ends: supporting and challenging other group participants. Given the textual ideology principles of authority, this can be understood as the ultimate form of agreement and disagreement. Joy from the LC-MS women offers representative cases of each. The example of support comes from a contrast that had been drawn between the LCMS tradition and "megachurches." The group firmly agreed that even though they boast impressive memberships, megachurches are too lax on their members:

Sandy: The traditional churches, in their 1940s, 50s style, have lost the ability to attract outside people. Clearly they're shrinking. Our church body is shrinking.

Joy: So, they're in the pews. But, what are they hearing, fluff that isn't doing them any good?

Katherine: That's what the pastor started preaching, because that's what they wanted to hear. And, that's what kept bringing them in, kept bringing the dollars in so they could put it into building the statues and the fountains and everything else.

[[Multiple women agree]]

Joy: Yeah, Jesus did not make it palatable. When the rich, young man wanted to know what he could do to be saved, Jesus didn't say, "Well, come and follow me for a while, just kind of listen in." He said, sell all you have and give it to the poor. He set it right out, this is what it is and the guy couldn't do it and he turned and left and Jesus didn't follow after him and try to make it sound appealing. And, I think there's too much of trying to make it sound appealing and entertaining. It goes on it a lot of churches. They're there, but what good is it?

In this example, Joy uses the story of Jesus and "the rich, young man" to support what Sandy, Katherine and others were saying about the growing trend of megachurches. In contrast, Biblical texts can be recontextualized to challenge another participant's contribution. During one of my final recordings, the topic of tithing arose in response to a study guide question:

Sandy: What dangers await Christians who submit to legalistic rules and regulations?

Elinor: Well, even saying you have to tithe can become legalistic.

Sandy: Uh huh. Yeah.

Jane: Yeah. A lot of churches do that.

Sandy: Do we do that?

Sherry: They suggest that that's what God's wants you to do because that's what it says in the Bible. But, you know, you're not forced to. I don't think anybody's pushed out of the church if they don't do it. It's just that, I think that what the church does is try to show you what actually God's blessings are. And, when you do tithe what you're doing it for and why and how it helps you in the long run and what God says he will do for you if you do that. But I don't think that if you don't, I don't know. It's really, actually, a rule that He's made. Don't you think? That's what it is in the Bible. That's what he's asked us to do. And, if we can't do it, if we don't have the money to do it that's one thing.

Sandy: You can't take ten percent of what you don't have.

Frita: It doesn't say in the New Testament to tithe. It says to be a cheerful giver. In the Old Testament they did tithe, but in the New Testament it doesn't SAY, because we had a pastor that used to really, "It doesn't say anything about tithing." He said, "You should, because tithing means, if you have a lot of money you shouldn't just be giving a tenth, you should be giving more." You're supposed to give from the heart.

Joy: But, Jesus tells the Pharisees, I don't want you to just tithe, I want you to bring your offerings. They were making a big to do about tithing their cumin and whatever their spices were.

In this example, Joy uses Jesus' words to the Pharisees as a means of arguing against Frita's stance toward tithing. These two styles of Biblical recontextualization occurred in all seven groups. The strategy of support was far more common for each of the 14 participants. In the final section of this chapter, I consider the larger resonance of this trend.

Recontextualization as Meta-Commentary

The second style of recontextualization encompasses the use of scriptural texts to provide a meta-commentary on the Bible. I divide this style into two distinct types: (1) linking scriptures together and (2) interpreting another Biblical text. To demonstrate these I use Rick's participation in the Monday Group.

They had just read Hebrews 8:10-13. They puzzled for a moment over whether the time of the *new covenant* described in these three verses has arrived.

Janice: So, has this time come yet?

Rick: Uh, yeah. Because he has just got done talking about what Jesus' ministry is. So, people who, what's that say...[[trails off]]

{9.0}

Janice: Just that not everybody *knows Him from the least to the greatest*.

Rick: Right. Not everybody knows him.

Janice: So, how can it be now?

Rick: You're thinking it's some time in the future when Christ comes back and heaven is

a reality?

Janice: I'm not really thinking, just making an observation.

[[Laughs]]

Janice: Like, you mentioned that girl [earlier]. She didn't know anything about the Bible

or God and she's right here in the midst of all of us.

Taylor: I wonder what that Hebrew word translates, though, as "know"? Knowledge of?

Rick: Yeah.

Taylor: You know, there's a lot of real, very relational definitions of that word.

Janice: Mhmm. But, we're still supposed to be teaching our neighbors.

Taylor: Oh, exactly.

Janice: And, it says, no longer will a man teach his neighbor.

Taylor: That's right.

Rick: That sounds like the, yeah, the total promise right there. But, I think it's all intertwined together. It's like, Romans 11:27 says, and this is my covenant with them when I take away their sins.

The use of recontextualization to link Biblical texts together was the most consistent strategy across the sample. It was the most common in four groups (LCMS Men and Women, Prayer Circle, Monday Group) and the second most common in two others (Iconoclasts and McGee Men). When participants engaged in this strategy they posited that two Biblical texts share semantic, rhetorical, and/or functional properties. Linking embeds the assumption that the linked Biblical texts mean or imply the same thing and/or are applicable to the same issue. In the above example, Rick uses this strategy to make sense of the confusion surrounding Hebrews 8:10-13 by paralleling it to Romans 11:27.

In a related fashion, Biblical texts were used to interpret other Biblical texts.

During the following week's meeting, Rick asks the group to reconcile a seeming tension between Hebrews 10:19 (Therefore, brothers, since we have confidence to enter the Most Holy Place by the blood of Jesus) and Philippians 2:12 (Therefore, my dear friends, as you have always obeyed – not only in my presence, but now much more in my absence – continue to work out your salvation with fear and trembling). The tension is between

salvation attitudes of *confidence* and *fear and trembling*. Rick responds to his rhetorical question:

"What God wants is our obedience. Yeah, there is a fear of God. But, it's a righteous fear, a Godly fear of who He is. Says in, what we'll read later on, it says our God is a consuming fire. And, it also says it is a fearful thing to fall into the hands of the living God. You know. So, but, we can have confidence as well. How do we do that? Well, how do you combine those two? Well, God wants a relationship with us. That's the whole context of this, really, is He wants a relationship with His people. Many people get the idea that He's a guy up there with a white beard ready to pound people down to the ground. Jesus said, if you have seen me, you have seen the father. How can we expect God to react to what we do when we sin? I would look at Jesus. How did he treat sin? He didn't accept excuses. He loved the people, but he said, like the woman caught in adultery. He defended her. Then, he said, go, don't sin anymore. Don't get caught in a place where this could happen. Just don't even get near it. Quite often, we find ourselves in that same situation. What do we do? Give in to it. But, God even has grace over that. My little children if any of you do sin, John goes on to say."

In this example, Rick uses the words of Jesus and John to resolve the seeming tension between the texts from Hebrews and Philippians. The use of Biblical recontextualization as an interpretive tool embeds the commentary that scriptural texts can (and should) be used to interpret other scriptures. When one text is unclear, the best way to clarify it is to consult another Biblical passage. This style of recontextualization was less common than linking, but still had an active presence in most groups. The only exception was the Iconoclasts, who did not read directly from a Biblical text. The styles of linking and interpretation are connected in that they both operate internal to the Bible, providing commentary on the nature of the Bible.

Recontextualization as Application

The third style of recontextualizing the Bible encompasses two types of text-external strategies, where the premise is that the Bible can be applied to contemporary life. The first strategy uses the Bible as a tool for critique. The object of this critique varied from the modern Church in America to the life of the speaker to non-Christians. The second strategy uses the Bible as a source of commonality. Parallels were drawn

between Biblical characters and events and contemporary personages and events. The Iconoclasts facilitator, Charlie, provides examples of these strategies, all of which occurred during their first meeting for <u>The Jesus I Never Knew</u>.

Charlie employed the strategy of critique far more than he did any of the other recontextualization styles. This partially reflects the text that was the focus of the discussion, The Jesus I Never Knew. The book thoroughly critiques how modern believers conceptualize Jesus. However, Charlie's participation in group discourse was more inclusive. He used a Biblical text to critique an earlier phase of his own spiritual development:

"Baptist Christianity fit me to a tee – performance orientation, hypocrisy. I was a hypocrite and nobody had to know about it. I could out perform anybody in terms of the vocabulary and the outward lifestyle. It took me a long time to realize that God is a judge of a man's inward heart, and that my righteousness is as filthy rags."

Charlie placed contemporary American Christianity, the object of the book's critique, at odds with scripture:

"We've taken Jesus and we've compartmentalized and put him in a spot that we're comfortable with. And, when you look at what Jesus says about himself, and the things he does in the Gospels, many times in the church we do exactly the reverse. Jesus said, Judge not less you be judged. And, we judge people EVERY single day that go wandering into our lives and into the church. That's why I was just so blown away when I read the Gospels this last time around, because the reverse culture, the upside down world that Jesus talks about we don't do. We put together our little systems, our orders, and we go about our business."

The "we" in this example could be construed as a localized critique, perhaps of the group or the local congregation. But, it is clear from Charlie's other remarks that "we" includes the broader category of American Christians. Lastly, Charlie distances contemporary American society from scripture:

"The thing that it's all about, at the end of the day, is what's presented in the Gospels. The Gospels are so revolutionary compared to how we live in our society. Give you an example: the last shall be first and the first shall be last. That's out of the Beatitudes. In our society, the first shall be first. Those who die with the most toys win. In God's

economy, those who have died with the most toys have a lot of toys on earth, but you ain't got nothing when you get to heaven."

In these examples, Charlie uses the Bible as the basis for criticizing various targets: himself, American Christians, and American society. This style of recontextulization contrasts deficient beliefs or practices to Biblical authority, timelessness, and ideals. It implicitly asserts a separation between what should be and what is happening.

Biblical texts were also recontextualized to suggest a parallel between the Bible and modern life. The Iconoclasts' second meeting opened with a question from Charlie: "What was your first image of Jesus?" Several participants affirmed that they could relate to the image described by the book's author. Charlie then related a strategy he uses for reading the Bible:

"I always try to analogize these situations into modern equivalents so we can understand them. I thought of the story when Jesus, when he threw the moneychangers out of the temple. You read that he threw the moneychangers out of the temple because they were selling {one word unclear}. And, I thought, alright, they were making profits. I thought what would be a modern example of that? I thought, I got it. Jesus comes in and he goes into {our church} and he takes the cash register out of the bookstore and pitches it into the parking lot. I thought that wouldn't be a bad, kind of working analogy."

In this example, Charlie draws the analogy between the story of Jesus and the moneychangers and the bookstore that operates out of the group's local church. This style of recontextualization sets up parallels between Bible events and personages and their contemporary equivalents. The implication is that modern life operates in much the same way as the world of the Bible. The example above is used as a form of critique, but this style was also employed as a demonstration of how individuals and groups were meeting the standard set forth by the Bible.

Formal Properties of Biblical Recontextualization

This chapter highlights forms of recontextualization that are not usually emphasized by other scholars in the field of discourse analysis. The latter group typically focus on the formal and stylistic features of discourse. For example, in her analysis of recontextualization in prayer and praise songs, Shoaps (2002) pays particular attention to marked prosody, meta-pragmatic devices, evidentiality, deictics, locutives, and specialized registers. This emphasis is consistent with the pioneering "discourse-centered approach" articulated by Sherzer (1987) and Urban (1991). This work proceeds from the notion that meaning is not only (or, in many cases, primarily) constructed referentially. Communicative events also embed meaning through stylistic, structural, and other formal properties of discourse. That rich stream of research suggests a valid question for this study: are there any significant formal properties of Biblical recontextualization?

Earlier I noted that I included all forms of reference as instances of recontextualization, clearly combining types of reference that differ in formal terms. For example, direct quotation accesses the original context of the Bible differently than simply making mention of a Biblical character, story, or event. We can distinguish this latter type of recontextualization as a form of "metonymy," a metaphorical device in which a part is enabled to stand for a whole, and vice versa (Lakoff and Johnson 1980). In their original discussion of recontextualization, Bauman and Briggs highlight the practice of metonymic substitution as a potentially important process (1990:77). They cite Keith Basso's (1990) work among the Western Apache as an exemplary account of how a single textual unit – a place name – can be imbued with rich, powerful, and cumulative cultural meaning (e.g., moral narratives). Metonymic substitution was a less

common style of recontextualization, but it did occur with some frequency. Table 15 lists the percentage of each speaker's recontextualization work that occurred as metonymy.

Speaker	% Metonymy	Total #	
Charlie	20	13	
Joel	12	3	
Roland	5	1	
Vincent	38	5	
Ted	20	16	
Sandy	46	11	
Joy	8	2	
Aaron	14	17	
Daniel	30	10	
Ned	20	4	
Darren	17	5	
Jennifer	-	-	
Rick	12	5	
Taylor	6	1	

Table 15: Metonymic recontextualizations for 14 Bible study participants

Metonymic reference is striking because of the dialogical construction of meaning underlying the ability to forego more direct quotation. Certain texts can communicate entire narratives and lessons using a single word or phrase. Participants were apt to invoke a particular range of Biblical texts metonymically. In particular, Old Testament character stories, the parables of Jesus, and Gospel accounts of Jesus' interactions were approached in this way, as the following examples demonstrate.

The following exchange is from the Tuesday Men. It took place at the opening of their discussion of Acts 4:23-27, and begins with the reading of the first study guide question: "What does their prayer tell us about their faith in the character, power and faithfulness of God?"

Eddie: I wrote down persecution, opposition, trouble can be seen as from God's hand and plan...

To expect trouble. But, in the midst of this, God provides boldness and healing and signs and wonders. One of the amazing things is that we're saved out of trouble. In fact, we get in trouble. And, this, I think, is a hard thing to face. As long as we don't have it too hard to think that our Christianity's getting us in trouble.

Ted: Well, and I think even that we tend to think of that in the opposite way. Much like the Jewish theological thought in *Job's* day. The thought was that Job must have done something wrong, or else he wouldn't have the trouble.

The story of Job is among the most common Old Testament texts that were recontextualized metonymically in small group discourse. The book of Job, some 42 chapters long, has been distilled to represent an ideal: faithfulness in the face of hardship. The social circulation of this text and the accompanying distillation of meaning have made it amenable to metonymic substitution. Indeed, simply saying the name "Job," as Ted does above, inserts this particular script – faithfulness in the face of hardship – into group discourse. The same process occurs with the genre of the parable.

As the Iconoclasts neared the end of the night's meeting, their reading from The

Jesus I Never Knew included this statement on the concept of grace: "In God's economy,
there are no undesirables." Charlie picks up this thread to begin the discussion:

Charlie: In Jesus, in God's Economy, there are no undesirables. If you have a heart for the Lord, and you are willing to say in your mind and in your spirit, "Dear God, I'm a sinner. I've got no hope, except for you," you're in. You've made it to the varsity. Not only did you make it to the varsity, you get to start.

Diane: Right, which makes going back to that original question all the harder. How is it our churches turned into these places where you have to have certain expectations in order to go through the door?

Lauren: Because that's not fair. God's way's not fair.

Jill: Yeah. You think of the murderer who's been committing all those, the last thing he does...[[Overlaps with LB]]

Lauren: "It's not right. How can you do all this rotten thing, and right before, you repent and you do it. It's just not right." We want fairness.

Charlie: I'm reminded of that Parable of the Laborer.

Lauren: Of the laborer, right.

Christine: Exactly.

Monday Group provides a similar example. As they neared the end of John 1, Rick responds to a footnote he read from his Bible:

"I would agree with the former part of this, saying that he's speaking directly against Gnostic teaching and that people who are Christians and go into that type of teaching and leave the Word, they're in REAL danger, spiritual danger. That's what I learned, from many different people, and it makes sense. It's like a choice of severing the relationship.

You have a relationship with God and now you go back into those things, which God has saved you from. I believe a person could come back from that if they repented. I mean, we have the example of *The Prodigal Son*. But, if a person continues to just deny the power of God, and chooses to live his life that way, I think they're in real danger."

In both examples, parables are invoked iconically, representing particular claims: "fairness" and 'coming back to God.' The efficacy of the metonymic property is evidenced in Charlie's use of the *Parable of the Laborer*, as two other participants were quick to affirm his reference.

Metonymy, then, occupies a particular position within the practice of recontextualization. The need for direct quotation is alleviated. The lives of characters and the unfolding of narratives are distilled for a core meaning that can be efficiently deployed with the substitution of a name. Through their circulation in small groups, other congregational settings, and other institutions this meaning has been sedimented in their memory as the most appropriate reading and interpretation.

The foregoing accounts for a number of theoretical and methodological issues involved in the practice of recontextualizing the Bible and sets the stage for the main concerns of this chapter. I begin with an analysis of how the interpretive environment of small group discourse impacts the style of recontextualization.

What does Bible study discourse tell us about Biblical recontextualization?

In this section I argue that the style of recontextualization employed by participants indexes the interpretive orientation of the group. In some groups, the interpretive orientation is shared, and the recontextualization style is consistent among all participants. In other groups, the interpretive orientation is contested, and multiple recontextualization styles operate.

This dynamic between interpretive orientation and recontextualization style has been observed in other institutional settings. James Collins (1996) has argued that in primary school classrooms the recontextualization of lesson texts aligns with one of two orientations: pragmatic or semantic. The former is concerned with pronunciation and the latter with establishing meaning. Elizabeth Mertz (1996) argued that in the discourse of law school classrooms legal texts are recontextualized in a manner consistent with the dominant ideology toward these texts cultivated through socialization into law school. Richard Bauman (1996), in his work with Mexican nativity plays, argued that differing relationships to the play script led to differing levels of faithfulness during recontextualizations. In each of these cases, the relation between interpretive orientation and recontextualization style is upheld.

Three case studies demonstrate this argument in the context of small group Bible study: the Tuesday Men, the LCMS men, and the McGee Men. In the Tuesday Men and LC-MS Men the group facilitators' contrasting recontextualization styles are indicative of the entire group. The McGee Men offer an example of multiple recontextualization styles due to competing interpretive orientations.

The Tuesday Men were described at length in chapter 1. They met weekly to read and discuss the New Testament book of Acts. The 23 sessions I recorded cover the first eight chapters. They used the LifeGuide series as a study guide to structure their discussions (see Appendix 3). The size of the group ranged from four to eight participants, with the same six men present on a weekly basis. The facilitator, Ted, is the church pastor and accounted for the vast majority of Biblical recontextualizations.

The interpretive orientation of the Tuesday Men was distinct within the project sample. The following interaction occurred during the group's reading of Acts 5:1-11, in which Ananais and Sapphira lie to the apostles about how much money the former received for selling their land. The study guide question asks, "What have Ananais and Sappharia done that is not consistent with the Christian community?" The men easily identify the sin as lying. This leads Ted to pose his own question: "I think a big question that people have that just read this is why did they die?"

This question returns us to the interpretive style of explanation and the process of textual inferencing. In this case, Acts 5 does not explicitly announce the reason for, or cause of, Ananais and Sapphira's death. Ted's question requires the men to infer from their backdround knowledge the reason for the couple's death (cf., Fairclough 1989). The most frequent explanation inferred from the text is asserted by George, who cites a footnote from his study Bible:

"I was thinking they'd want to nip it in the bud. I'll read you what this says. 'This act was judged harshly because dishonesty, greed, and covetousness are destructive in a church, preventing the Holy Spirit from working effectively. All lying is bad, but when we lie to try to deceive God and his people about our relationship with him we destroy our testimony for Christ."

This inference, that their death was the result of God's judgment, draws on a theological form of background knowledge emphasizing God's judicial character. Ted immediately offers an alternative inference:

"I don't think God zapped them. I think the Church was so committed to this kind of lifestyle that when they realized what they'd done, I mean, there's such a sensibility. There's that great line in the hymn 'Take Time to Be Holy.' And, there's a number of things that are asked for by the writer of that hymn, but one of them is a sensibility to sin. A sensing, an ability to sense and to be clear when sin comes. And, I think that was so much a part of this community. And, when Ananais and Sappharia realized what they'd done it just took them out. And, that's pretty hard. I think one of the reasons why that's hard for us to believe is because we tend to be so far from it. I think we're a long ways from that."

Ted's inference, that their death was the result of overwhelming guilt, draws on background knowledge: the ideal of the first century Church and the contemporary church's distance from that ideal state.

The group instantly, and positively, responded to this inference, which indexes the interpretive orientation of the group described in chapter 3. They read Acts as a running comparison between the Church of Acts and the modern American Church. They inevitably concluded that the current Church is at odds with or deficiently represents the standard set by the Church of Acts. This mode of approaching the Bible is not inevitable. Like any text it is polysemous, and can be engaged from multiple interpretive orientations. Considering the abundance of potential proof-texts for charismatic expressions in the first eight chapters of Acts, a likely alternative might be a running discussion on the availability of the gifts of tongues and healing for modern believers.

Given this interpretive orientation, what should be expected from Ted's recontextualization style? Consider the following three examples. The first comes from a discussion of how the group understands the role of the Holy Spirit in their lives. After a lengthy pause, Ted interrupts the silence.

"I like David Siemens, who was our pastor when we were at seminary. He's a very practical guy. And, I like practical theologians. And, he said, "I get up every morning and I pray, Holy Spirit, lead me today. And, then, I trust that the Holy Spirit's leading me, until the Spirit comes and checks me and either stops me or leads me in a different direction." And, I like that. I think that's surprising. I think we waste a lot of time. I think folks spend a whole lot of time waiting for signs and wonders and miracles, when what they mostly need to do is say, "Holy Spirit lead me today," and then go in the direction that makes sense and trust that that's the way it is. When we get later on here in the book of Acts you see Paul saying, the Holy Spirit stopped us, kept us from going into Macedonia. We don't know how that happened. But, I suspect, it may have been something very practical."

Second, during a discussion of the relational concept of accountability, Ted comments:

"Well, the practice of confession and going to another person who represents God, I think we misinterpret Luther sometimes when we talk about the priesthood of all believers,

saying that means I can just go to God myself. Luther's point was that anybody can be a priest to another, wasn't that we can just do it on our own. I think in our zeal to move away from what may have been, what was some negative things around the hierarchy of the Church, we have lost some of the importance of confession and accountability and some of the things that are gifts. I mean, scripture talks to us about the gifts of confessing to one another, and really being in that level. It's a whole other level of relationship."

The final example comes from the group's reading of Stephen's speech in Acts 7, where he defends his proclamation of Jesus as Christ to the Jewish Sanhedrin.

"I think Stephen is a guy filled with grace and love, filled with the Holy Spirit. He didn't want to just beat them. I think sometimes we just want to win. And, if you just want to win then I think it's anger and those kinds of things that come out. If I just want to be right and point out where you're wrong, that's a whole different attitude. I mean, it's what Paul talks about when he talks about speaking the truth in love. Stephen is speaking the truth."

These three recontextualizations coalesce around a single interpretive feature.

Ted's recontextualization style posits a disconnect between the actions and motivations of Biblical characters and that of modern believers. Paul's words in regard to the Holy Spirit, the scriptural call to confession, and Paul's directive about speaking the truth are all drawn into the group's discourse in ways that imply dissonance. Ted's recontextualizations work to support the group's interpretive orientation, to demonstrate where the modern Church falls short of the example provided by Acts. This relationship between textual orientation and recontextualization style is evident in the other project groups as well. A contrasting case comes from the LCMS Men.

The LC-MS Men met weekly at a local restaurant. They spent 12 meetings studying the Old Testament book of Proverbs. The group did not use a study guide, preferring to discuss "what verses struck them" from an assigned number of chapters.

The size of the group was considerably larger and more variable than the Tuesday Men, with between 10 and 27 participants from week-to-week. The group typically had 18 and

25 men present. The facilitator, Daniel, was the church pastor and the most apt to recontextualize the Bible in group discourse.

As with the Tuesday Men, the group had a distinctive interpretive orientation. Consider the following example. During their discussion of Proverbs 31, for example, one of the participants inquired about verses six and seven: Give beer to those who are perishing, wine to those who are in anguish; let them drink and forget their poverty and remember their misery no more:

. . .

Daniel: Here's what I've looked at with this. Will I spend my days in drunkenness when I have been made rich by the Kingdom of God? I'm not sure it's really arguing in favor of keeping poor people drunk. I don't think that's really the intention. I think the accent's on verse 4: It's not for kings, O Lemuel – not for kings to drink wine, nor for rulers to crave beer. I think it's a statement about, if you've got important stuff you need to keep your wits about you. Let people that don't have important stuff, let them get hammered all day; but, not if you're going to be a king. You need your wits about you. And, to think about our own lives as people of God. We are not the poor. We are not the impoverished. And, it's good for us to keep our wits about us.

. . .

Aaron: I think it is tongue-in-cheek. It's almost backwards. It's almost, doesn't really mean what it says. It means just the opposite. It's not good to be drunk. So, the fools that are drunk probably are poor. Those that are bitter at heart probably got that way because they've been drinking too much wine.

. . .

Daniel: What do you make of this? Is this prescriptive? Are we supposed to go out and encourage poor people to get hammered?

Ned: Well, isn't four like what we talked about earlier, all the stuff about riches aren't necessarily millions of dollars, but *more than conquerors through the Gospel* type of thing? How come it changes when it talks about alcohol? For those who are poor, maybe a couple of drinks will loosen them up.

Andy: Well, it says give beer to those who are perishing. Does that mean dying?

Man: Or, does that mean going to hell?

Andy: Or, does that mean going to hell? Right. Which one does it mean? If it means they're dying and they're miserable...[[interrupted]]

Daniel: Is it just early form of hospice?

Andy: Right. Is it like a drug so you don't have to suffer? You won't realize the pain you have or the suffering you're going through, because it will keep you in a comatose state. They didn't have morphine back then, but they had beer and wine. {2.0}

Aaron: Because of what it says just prior to that, like you mentioned, in positions of authority that don't drink, keep a clear head. Then, what follows is kind of satire. Who gets drunk? The fool. Those that are bitter. I think it's kind of backwards, like you said.

Daniel: Saying, if you're gonna be a leader you need your wits about you. Let people that aren't going anywhere in life, if they want to waste their life in poverty and drinking. Andy: Or, is there a danger saying that's satirical, and really don't take its true meaning. Then, other people who go to other passages of the Bible and say the same thing. So, now we're picking it apart. So, you'll interpret it the way you want it and I'll interpret it the way I want it to fit my needs and my beliefs. There's a danger there. Daniel: We make a statement: I think I talked a little bit about this. Understanding the Bible literally and understanding the Bible literalistically. We want to understand the Bible literally, which means we take the pieces and parts as they're intended to be. That, an historical passage we understand as history. A poetical thing we understand as poetry. A figure of speech we understand to be a figure of speech. A, what's that called, a hyperbole. We say, "I could eat a thousand horses." Well, we aren't asking for a thousand horses. And, the trick is, though, to figure out what kind of literature is this? Is this a satirical statement? Is there a poetical statement? Is this a command? Literalistically would be to take everything and just say, "oh, yeah. Let's get a Budweiser truck and go start passing it out." But, what we might do is come in to a question about: well, what kind of literature is this?

This example characterizes the interpretive orientation of the group quite well.

Throughout their study of Proverbs, the LC-MS men wrestled with interpretive dilemmas. Here the dilemma is assigning the text to the proper genre. What is striking is their inclination to wrestle with questions of meaning internal to the text. This contrasts sharply with the text-external focus of the Tuesday Men.

One of the recurring interpretive dilemmas the group faced was the seeming contradiction between promises from Proverbs and other examples, from the Bible and from the everyday lives of participants. The question arose on several occasions whether or not a text should be considered as a promise from God at all. Their first encounter with this came in their reading of Proverbs 3 and 4. Pierce, a non-LCMS member but the son of a lifelong member, challenged the group:

"As your non-denom brother, I feel obligated that we don't skip over seven through ten, which reads like this: Do not be wise in your own eyes; fear the Lord and shun evil. This will bring health to your body and nourishment to your bones. Honor the Lord with your wealth, with the firstfruits of all your crops; then your barns will be filled to overflowing, and your vats will brim over with new wine. And, you know, the Word became flesh and dwelt among us. Jesus became flesh as the Word, and the Word we believe in has right there healing and prosperity. It's in there like Prego. I mean, it's in there."

Daniel's deftly dulls Pierce's edgy delivery: "Yeah. That's one of the issues that you have to deal with Proverbs. How much of this is promise? How much of this is just generally the case?" Several participants then traded examples from the Old Testament that supported and challenged the reading of the verse as a promise.

Pierce's denominational and theological distance from the group fueled this particular interaction. But, the pursuit of this ongoing interpretive dilemma also came from Daniel and other longtime LC-MS members. The following example comes from their study of Proverbs 20, 21, and 22. Sam, an early adult convert from the Baptist tradition, now in his early 50s, posed a question: "Twenty-two, verse six, *Train up a child in the way he should go, and when he is old he will not turn from it.* I guess my question is, how much of that can you hang on to?" After competing examples were offered, Daniel responded:

"I think this is one of those words, like I suppose all the words, that we need to understand in terms of letting Scripture interpret Scripture. You always have to see this in the context of everything else that's said. For instance, train up a child in the way he should go and when he's old he won't depart. 'Oh, my kid turned out to be this wayward ner-do-well, good-for-nothing. Maybe I failed. It's my fault.' Well, but then we balance that out with another verse that says, the soul that sins shall die. There's a personal culpability, and so we balance it out."

As with the Tuesday Men, the interpretive orientation of the group's discourse is mirrored in the facilitator's recontextualization style. Consider the following three examples. The first is a concluding statement Daniel gives after the group's discussion of Proverbs 14,15, and 16:

"I wanted to cross-reference Proverbs 16:4 that we were talking about for a while: The Lord works out everything for his own ends, even the wicked for a day of disaster. I wonder if that one's somewhat connected to the Parable of the Weeds and the Wheat where Jesus tells this story about, no, let the weeds grow up with the wheat. At the right time, God works it out; He'll take the harvest. And, if it's not just another word about, hang on, God's watching. God's gonna work things out. And, if you trust in Him it's all gonna come out in the wash."

Second, Daniel frames the group's discussion of Proverbs 17 with an introduction that ended as follows:

"In a number of the verses we're looking at today I think there's an accent on truthfulness. And, maybe John 8:43 would be a good one to focus some of those verses we're looking at. Jesus is kind of squawking with these, some of the Jews. Some of whom, it looks like, believed in Him, but some of them continue to give him a hard time. And, they were talking about, we're Abraham's children. And, Jesus said, I don't think that's who your dad really is. There's a number of accents on, in the chapters we're looking at today, on truth. And, this is a thing I hammer on for the catechism kids. When they're lying, they're speaking Satan's language. It's not unlike what Jesus says about adultery. He said: Okay, that's your goal not to commit adultery. I'm gonna up the stakes. Whoever looks on a woman has already committed adultery."

Lastly, from the group's reading of Proverbs 26 and 27, one participant raises Proverbs 26:12 for discussion: Do you see a man wise in his own eyes? There is more hope for a fool than for him. Daniel responds:

"What's the thing in the New Testament about the Pharisee praying and then the other guy goes in and says: Be merciful to me a sinner. But, about the Pharisee it says: the Pharisee praying to himself. And, the accent was on himself, rather than real prayer. And, this is wise in my own mind."

Clearly this is not the same style of recontextualization practiced by Ted with the Tuesday Men. The interpretive work accomplished in recontextualization is not a matter of defining the relation between the Bible and the modern believer. Daniel's recontextualizations are involved in establishing parallels, but parallels among scriptures. More specifically, they are parallels between the Old and New Testaments. The Parable of the Weeds and the Wheat, Jesus' words about adultery, and the prayer of a Pharisee are introduced in order to posit a commonality in principle between the two testaments. Daniel is embedding an interpretation that links the Old and New Testaments. It is an interpretation that these two testaments do indeed speak to one another, inform one another, and combine to present a single, cohesive narrative. The text-internal emphasis of the group's discourse, particularly their dealings with the category of 'God's

promises,' presents a tension that problematizes the Bible's consistency and clarity.

Daniel's recontextualizations mediate this. Linking Old and New Testament texts affirms the Bible's continuity, balancing the ongoing interpretive dilemmas encountered by the group.

Not every group is as uniform in its interpretive concerns as the Tuesday Men and the LCMS Men. Small group discourse can be characterized by a co-mingling of interests. In these cases, the link between textual orientation and recontextualization results in a variety of styles for incorporating Biblical texts into group discourse. I use two participants from the McGee Men as a case study.

Unlike the other two groups, the McGee Men met every other week. The group is closer in size to the Tuesday Men, with five core participants and nine total members. I have 12 recorded meetings that cover their study of the New Testament book of James. They used a verse-by-verse commentary by the Fundamentalist commentator J. Vernon McGee (1955). The two participants I consider, Joel and Vincent, approached the Bible study experience in markedly different ways. In turn, their recontextualization styles diverge. The following example demonstrates this difference. This interaction comes from their response to James 4:9: Be afflicted and mourn and weep. Let your laughter be turned to mourning and your joy to heaviness:

Jason: I have a little problem with this. I mean, being a follower of God and having God in your life should be a joyful experience.

Dennis: Yep. Definitely.

Joel: But, there's a time to mourn. For instance, right after the Tsunami, with dead people all over the place. There's a time not to celebrate and be joyful, but a time to have a feeling of massive heaviness and mourning. It's just like in the first chapter of James. Now, how do you reconcile this? He says, my brothers, count it all a joy when you fall into all kinds of temptations. In all things that happen to you that are bad, count it a joy. And, so, now he says weep and mourn. They're both part of human existence, aren't they?

. . .

Vincent: This is the way The Message says verse nine: quit playing the field. Hit bottom and cry your eyes out. The fun and games are over. Get serious. Really serious. Get down on your knees before the Master. It's the only way you'll get on your feet. Kind of interesting. I went through a period of time when I was being broken. And, a lot of things happened and I was a Christian, but a couple of people came into my life uninvited. My girlfriend went and bought me a Living Bible. Not because she was a Christian, but because she knew I wanted one and I wasn't going to go into a Bible store and get one. You know, that would indicate that I had some needs and didn't have it all together. And, I started reading it and I got into a Bible study through one of these individuals, and was just really accepted... I was being broken out of my circumstances. My heart was being opened as I was reading and studying and being discipled, if you will. And, then, as I started getting on my feet, very slowly, month after month, year after year, I was able to do more again and got more involved. And, the balance started changing. And, I started being less broken and I started, I was starting to, my attention was, and pretty soon my devotions were getting shorter and shorter. My attention was getting on whatever was interesting at the time, excited and enthusiastic and so on. And, then I, only more recently again, have as circumstances kind of broken me again and the balance has gone back. So that I'm now going in, I'm receiving a lot more instruction and training and preaching in unexpected ways. And, my heart is starting to open up again. So, when I read any of these...I hang on every word because it's so meaningful to me, because my heart is more open and less guarded.

What contrast do the two men enact? What types of interpretive work are they pursuing? In Vincent's case, the focus is clear. How does the Bible speak to my life? How does it apply to me? Where do I see myself in this text? Joel's focus is much closer to the LCMS Men's emphasis. He is concerned with an issue internal to the text, specifically a potential tension between Biblical texts. This co-mingling of interests characterizes the group's discourse as a whole, with several members following in Vincent's stead and several others favoring Joel's approach to reading and interacting with the Bible.

The struggle to balance these interpretive styles proved a difficult one for the group (cf., Chapter 3). Several participants commented to me in private that members have joined and left based on their evaluation of the group's emphasis. One member stopped coming because the group was "not intellectual enough," desiring more text-internal discussion. Yet, another member stopped coming because the group was "too

intellectual," frustrated with the lack of personal application. Clearly, both men were correct in their evaluations given the back and forth movement of the group's discourse.

As with the Tuesday and LC-MS Men, the interpretive orientations of Vincent and Joel are reflected in their recontextualization practice. My analyses show 34 text-external recontextualizations by Vincent, compared to only 10 involved with linking or interpretation. On the other hand, Joel employed 45 text-internal recontextualizations and only four cases of critique or parallel. One example from each will suffice. In response to James 2:11-13, Joel links his reading of the text to another Biblical text:

Joel: I like that last thing I think, everything's summed up in that: Mercy trumps judgment. That's what I think. It trumps judgment. And, then, they might have said the same thing Jesus said. He said, then which does the Pharisee neglect? Justice. Mercy. Love of God. Micah says, what does God require of you? Justice, mercy, and to walk humbly.

In contrast, Vincent looks to scripture for a parallel when discussing the issue of tithing:

Vincent: It's giving back to God ten percent of what he gives me type of thing. And, so there's teaching there. It's not a begging, but it's kind of like informing people of what their, well think about Paul and when he asked for money. I mean, he was pretty direct and almost relentless. But, he, in the letters we have, he says, Give but do not give grudgingly. Don't give beyond your means. Give from your ability to give.

In sum, there is a distinctive interaction taking place in the discourse of small group Bible study. This interaction is between the interpretive orientation of the group and the preferred recontextualization style of its participants. This interaction is not unique to Bible study. As described earlier, the same dynamic has been observed in legal (Mertz 1996), educational (Collins 1996), and theatrical (Bauman 1996) institutions.

What does Biblical recontextualization tell us about Bible study discourse?

I have just illustrated how the practice of recontextualization can be read in light of group discourse. The remainder of this chapter asks what can be observed about group

discourse based on recontextualization. I argue there are (at least) three larger discourses evident in the way participants recontextualize the Bible.

Textual Ideology

I described in chapter 2 the nature of the textual ideology surrounding the Bible amongst the individuals and groups individuals involved in this research. Textual ideology refers to the constellation of beliefs, attitudes, assumptions, and expectations believers hold about the Bible. Six principles form the core of this ideology. The first two principles deal with the Bible's authoritative character as the Word of God and as a timeless textual artifact. The second set of principles encompasses the relevancy of the Bible to modern believers. It is a text that always promises application, and it can be precisely applied to individual circumstances. The final two principles deal with the Bible's textuality, namely its character as a cohesive narrative and the aesthetic perfection of its language. The central claim of this dissertation is that small group discourse occurs against the backdrop of this ideology, creating situations of interaction between textual ideology and textual practice. Recontextualization is one area where this relationship unfolds.

Each of the recontextualization strategies I have described – linking, interpretation, critique, and parallel – are imbued with principles of the textual ideology surrounding the Bible. Linking is closely tied to the principle of the Bible as a cohesive narrative. To suggest that discrete Biblical texts have the same meaning, implication, or application assumes the entire operates as a whole, as opposed to as a collection of disparate texts. The practice of bringing together scriptures – often from different books, testaments, authors, genres, and time periods – accomplishes this ideological work quite

well. The textuality of the Bible is embedded in this style and is restated each time it happens. This principle of textuality is also part of the recontextualization strategy of interpretation. The meta-commentary evident here goes back to the Reformation maxim that scripture should be used to interpret scripture (Hagen et al. 1985). This assumes the ideological principle of authority. Scripture should be used to interpret scripture because no other text or means of interpretation carries the absolute authority and divine sanction that the Bible does. In short, scripture is its own best interpreter because it is the text best qualified to do so. Thus, whenever this style of recontextualization occurs multiple principles of the textual ideology surrounding the Bible are reproduced.

The two text-external strategies are equally infused with ideological principles. The use of Biblical texts for critique and parallel are primarily implicated in the relevance of scripture. This is an appropriate (and, at times, necessary) strategy because the Bible does carry the promise of direct and precise application. These strategies are also bearers of authority principles. The use of Biblical texts to convict or affirm can be read as the ultimate form of conviction and affirmation. As the Word of God, this resource trumps all others for establishing dissonance or commonality. The Bible's timelessness is also embedded in these strategies. Indeed, this is what enables these texts to be used as bases of critique and parallel. Because the Bible is the same today as it was when it was written, it can serve as the source of ideals and standards that are or are not being met. Similarly, this sense of timelessness allows individuals to assert that they and others have the same experiences as Biblical characters and events.

Protestant Discourses

The practice of recontextualization also indexes larger discourses operative among contemporary American Protestants. Ted from the Tuesday Men exemplified the interaction between textual orientation and recontextualization style by continually critiquing the modern Church, on the basis of the Church portrayed in the book of Acts.

Ted's approach to recontextualization also indexes the discourse of church growth within American Protestant culture.

ICUMC has been at its present locale since 1914. The arrival of a controversial pastor in 2000 sparked a series of internal disputes. According to several church members, the church almost closed permanently. In July of 2002 new hope arrived when Ted assumed the role of senior pastor. A seasoned pastor of 20 years, Ted approached the problems at the church as a challenge and an opportunity. The restructuring efforts continue, as Ted orchestrated the launch of a new worship service in late 2006.

Ted's approach to altering the life of the congregation is closely tied to the network of "church growth." Comprised of (mostly non-denominational) institutions, the church growth world of publishers, books, authors, and consultants offers a seemingly endless set of resources for advice and strategy on 'how to have a growing church.' In short, the goal is to (re)plan, (re)organize, and (re)structure your congregation in a way that will attract new people. A relatively recent addition to this broad range of institutions is the "emerging church movement." Ted is quite familiar with the variety of church growth resources, but he is particularly well versed in this sub-genre.

"The 'bible," as Ted classed it, "of the emerging church movement" is Dan Kimball's (2003) The Emerging Church: Vintage Christianity for New Generations. The

author includes a chapter entitled "What is 'Church?" On the opening page he responds to this question: "...the church is not the building, nor is it the movement. The church is the people of God who gather together with a sense of mission (Acts 14:27⁶⁹)" (ibid: 91). Kimball recontextualizes sections from Acts six more times in this chapter. The central message of the book is a call to return to "vintage Christianity." Throughout the book Kimball cites the first century Church of Acts as the model to be recreated. Ted's choice of Acts as the group's book of study is likely a product of his familiarity and interest in cultivating an "emerging church" at ICUMC. Moreover, the interpretive orientation of the group, and by extension Ted's recontextualization style, can be situated within this larger discourse. The constant establishment of dissonance makes a great deal of sense in light of the discursive connection with this movement.

Interactive Norms

A third type of larger discourse that can be observed through recontextualization are the interactive norms that characterize small groups. As described in chapter 1, small groups de-emphasize conflict, opposition, and sustained disagreement. The norm for group discourse is not to argue, not to be contentious. This aligns with the prevailing norm in small group discourse, the desire for spiritual growth. Behind the variety of interpretive orientations presented in this chapter is an expectation of spiritual edification common to all of the project groups. Participants come to small groups "to be

⁶⁹ "On arriving there, they gathered the church together and reported that all that God had done through them and how he had opened the door of faith to the Gentiles." (NIV)

(spiritually) fed," "to grow in their relationship with the Lord," and "to grow in fellowship with one another." The result is a congenial, encouraging environment⁷⁰.

With this normative context in mind, consider Table 16:

	Charlie	Joel	Roland	Vincent	Ted	Sandy	Joy
Support	20	18	4	5	16	4	9
Challenge	3	3	2	3	5	2	2
	Aaron	Daniel	Ned	Darren	Jennifer	Rick	Taylor
Support	9	12	4	1	1	4	9
Challenge	3	-	5	-	-	1	-

Table 16: Recontextualization as interactive support and challenge

These numbers indicate the recontextualizations employed toward two contrasting interactive ends. The imbalance is striking. The top row is consistently, and often markedly, higher. This row represents recontextualizations used to support the contributions of another participant. The second row indicates recontextualizations used to challenge the contributions of a fellow group member. The pattern of recontextualization appears to be involved in the larger discourse of maintaining group norms of mutual edification and spiritual growth. What, then, do we make of Ned, the anomaly in the table?

Ned first began coming to the LC-MS Men on June 2nd, 2005, two meetings prior to their study of Proverbs. At the time, Ned was the newly appointed youth minister s at the church. He had recently graduated from Concordia University, and was 25 years younger than the next youngest group member. Given this position, I was surprised by the role Ned assumed in group discourse.

Despite their enjoyment of debate, the LCMS Men were consistent with the rest of the sample in terms of sustaining an expectation of spiritual growth. None of the

⁷⁰ The reality of the edification norm is not tantamount to an absence of disagreement. Indeed, all of the project groups disagreed about an issue at some point. For some groups, like the LCMS Men, the promise of provocative discussion was a major part of why members found the group attractive.

participants took himself to be the group's devil's advocate, always challenging, always disruptive. Pierce, the non-denominational participant who appeared earlier in the chapter, is a good example. In each of the first three meetings in the Proverbs study, Pierce charged that the group was ignoring the "promises of scripture." After a particularly heated disagreement in the third session, Pierce withdrew from the group and did not return. It struck me, then, when Ned gravitated toward a contentious role. He was quick to correct the contributions of other participants, including those whose contributions were highly valued within the group. The following rebuttal is typical. It followed a comment by a long-time member of the congregation who opposed homosexuals that "call themselves Christians:"

"I think we gotta be careful of some, when we first brought, I believe you said, "They say homosexuality is okay, but yet they call themselves Christians.' Yes, I believe that very scripturally homosexuality is wrong...[But] I'd have a tough time saying no one in the ELCA church is gonna be in heaven because they agree with homosexuality. They profess Jesus as Lord and it's by grace they've been saved. I would agree that the things that have all been said on abortion and homosexuality and absolute truth, with Biblical principles and all these things, is absolutely correct. But, I get kind of shaky when we start calling into question people's salvation because of things they do or things they believe, aside from *Christ alone*.

Although Ned hedges his statement, the disagreement and the challenge are explicit. A closer look at Ned's recontextualizations, however, show that all of the challenges come at the beginning of his tenure with the group. His participation in discussion drops off rather suddenly after the first few meetings. When he eventually returned to being more talkative, the lack of challenges and the increase in supports via recontextualization are obvious. This seeming exception, then, turns out to further support the argument I am making. Ned's example provides us a case of socialization into group discourse, and subsequently, into a more appropriate style of recontextualization. In particular, it is a mode of recontextualization that buttresses an

underlying interactive norm of group discourse – keep things encouraging, helpful, edifying.

Concluding Remarks

I have presented three basic arguments in this chapter.

- ➤ Biblical recontextualization occurs as both direct quotation and as metonymic substitution. In the latter form, narratives are distilled for a core meaning so that only a single name or term is required for recontextualization. References such as "Job" or "The Parable of the..." represent entire lessons and stories.
- The pattern and style of recontextualization align with a group's textual orientation to the Bible. This is true in cases of uniform orientations, as with the Tuesday Men and LCMS Men, and cases of competing orientations, as with the McGee Men.
- ➤ Biblical recontextualization draws attention to larger discourses operative in small groups; for example, textual ideology, culturally salient discourses of American Protestantism, and the interactive norms of group discourse.

My overall goal has been to demonstrate how recontextualization within small groups is consistent with recontextualization work in other institutional settings. The data also speak to macro-level claims about recontextualization. Bauman and Briggs originally noted that this intertextual process contributes to the work of traditionalization, or "the symbolic construction of discursive continuity with a meaningful past" (1990:78). Similarly, Silverstein and Urban remark that recontextualization is a "building block of culture" (1996: 1).

These claims are supported by the significant cultural work accomplished by the recontextualization practice of small group participants. The constitutive features of group discourse – interpretive, ideological, and interactive – are all indexed by and embedded in the various styles of recontextualization. Moreover, the continual invocation of Biblical texts anchors small group discourse as a distinctively Christian practice. As an institutional setting, small groups rely on these micro-level strategies to communicate a shared religious identity, as well as a shared commitment to being rooted in their foundational text.

Chapter 6: Conclusions, Contributions and Further Questions

In this dissertation I have described and analyzed the discourse of small group Bible study. The central theme has been the ways in which participants use various forms of discursive practice to articulate and reflect on the textual ideology they surround the Bible with. My goal has been to demonstrate that this textual ideology is at the center of group discourse, even when it is not the explicit subject of conversation. Through the course of this dissertation, I hope the reader has also gained an appreciation for the complexity, dynamism, and meaningfulness that characterizes small group life. Group members understand this practice to be among the most significant and enjoyable activities they engage collectively as adherents to the same faith. I have attempted to communicate a sense of this importance.

To conclude this dissertation, I address several issues. I begin with a brief review of the previous chapters. I then move to some larger consequences my arguments present for the current shape of American Protestantism. And, I close with a commentary on the broader significance of the central theoretical construct of textual ideology-textual practice.

Chapter Summaries

In the introductory chapter I outlined the theoretical and methodological frameworks that organized my research. The two theoretical frameworks are the anthropology of discourse and the anthropology of Christianity. This study exemplifies how an individual discourse event – small group Bible study – can be analyzed to understand macro-sociological issues – Bible interpretation among American Protestants. I engage the anthropology of Christianity primarily through my analysis of Protestant

Biblicism. This work also contributes the first in-depth, comparative analysis of small group Bible study to the extensive scholarship among anthropologists and sociologists of American religion. I conclude this chapter with a review of the methodology I used to sample, collect, and analyze the project data.

Chapter One describes the six project congregations and the eight Bible study groups that are the focus of my analyses in chapters two through five. The ethnographic portraits of the six churches focus on the issues and events that defined the life of each congregation during the research year. The ethnographic portraits of the eight project groups communicate a sense of the experience of being present in each setting. This chapter also addresses several themes fundamental to small group life: interactive frames, participation structure, text selection, and Bible use.

Chapter Two introduces the theoretical construct of textual ideology. I explicate textual ideology as a concept, then present the six core principles that constitute this ideology for the project groups. These principles signify the qualities of absolute authority, ongoing relevance, and purposeful textuality that believers attribute to the Bible. I also address the inter-relatedness, historicity, sharedness, and canonical indexicality of these principles.

Chapter Three analyzes the discourse of Bible interpretation in the small group setting. I argue that the project groups relied solely on three styles of Bible interpretation: finding application, establishing meaning, and explaining Biblical texts. I review each style, and then discuss their inter-relation in group discourse. Ultimately, I argue that the consistent use of each style, and their intricate relation to textual ideology principles, identifies small groups as interpretive communities.

Chapter Four addresses the textual and experiential resources that Bible study participants bring to this event. The organization of these resources in each group setting form an interpretive matrix that provides a discursive filter for interacting with the chosen texts of study. The use of individual resources in group discourse is shown to embed and (re)articulate textual ideology principles.

Chapter Five examines the practice of recontextualizing the Bible in small group discourse. I address the formal properties of recontextualization, the function of Biblical recontextualization in group discourse, and how this practice reflects wider cultural concerns among American Protestants. The central argument of this chapter is that the interpretive preoccupations in each group are reproduced through the favored style of Biblical recontextualization employed by participants. And, the particular styles of Biblical recontextualization reflect textual ideology principles.

Further Consequences of Textual Ideology, Textual Practice

The dissertation has taken a discourse-centered approach to the analysis of group Bible study, and in particular, the dynamic between textual ideology and textual practice. By its nature, this approach entails detailed attention to the micro-sociological features of conversation and interaction. However, this dissertation also draws out larger issues significant for the study of American Protestantism. These appear in the dissertation – e.g., the link between recontextualization and the priorities of the church growth movement – but I would like to provide a more explicit consideration of such issues here. I present three areas where the arguments of this dissertation reveal larger consequences: the interpretive imagination of American Protestantism; the structuring influence of Protestant traditions; and the importance of small groups as a site of dialogue.

The Interpretive Imgagination of American Protestantism

Bible interpretation in America has been examined in theological, historical, demographic, literary, hermeneutic, and epistemological terms within the social sciences (e.g., Crapanzano 2000; Harding 2000; Malley 2004; Wimbush 2000). This dissertation contributes to the ongoing understanding of the interpretive imagination among American Protestants by employing a discourse-centered analysis. In particular, the arguments I present provide a clear statement on how the discourse of Bible interpretation unfolds among *actual* social actors in *actual* social practice – what a colleague of mine terms "actually, existing Christianity⁷¹."

One lesson of this dissertation is that scholars must move past cliches in depicting how Christian believers interpret the Bible. Understanding this practice in terms of textual ideology and textual practice is far more productive than relying on received categories of 'literalism,' 'inerrancy,' and 'Bible-believing.' As Malley (2004) observes, literalism is not so much a hermeneutic tradition as it is a symbol of identity, a signifier of community membership. Such clichés gloss over the intricate, meaningful, and contested interpretive processes that occur among American Protestants. This dissertation allows for a more nuanced, exacting view of Bible interpretation by identifying specific princples (e.g., authority, relevance, texuality) and styles (e.g., application, establishing meaning, explanation). In short, the interpretive imagination among American Protestants requires attention to the inter-subjective discourse occuring around the issue of Bible interpretation.

Similarly, the data I present clearly illustrate that these believers are active Bible readers. In other words, they do not passively employ methods and strategies received

⁷¹ I am grateful to Jon Bialecki for this phrase.

from the pulpit, Christian media, or the hermeneutic instruction of denominational tradition. Rather, these influences are integrated into group-specific modes and individuals' concerns for Bible interpretation. Yet, this dissertation simultaneously reveals a bounded, restricted interpretive environment being cultivated by small group discourse. These groups of readers are creative, but limited in the range of interpretive procedures they employ. Recall the strict adherence among the Prayer Circle to the style of application, and a similar strictness among the LCMS Men for keeping to issues of establishing meaning. Moreover, recall the predictability of the structure that defines the interpretive process when groups attempt to establish meaning. In short, the interpretive imagination of American Protestants is both creative and restrictive, actively engaged and limited in scope.

Lastly, my analysis of the interpretive matrix demonstrates the important point that processes of individual and collective interpretation do not take place in a cultural vacuum. They take place against the backdrop of a myriad of knowledge forms, texts, and experiences that individuals bring to their respective groups. Bible interpretation is shown to be a practice defined by who readers are, the multiple forms of identity that constitute this sense of Self, and the organization of these Selves in cohesive group environments. This echoes the lessons from scholars interested in "lived religion" (see Hall 1996; Roberts 2005). Scholars give much attention to the forms of Christian media believers consume – what radio programs they listen to, who they watch on television, which websites are visited, and whose books are purchased. This dissertation suggests that attention to these resources must be complemented by what interests Bible readers (e.g., the Iconoclasts and 'history'), how they define themselves (e.g., the LCMS Women

and denominational identity), and what they do beyond their religious lives (e.g., Rick and his experience as an Independent Business Owner). In short, the interpretive imagination of American Protestants emerges from the Protean Self of the modern subject.

The Structuring Influence of Protestant Traditions

The practice of small group Bible study is also shaped by the Protestant traditions that groups are bearers of. The unfolding of group discourse, then, must be situated within the trends and concerns occupying the attention of American Protestantism. The example of the Emerging Church movement being an explanatory context for Pastor Ted's recontextualization practice is a prime demonstration of this dynamic. I consider three other such cases below.

Scholars of American Religion have observed a colonizing effect of contemporary Evangelicalism on other Protestant forms in the United States (e.g., Wuthnow 1988). In short, due to the success of Evangelical institutions in attracting members, other Protestant institutions are adopting the emphases and practices associated with the Evangelical movement. This dissertation demonstrates this trend in two important ways.

First, the predominance and greater cultural capital of 'application' as a style of Bible interpretation is consistent with the Evangelical emphasis on a "personal relationship with Jesus Christ" (i.e., the favoring of individual faith over institutionalized forms of Christianity). As described in Chapter Three, this interpretive style is not only more numerous among the project groups, it also functions as a trump in group discourse. Its capital is revealed by its ability to defuse problematic situations arising from other interpretive styles. This illustrates a greater importance placed on using the Bible as a

medium to establish relevance for the faith of the individual than for engaging in meaning-defining hermeneutic activities.

Second, this same emphasis on increasing faith through increasing the strength of relationships is reflected in the interactive frames of small group Bible study. As described in Chapter Two, the two interactive frames that predominate in this setting are the cultivation of intimacy and the pursuit of analysis. The over-representation of the former within the project sample speaks to the larger cultural trend of Evangelicalism as a colonizing force. Though most groups balanced both frames, greater attention was given in all but a few cases (e.g., LCMS Men) to the frame of intimacy. The McGee Men are an emblematic case of this shift, as the participation of Russell and Vincent continually direct the group toward a greater emphasis on intimacy. The focus on personal relationships distinctive of contemporary Evangelicalism, it seems, extends to the structures of expectation that characterize small group Bible study.

Individual Bible study environments also reflect emphases in denominational history. The LCMS Men provide a clear case of this. The Lutheran Church-Missouri Synod prides itself on being well educated in Biblical history, textuality, and hermeneutics (Noll 1992). This has many manifestations in local congregational life, most notably the requirements of youth catechism classes and new member classes for adult converts. Many of the LCMS Men participants are lifelong Lutherans, and the adult converts have adopted this stress on denominational identity. Unlike the above example involving contemporary Evangelicalism, this dynamic belies what many scholars say about American Religion. The importance they place on denominational identity runs

counter to the majority position that contemporary American Protestantism exists in a "post-denominational" world (see Wuthnow 1988).

Lastly, this dissertation bears out the observation among scholars of American religion that congregations are "local cultures" (e.g., Ammerman 1997). Congregations are said to consist of local stocks of knowledge and activities, which members invoke to satisfy practical needs, establish a sense of identity, adapt to changing community environments, and deal with intra-group conflicts (cf., Becker 1999; Eiesland 2000). This dissertation, particularly the analysis in Chapter Four regarding the interpretive matrix, illustrates how the 'local culture' of congregations is also an interpretive resource. Bible study participants look to the life of the local congregation as they read and discuss Biblical texts, and the issues of belief and practice those texts present. Congregations do indeed contribute to how its members go about making sense of the world and the situations they encounter.

Small Groups as a Site of Dialogue

This dissertation represents the first thorough examination of small group Bible study as a cultural practice. Other scholars (Crapanzano 2000; Davie 1995; Lehtinen 2005; Malley 2004; Roberts 2005; Wuthnow 1994b) have presented a focused analysis of group Bible study, but none can claim to be systematic, comparative, and in-depth. The arguments I present confirm and extend the central claim of this previous research – that small group Bible study is unique in congregational life as a potential site of open, critical, and reflexive dialogue.

I demonstrate in my analyses that small group Bible study is integral to the formation, affirmation, and negotiation of belief and identity among American

Protestants. The small group experience allows participants to engage issues central to their faith, and to do so collectively. What occurs in small group Bible study has direct consequence for individual faith and local congregational life, and ultimately reflects issues of macro-sociological concern (e.g., the colonizing force of American Evangelicalism, the status of Bible interpretation among American Protestants).

I emphasize throughout this dissertation that small group Bible study is a structured and a structuring institution, central to the meaning making processes whereby American Protestants publicly communicate who they are, who they are not, and what they care about. Using a comparative ethnographic approach, I have argued that much of small group discourse is organized around the textual ideology that surrounds the Bible. How groups read, understand, use, interpret, and talk about scripture is central to understanding the nature of small group life, as well as the current shape of American Protestantism. Thus, this dissertation is a call for further research into a complex institutional form significant to the changing shape of Protestantism in the United States.

Broader Significance

This dissertation is primarily of interest to scholars concerned with questions regarding discourse and Christianity. However, I believe the arguments I have presented are relevant beyond these particular domains. I conclude, then, with some observations and suggestions for how my analyses could provide insight for other social scientists. Several concepts employed in this dissertation are already well-developed in the anthropological and social science literature, for example: "discourse," "interpretive communities," "interpretive resources," and "recontextualization." I hope my application of them will aid other scholars to find further uses for these analytical tools, other ways

of directing them toward research questions, and unexpected nuances for integrating them into analyses. For other concepts, however, I believe this research offers a source of creative insight. The primary example is the issue of textual ideology.

Textual Ideology

The central concept employed in this dissertation – textual ideology – is also the aspect of this work that should be of most interest to other scholars. I have used the relationship between American Protestants and their sacred text, the Bible, as a case study to demonstrate the importance of this concept. Below, I identify three possibilities for a broader application of textual ideology, although I suspect there are numerous other areas where this would be fruitful.

1) Comparative Scripturalism. In his ethnography of Evangelical Biblicism, Brian Malley (2004) suggests the need for an anthropology of comparative scripturalism. That is, anthropologists should devote ethnographic and theoretical attention to the ways in which different religious traditions construct relationships with their respective sacred texts. I argue that the concept of textual ideology is one axis point at which such research could be conducted.

In this dissertation, I have demonstrated multiple ways that American Protestants rely on and integrate the textual ideology surrounding the Bible into their religious discourse. A comparative framework would consider how these issues apply to other Christian and non-Christian traditions. I propose the following as other useful research cases: Progressive Christians and the Bible; Native Christians and the Bible; Immigrant Christians and the Bible; the Latter Day Saints and the Book of Mormon (and the Bible); Orthodox and Reformed Jews and the Old Testament; Muslims and the Quran; varieties

of Buddhism and the Tripitaka; and varieties of Hinduism and the Vedas, Upanishads, and Bhagavad-Gita.

Research in comparative scripturalism would allow for greater understanding of the role of sacred texts in the hermeneutic and practical imaginations of different religious traditions, and how adherents from those traditions use scriptures in processes of social practice. Textual ideology is a concept that can be assessed in each of the traditions listed above to pursue these relationships. Understanding the assumptions, beliefs, and expectations religious readers bring to their respective texts also opens up a space in which to consider the points of intersection and divergence among different religious traditions. The dynamic of textual ideology and textual practice can be used to explore questions of discourse (as I have done in this dissertation); the relation between text and action; the link between religion and literacy; intertextual acts of performance; and narrative constructions of reality. The analysis of textual ideology may be the ideal place to initiate a comparative scripturalism given the foundational role that scriptural texts (and their accompanying ideologies) play in many religious communities.

Of course, it is important to note that I am not advocating the simple exportation of my analysis in this dissertation to the study of the Bible and other scriptures in other religious contexts. The Bible as a text has experienced a particular trajectory in the history of Protestantism, and American Protestantism specifcally (Noll 1992). Malley (2004) also observes this, noting that the Bible as an object of interpretation is indicative of Protestantism but not other traditions, which may place greater emphasis on artifactual and ritualistic uses of sacred texts. This history allows for a current situation where readers do carry a defined set of well-articulated ideological principles. One cannot

assume this is the case in other settings. For example, a study of the textual ideology surrounding the Bible is likely to look very different among American Catholics, whose Christian tradition has had a fundamentally different relationship with the Bible (Noll 1992). This is equally true for the study of scriptures in other religious traditions. Yet, some form of textual ideology will exist and this dissertation provides an example of why such analysis is useful. I suspect the primary task in using textual ideology as a structuring concept for a comparative scripturalism is to identify the theoretical foundation underlying particular ideological principles (e.g., 'Authority' underlies "the Bible is the Word of God" and "the Bible is the same today as when it was written"). Once elaborated, these theoretical bases could then be used to address how textual ideology is operative in a particular cultural setting.

2) Legal Anthropology. Textual ideology is also applicable to anthropological studies of text-based legal institutions. Like religious readers, legal readers in Western and non-Western settings operate in a highly intertextual universe. They constantly interact with national Constitutions, state and local Constitutions, law books, and other judicial statements. Understanding the expectations, beliefs, and assumptions readers have about these texts are fundamental to understanding how they interact with and deploy them in legal decision-making processes.

Elizabeth Mertz (1996), Vincent Crapanzano (2000) and others (e.g., Philips 1998) have conducted research on related questions regarding the role of texts in legal institutions. An analysis of the ideologies surrounding legal texts would aid further research into questions of hermeneutic practices, intertextual processes of reasoning and meaning construction, and the relationship between legal reading and legal practice. As

with comparative scripturalism, a crucial task is to identify the extent to which communities of readers share ideological principles, and to what extent such ideologies are actively contested. Such points of divisiveness are tremendously productive for observing cultural assumptions and reasoning processes in action.

3) Anthropology of Education. A third area where textual ideology promises to be a useful concept is for anthropological studies of educational institutions. Primary, secondary, and post-secondary pedagogies are often organized around textbooks.

Different types of social actors (e.g., administrators, teachers, and students) may read these texts in quite different manners. This can be true for certain genres of texts as well as specific texts. An analysis of the relationship between textual ideology and textual practice for different social actors in educational institutions would provide key insights into pedagogical techniques and problems, learning styles and difficulties, and sources of tension between different educational roles.

Cross-cultural research on literacy is a particular area where textual ideology is applicable. Anthropologists have devoted extensive attention to the processes and practices of literacy development in predominantly non-literate cultural settings (e.g., Street, ed. 1991). Textual ideology has a distinct contribution to make to this important research. It raises a fundamental question: what set of expectations and beliefs do individuals and groups bring to particular genres and specific texts. Attention to the issue of textual ideology can shed light on the decision-making processes behind what texts are chosen to teach reading, and what texts are pursued after a functional literacy has been achieved. Anthropological studies of literacy have repeatedly demonstrated that learning to read means learning other cultural attitudes and assumptions as well, in particular those

that are ideological in nature. Again, understanding the textual ideologies surrounding genres and specific texts aids the anthropological attempt to explore what else readers learn besides the ability to read.

In this dissertation, I have used the Bible as a case study to demonstrate the usefulness of textual ideology in conjunction with a discourse-centered framework. In the previous six chapters I have presented arguments that may be applied by scholars to other areas of inquiry, including studies in comparative scripturalism, legal anthropology, and the anthropology of education. And, undoubtedly, ethnographic work in these areas will raise new questions and insights in regard to the utility of textual ideology for anthropological research.

APPENDICES

APPENDIX A

Sample Transcript Page

Transcription Conventions

Italics: Recontextualization **Bold**: Interpretive Resource =//: Interruption/overlap

{}: Pause

((laughs)): Laughs

[[...]]: Contextual information

{...}: Words unclear CAPS: Emphasis

Underline: Reported speech

TRANSCRIPT: T001.2c.010405

LOCATION: Second Floor Lounge - Church Name

PARTICIPANTS: Name (BH); Name (DS); Name (FA); Name (JP); James Bielo (JSB)

TOPIC: Thru the Bible Series: James – Week 1 (James 1:23-25)

DS: [[Reads James 1:23-25]] [James 1:23-25] He has several pages of comments on this. {5.0}

DS: I think he makes a, he makes an interesting little comment on page 41 concerning the fact that in many cases we are the only Bible that some people read. He says, we need a new translation of the Bible that will be known as the Doers Translation. That's a good translation, the Doers Translation [McGee], he says, alluding to the fact that in the things we do we often reflect our faith and other people see that in us whether we want them to or not. It's kind of a two-edged sword I think. If we profess to being Christian and those that we contact through work or socially or whatever know that we profess this faith, then they come to some judgment about how effective we are at it in terms of their perception, at least, of how they see Christians living out their life. The little poem at the bottom of the page is sort of interesting, The Gospel is written a chapter a day, by deeds that you do and by words that you say. Mean read what you say, whether faithless or true, say, what is the Gospel according to you? [McGee] You have comments? BH: Well, I just have my usual comment on what they call the Pauline paradox. If Paul were writing that he'd probably say be ye hearers of the word, and not doers only [Paul]. DS: Oh.

J: Well, I think it might be well to look at a little history of James. See James didn't get into the canon until the third century, and neither did Revelation. The canon was formed, I think it was at, what, the Council of Nicea? And=

BH: //Yeah.

J: =Then, in the second century, other books were added. But, James and Revelation did not come along until the third century as accepted. And, I think part of the reason was the dilemma that you mentioned. They felt it was a little bit counter to Paul.

BH: Mhmm.

J: You know, believe and that's all. And, James comes along and says faith without works is dead, doesn't mean anything [James]. And, so I think that contradiction appeared, but really it is NOT a contradiction.

BH: Not, not as much as it appears.

APPENDIX B

Research Questionnaire

Dissertation Research on Small Group and Bible Study Life: Questionnaire

Dear Bible Study Member -
Allow me to thank you right away! I know everyone keeps a busy schedule. Your completion of this questionnaire is greatly appreciated, and will be a tremendous help to the quality of the research. There are two sections in this questionnaire, each of which should take between 15 and 20 minutes to complete.
The information you provide in Section A will be used for two purposes: to enrich my observations of your group's Bible studies; and to better understand the scope and variety of experiences that are brought to Bible study.
Section B is designed to gather information regarding your attitudes toward, and views of, the Bible. It asks you to reflect on how you understand the nature of the Bible. This information will be used to further my understanding of how the Bible is read and discussed in your group's meetings.
Your continued participation in this project is vital to its success! I'll look forward to receiving the completed questionnaire back from you. Again, my thanks are abundant!

Section A: Autobiographical Information
BACKGROUND INFORMATION:
Year you were born:
You are: [] Male [] Female
Current Marital Status: (place an "x" between the [])
[] Never Married [] Divorced [] Widowed

[] Living with Partner

[] Separated

[] Married

How many children do you have?			
Check your highest level of education:			
[] Less than 8 th grade (grammar school)			
[] Through 8 th grade (grammar school)	[] Some	high schoo	I
[] High school graduate	[] Some	college	
[] Technical school	[] Comp	leted colleg	e
[] Some graduate school	[] Comp	leted gradu	ate schoo
What do you do for a living? (Or, what did you have worked in multiple occupations, please) If you
How many years have you worked at this occup	ation?		
How many years have you lived in [location]?			
Have you lived elsewhere? [] Yes []] No		
If Yes, please list where and when?			
Please indicate the political party whose candid	dates you mo	st often vo	te for in
] Independer	nt	
[] Democrat] Other:		
**************************************	*****	*****	****
How long have you been at your present church	h?	_	
Are you an official or registered member of the	his church?	[] YES	[] NO
Have you ever been a member of (or regularly denomination or a non-denominational church?		church of (another [] NO
denomination of a non-denominational criufch?		נן יכט	

•	d YES, which denoming as apply to you)	ations or non-d	lenomii	national churches?
·	[] Southern Baptist	[] American B	aptist	[] Independent
[] Catholic	•		•	
[] Episcop				
•	[] Missouri Synod	[] ELCA	[] WE	LS
	t: [] United Me		[] Fre	e Methodist
[] Presbyt	terian			
[] United	Church of Christ (UCC	C)		
[] Disciple	es of Christ			
[] Brethro	en			
[] Church	of Christ			
[] Church	of the Nazarene			
[] Wesley	an			
[] Assemb	olies of God			
[] Vineyar	d Fellowship			
[] Other(:	s), please list			
	ominations, or non-der neck as many as apply) [] Southern Baptist			, ,
[] Catholic	c			
[] Episcop	al			
Lutheran:	[] Missouri Synod	[] ELCA	[] WE	LS
Methodis	t: [] United Me	thodist	[] Fre	e Methodist
[] Presbyt	terian			
[] United	Church of Christ (UCC	c)		
[] Disciple	es of Christ			
[] Brethro				
[] Church	of Christ			
	of the Nazarene			
[] Wesley				
[] Assemb	dies of God			
	d Fellowshin			

[] Other(s), please list
On the scale below, indicate by bolding (or, circling) a number how important it is for you to be in your present denomination?
Not important Very important 1 2 3 4 5 6 7 8 9 10
For you, what are the three most important features of your denomination?
1.
2.
3 .

How often do you attend Sunday worship service at your church? [] 3-4 times a month [] 1-2 times a month [] Less than once a month
Are you now a church officer (for example, elder, deacon, board member)? [] Yes [] No
Have you ever held an elected position at your current church? [] Yes [] No

BIBLE STUDY PARTICIPATION:
How long have you been a member of this Bible study group?
Are you now a member of other Bible studies at your church? Yes [] No []

Are you now a church?	member of a Yes []	Bible study tl No []	hat does not m	eet as part o	f your
If Yes, plea	se describe:				
In general, do religious or spi	•	• -		or a non-scrip n-scriptural bo	
Why do you	prefer that t	type of book?			
Are you now, o Yes []	r have you evo No []	er, been a lea	der of a Bible :	study?	
If Yes, wha study?	t do you prefe Leading []	_	Bible study, or t leading []	_	
Why do yo	ou prefer this	role?			
Do you prepare assigned readi	•		•		
If Yes, ho	w much time o	do you spend p	oreparing?		
On the scale b is for you to be			_	umber how im	portant it
Not important 1 2	3 4	5	- 6 7	Very impo 8 9 ·	rtant 10
For you, what o	are the three	most importo	ant qualities of	this Bible st	udy group?
1.					•

If "Yes", please describe:

2.

3.

, , ,	st or cable	news?	[]Yes	[] No
If "Yes", please check whi	ch network	(s) you us	ually watch	١.
[] ABC	[] NB	·	•	
[] <i>C</i> BS	[] FO	X		
[] CNN	[] MS	SNBC		
[] Headline News	[] C-S	SPAN		
[] PBS				
[] Other, please list: _		-		
o you regularly read the newsp	aper?	[] Yes	[] N	lo
If "Yes", please check which	n newspapeı	r you usu	ally read.	
[] State Journal	• •	•	•	
[] New York Times				
[] USA Today				
[] Washington Post				
[] Wall Street Tours	•			
[] Wall Street Journa	l			
[] Other, please list: _				
[] Other, please list: _				her thar
				her than
[] Other, please list: _ o you subscribe to, or regularly ewspapers? [] Yes [] No	y read, any	news pub	lications ot	her than
[] Other, please list: _ o you subscribe to, or regularly	y read, any	news pub	lications ot	her than
[] Other, please list: _ you subscribe to, or regularly wspapers? [] Yes [] No If "Yes", please check all pu	y read, any	news pub	lications ot	her than
[] Other, please list: _ o you subscribe to, or regularly ewspapers? [] Yes [] No If "Yes", please check all pu [] Time	y read, any Iblications 1	news pub	lications ot	her than
[] Other, please list: _ you subscribe to, or regularly wspapers? [] Yes [] No If "Yes", please check all pu [] Time [] Newsweek	y read, any oblications t	news pub	lications ot	her than
[] Other, please list: _ o you subscribe to, or regularly ewspapers? [] Yes [] No If "Yes", please check all pu [] Time [] Newsweek [] US News and World [] Other, please list: _	y read, any oblications t	news pub	lications ot	
[] Other, please list: _ o you subscribe to, or regularly wspapers? [] Yes [] No If "Yes", please check all pu [] Time [] Newsweek [] US News and World	y read, any oblications t	news pub	lications ot	

	If "Yes", pleas	•		isten to: _			
Do	you watch Chri If "Yes", pleas		•				
Do	you subscribe · If "Yes", pleas	_	•		•		
Do	you regularly c If "Yes", pleas	•					
i stic Do	you regularly r	Authors: ead books tments cle	by Christi early have (an authors	(that is,	ones wl	hose religio
	[] Yes How many boo	·	istian auth	•		•	•
Do	you have a favo [] Yes Who is the au	[] N		r(s), who is	: NOT a l	Bible au	thor?
	What texts h		ad by him/	her?			

Do you have a f	avorite Chris [] N	stian text(s), th lo	nat is NOT th	ne Bible?	
What text	(s) is this?				
Who is the	author?				
Have these tex study?	ts and autho	ors been helpful [] No	for your par	ticipation in B	ible
If Yes, how	v so?				
Have these tex Bible?	ts and autho [] Yes	ors been helpful [] No	for your und	lerstanding of	the
If Yes, hov	v so?				
Do you have a f different from			•	nay be the san [] No	ne or
If Yes, who	is this comr	mentator?			

Section B: Understandings of the Bible

SPECIFIC QUESTIONS ON THE BIBLE:

How often do you read the Bible outside of church-related activities [] Everyday [] More than once a week, but NOT everyday [] Once a week [] Twice a month [] Once a month [] Rarely
Do you usually underline or highlight your Bible when you read it? [] Yes [] No
If Yes, why do you do this?
Do you take notes or journal your thoughts when you read the Bible? [] Yes [] No
If Yes, why do you do this?
Which version, or translation, of the Bible do you use most often? [] Good News Bible [] King James Version [] New King James Version [] Living Bible [] New Living Translation [] The Message [] American Standard Version [] New English Bible [] New International Version [] Today NIV [] Revised Standard Version [] New Revised Standard Version [] Other, please list:

Why do you use this translation most often? Have you always used this version, or translation? []Yes [] No If "No", what did you use previously? Why did you choose a new version or translation? Do you usually use a Study Bible? [] Yes [] No If "Yes", which Study Bible is it and why did you choose it? Is there a version, or translation, that you prefer never to use? [] Yes [] No If "Yes", which version or translation is that? How often do you compare how different versions of the Bible translate the same passage(s)? Do you have a favorite book, or section, of the Bible? [] Yes [] No If "Yes", which book, or section, is it?

[] Yes

[] No

Do you have a favorite verse in the Bible?

If "Yes", which verse is it?

Do you find some books/authors of the Bible easier to memorize than

[] Yes	[] No	If "Yes", what books/authors?
Are there boo	oks in the Bible y	ou struggle to believe?
[] Yes	[] No	If "Yes", what books?
Are there boo	•	ou struggle to understand? If "Yes", what books?

BIBLE READING:

Section 1:

When I don't understand something in the Bible, or want to better understand something in the Bible, I do the following things. (Please check the appropriate response to each statement by placing an "x" in the box.)

	Never	Some- times	Often
Consult a study aid			
(like a concordance or Bible dictionary)			
Consult the footnotes in my annotated Bible			
Consult my favorite Christian author			
Consult a different section in the Bible			
Consult sources for the original Bible languages			
Bring up the question in Bible Study			
Go to the library			
Go to the Internet			
Ask my spouse			
Ask a friend			
Ask a family member			
Ask the pastor			
Ask a knowledgeable congregation member			
Pray for guidance			
Wait for the Holy Spirit to guide me			
Other, please describe briefly:			

					l
1			i		ľ
					l .
1					1
1				 	

Section 2:

The following section asks you to compare your experience of reading the Bible with other reading experiences. For each option, please check how similar or dissimilar you feel the experience is compared to when you read the Bible. (Please check the appropriate response to each statement by placing an "x" in the box.)

Reading the Old	Very	Similar	1	Different	Very
Testament is like	Similar		Opinion		Different
Reading a history book					
Reading a science book					
Reading poetry					
Reading a novel					
Reading a guide for life					
Reading a law book					
Reading a newspaper					
Reading a self-help book					
Reading a biography					
Reading a theology book					
Reading a comic strip					
Reading a letter					
addressed to you				<u> </u>	

Reading the New	Very	Similar	No	Different	Very
Testament is like	Similar		Opinion		Different
Reading a history book					
Reading a science book					
Reading poetry					
Reading a novel					
Reading a guide for life					
Reading a law book					
Reading a newspaper					
Reading a self-help book					
Reading a biography					
Reading a theology book					
Reading a comic strip					
Reading a letter					
addressed to you					

ATTITUDES TOWARD THE BIBLE AND BIBLE READING

Please indicate how much you agree or disagree with the following statements by placing an "X" in the appropriate box.

	Strongly Disagree	Disagree	No Opinion	Agree	Strongly Agree
1) When I have a problem in	o rough or			<u> </u>	
my life, I often turn to the					
Bible.					
2) When I experience great					
joy in my life, I often turn to					
the Bible.	_				
3) I often use the Bible to					
console a friend/family					
member in need.					
4) I rarely see any					
relationship between the Bible					
and my own life.					
5) When I read the Bible, I					
often feel that it mirrors my					
own life experience.					
6) I am reminded of particular					
Bible verses in the course of					
my daily life.					
7) I have made major life	!				
decisions based on the Bible.					
8) I often find the Bible					
relevant for understanding	:				
national and global issues.					
9) I only read the sections of					
the Bible that my study group					
is reading.					
10) I regularly read the Bible					
by reading at the first place I					
open it to.					
11) A devotional book guides					
my reading of the Bible.					

12) 7	T	1			T
12) I use a study guide or					
commentary when I read the					
Bible.		<u> </u>			
13) The Holy Spirit guides				:	
where I will read in the Bible.					
14) The Holy Spirit is present				,	
when our Bible study group					
meets.					ļ
15) I believe the Bible is					
divinely inspired.			ļ		
16) I believe God guided the					
formation of the Bible as it					
now appears.					
17) I believe the Bible is the					
only text inspired by God.					
	Strongly	Disagree	No	Agree	Strongly
	Disagree	ļ	Opinion		Agree
18) I believe the Bible is					
authoritative over any other					
text.					
19) I believe the Bible is					
completely without error.					
20) I believe the Bible					
contains NO inconsistencies.					
21) I believe parts of the					
Bible are NOT applicable in					
today's world.					
22) I believe the meaning of					
the Bible is the same today as					
it was when it was written.					
23) I believe the Bible should					
be used as the basis for					
current legal decisions.					
24) I believe the Old and New					
Testaments are equally					
important to the Christian					
faith.					
25) I spend equal amount of					
time reading and studying the					
Old and New Testaments.				1	
26) I believe the Bible is the			 		
Lo, I believe the bible is the		1			1

					,
best way to communicate with					
God.	ļ				
27) The Bible contains					
answers to all of life's					
questions.					
28) I believe the entire Bible				ł	
should be read literally.					
29) I believe it is important to					
know who wrote the books of					
the Bible, and when they were					
written.					
30) I believe it is important to					
consult the original languages					
of Biblical texts.					
31) I believe it is important					
that I be the one to consult					
the original languages.					
32) I believe it is important to					
know the cultural context of					
the Bible's authors and					
audiences.					
33) I believe the ability to					
memorize scripture is a gift.					
34) I believe it is important to					
know when archaeological or					
historical evidence may					
conflict with the Bible.					
35) I believe it is important to					
know when archaeological or					
historical evidence may					
support the Bible.					
	Strongly	Disagree	No	Agree	Strongly
	Disagree		Opinion		Agree
36) I have experienced					
getting different meanings					
when I have read the same					
passage of scripture at					
different times.					
37) It bothers me when there					
is disagreement during Bible					
study regarding the meaning					

of a particular passage of scripture.			
38) I prefer to concentrate on one Bible verse at a time when I am reading.			
39) I prefer to consider a section of verses together when I am reading the Bible.			
40) I feel free to disagree with things I read in the Old Testament.			
41) I feel free to disagree with things I read in the Gospels.			
42) I feel free to disagree with things I read in the rest of the New Testament.		•	

THE END!

I'd like to extend one last thank you! Your completion of this questionnaire makes this a truly cooperative research project!

APPENDIX C

Sample Study Guide Sheet for Tuesday Men

You Will Be My Witnesses <u>Acts 1</u>

I remember telling a friend, "If I were dying, what I would most need would be confidence that all I believed about Jesus were true. I would want you to read Scripture to me, pray with me, and talk to me about Jesus and heaven." This confidence is not only what I need most when I am dying, but also when I am living.

During the days between his resurrection and ascension, Jesus built the confidence of his disciples. He demonstrated and spoke truth about himself. And then he left them with a clearly defined task and the promise of the power to carry out that task. Thus, Luke was able to write with confidence to Theophilus about Jesus.

- 1. When has your faith in Jesus Christ been encouraged by the words of others?
- 2. Read Acts 1:1-11. What did Luke, the author of Acts, report to Theophilus about Jesus' last days upon earth? What gives credibility to this report?
- 3. How is your hope and confidence in Jesus affected by what you learn of him in this passage?
- 4. Put yourself in the shoes of the apostles. How would you feel if you were the first to be given the task described in verse 8?
- 5. How are we equipped for this task, according to the passage?
- 6. Imagine that you were there, looking into the sky. How do you think the followers of Jesus were affected by the promise that he would return?
- 7. How are you affected by that promise today?

REFERENCES

PRIMARY REFERENCES

Arterburn, Stephen, et al.

2000 Every Man's Battle: Winning the War on Sexual Temptation One Victory at a Time. Waterbrook Press.

Brown, Dan

2003 The Da Vinci Code. New York: Doubleday.

Chopra, Deepak

2001 How to Know God: the Soul's Journey into the Mystery of Mysteries. New York: Random House.

Cole, Neil

1999 Cultivating a Life for God. ChurchSmart Resources.

Ehrman, Bart D.

2003 The Lost Christianities: the Battles for Scripture and the Faiths We Never Knew. New York and Oxford: Oxford University Press.

Harnish, James

Journey to the Center of the Faith: an explorer's guide to Christian living.

Nashville: Abingdon Press.

Jackson, Bill

1999 The Quest for the Radical Middle: a history of Vineyard. Cape Town: Vineyard International.

Jackson, Wayne

2003 "The Enigmatic William Barclay." http://www.christiancourier.com/articles/read/the enigmatic william barclay

Kimball, Dan

2003 The Emerging Church: vintage Christianity for new generations. Grand Rapids, MI: Zondervan Publishers.

Kiyoski, Robert T. and Sharon Lechter

1998 Cash Flow Quadrant: Rich dad's guide to financial freedom. New York: Warner Books.

Maxwell, John

1998 The 21 Irrefutable Laws of Leadership. Nelson Business.

2003 The Maxwell Leadership Bible: Lessons in Leadership from the Word of God. Nelson Bibles.

2004 Today Matters: 12 Daily Practices to Guarantee Tomorrow' Success. Center Street.

McGee, J. Vernon

1955 James (Thru the Bible). Nashville, TN: Thomas Nelson Publishers.

Moore, Beth

1997 To Live is Christ: the life and ministry of Paul. LifeWay Christian Resources.

Osteen, Joel

2004 Your Best Life Now: seven steps to living at your full potential. Faithwords.

Yancey, Philip

1995 The Jesus I Never Knew. Grand Rapids, MI: Zondervan Publishers.

Warren, Rick

2003 The Purpose Driven Life. Grand Rapids, MI: Zondervan Publishers.

Wilkerson, Bruce

2000 The Prayer of Jabez. Sisters, OR: Multnomah Publishers.

Zondervan

2005 The Story: Encounter the Story of Scripture in a whole new way. Grand Rapids, MI: Zondervan Publishers.

SECONDARY REFERENCES

Abu-Lughod, Lila and Catherine A. Lutz

"Introduction: emotion, discourse, and the politics of everyday life." In Language and the Politics of Emotion. Catherine A. Lutz and Lila Abu-Lughod, eds. Cambridge: Cambridge University Press. Pp. 1-13.

Ammerman, Nancy T.

1997 Congregation and Community. New Brunswick, N.J.: Rutgers University Press.

Bahktin, Mikhail

[1934] 1981 The Dialogic Imagination. Austin: University of Texas Press.

Balmer, Randall

[1989] 2006 Mine Eyes Have Seen the Glory: a journey into the evangelical subculture of America. New York, Oxford: Oxford University Press.

Bartkowski, John

1996 "Beyond Biblical Literalism and Inerrancy: Conservative Protestants and the Hermeneutic Interpretation of Scripture." IN *Sociology of Religion*. 57(3):259-272.

Bass, Dorothy

"Congregations and the Bearing of Traditions" In American
Congregations: Volume 2: New Perspectives in the Study of
Congregations. James P. Wind and James W. Lewis, eds. Chicago:
University of Chicago Press. Pp. 169-191.

Basso, Keith

1990 Western Apache Language and Culture: essays in linguistic anthropology. Tucson: University of Arizona Press.

Bauman, Richard

- 1996 "Transformations of the Word in the Production of Mexican Festival Drama" In *Natural Histories of Discourse*. Michael Silverstein and Greg Urban (eds.) Chicago: University of Chicago Press. Pp. 301-327.
- 2004 A World of Others' Words: Cross-Cultural Perspectives on Intertextuality. London: Blackwell Publishing.

Bauman, Richard and Charles L. Briggs

"Poetics and Performance as Critical Perspectives on Language and Social Life" IN Annual Review of Anthropology. 19:59-88.

Bauman, Richard and Joel Sherzer (eds.)

1974 Explorations in the Ethnography of Speaking. Cambridge: Cambridge University Press.

Bauman, Richard and Joel Sherzer

1975 "The Ethnography of Speaking." Annual Review of Anthropology. 4:95-119.

Becker, Penny Edgell

1999 Congregations in Conflict: Cultural Models of Local Religious Life. Cambridge: Cambridge University Press.

Bernard, H. Russell

2006 Research Methods in Anthropology: Qualitative and Quantitative Approaches. Fourth Edition. Walnut Creek: AltaMira Press.

Bielo, James S.

"'Walking in the Spirit of Blood': moral identity among born-again Christians." *Ethnology*. Summer (3): 271-289.

Bourdieu, Pierre

1977 Outline of a Theory of Practice. Cambridge: Cambridge University Press.

Bowen, John

1993 Muslims through Discourse: religion and ritual in Gayo Society.
Princeton: Princeton University Press.

Boyarin, Jonathan (ed.)

1992 The Ethnography of Reading. Berkeley: University of California Press.

Boyarin, Jonathan

"Voices Around the Text: the ethnography of reading at Mesivta Tifereth Jerusalem." In *The Ethnography of Reading*. Jonathan Boyarin (ed.) Berkeley: University of California Press.

Briggs, Charles L. and Richard Bauman

1992 "Genre, Intertextuality, and Social Power." *Journal of Linguistic Anthropology* 2(2): 131-72.

Burt, Ben

1994 Tradition and Christianity: The Colonial Transformation of a Solomon Islands Society. Chur, Switzerland: Harwood Academic Press.

Cannell, Fenella (ed.)

2006 The Anthropology of Christianity. Durham: Duke University Press.

Collins, James

1996 "Socialization to Text: Structure and Contradiction in Schooled Literacy" In *Natural Histories of Discourse*. Michael Silverstein and Greg Urban (eds.) Chicago: University of Chicago Press. Pp. 203-228.

Comaroff, Jean and John Comaroff

1991 Of Revelation and Revolution: Christianity, Colonialism, and Consciousness in South Africa. Vol I. Chicago: University of Chicago Press.

Crapanzano, Vincent

2000 Serving the Word: Literalism in America from the Pulpit to the Bench. New York: New Press.

Davie, Jody Shapiro

1995 Women in the Presence: constructing community and seeking spirituality in mainline Protestantism. Philadelphia: University of Pennsylvania Press.

Dewey, David

2004 A User's Guide to Bible Translations: Making the Most of Different Versions. Downer's Grove, IL: InterVarsity Press.

Drew, Rob

2001 Karaoke Nights: an ethnographic rhapsody. Walnut Creek: AltaMira Press.

Duranti, Alessandro

2003 "Language as Culture in U.S. Anthropology." *Current Anthropology*. 44(3): 323-347.

Eiesland, Nancy L.

2000 A Particular Place: Urban Restructuring and Religious Ecology in a Southern Exurb. New Brunswick, N.J.: Rutgers University Press.

Elisha, Omri

"Sins of Our Soccer Moms: Servant Evangelism and the Spiritual Injuries of Class." IN Local Actions: Cultural Activism, Power, and Public Life in America. Melissa Checker and Maggie Fishman, eds. New York: Columbia University Press. Pp: 136-158.

Engelke, Matthew

2004 "Text and Performance in an African Church: The Book 'live and direct'." *American Ethnologist.* 31(1): 76-91.

Fairclough, Norman

1989 Language and Power. London: Longman.

Fish, Stanley

1980 Is There a Text in this Class? The Authority of Interpretive Communities.

Cambridge: Harvard University Press.

Forstorp, Per-Anders

1990 "Receiving and Responding: Ways of taking from the Bible" In Bible Reading in Sweden: Studies related to the translation of the New Testament 1981. Gunnar Hansson, ed. Stockholm, Sweden: Almqvist and Wiksell International. Pp. 149-169.

Foucault, Michel

"The Order of Discourse" In *Untying the Text*. Robert Young, ed. Boston and London: Routledge and Kegan Paul. Pp. 51-77.

Frake, Charles O.

"How to Ask for a Drink in Subanun." *American Anthropologist*. 66(2): 127-32.

Frye, Northrup

1981 The Great Code: the Bible and Literature. New York: Harcourt Brace.

Geertz, Clifford

"Deep Play: notes on the Balinese cockfight" IN *The Interpretation of Cultures: selected essays*, by Clifford Geertz. New York: Free Press. Pp. 412-53.

Goffman, Erving

1959 The Presentation of Self in Everyday Life. New York: Bantam Books.

Goodwin, Charles and John Heritage

"Conversation Analysis" In Annual Review of Anthropology. 10: 283-307.

Grice, H.P.

"Logic and Conversation" In Syntax and Semantics, vol. 3: Speech Arts.

Peter Cole and Jerry L. Morgan, eds. New York: Academic Press. Pp: 41-58.

Gumperz, John J.

"Sociocultural knowledge in conversational inference" In *Linguistics and Anthropology*. Muriel Saville-Troike, ed. Georgetown University Round Table on Languages and Linguistics, Washington, D.C.: Georgetown University Press. Pp: 191-211.

Hagen, Kenneth, et al.

1985 The Bible in the Churches: how different Christians interpret the scriptures. New York: Paulist Press.

Hall, David (ed.)

1997 Lived Religion in America: Toward a History of Practice. Princeton: Princeton University Press.

Hanks, William F.

1989 "Text and Textuality" In Annual Review of Anthropology. 18: 95-127.

Harding, Susan F.

- "Representing Fundamentalism: The Problem of the Repugnant Cultural Other." *Social Research*. 58(2):373-93.
- 2000 The Book of Jerry Falwell: Fundamentalist language and politics.
 Princeton: Princeton University Press.

Harkin, Michael E.

1997 The Heiltsuks: Dialogues of Culture and History on the Northwest Coast. Lincoln: University of Nebraska Press.

Hymes, Dell

"Toward Ethnographies of Communication: the Analysis of Communicative Events." *American Anthropologist.* 66(2): 12-25.

Irvine, Judith T.

1996 "Shadow Conversations: The Indeterminacy of Participant Roles" In Natural Histories of Discourse, Michael Silverstein and Greg Urban, eds. Chicago: University of Chicago Press. Pp. 131-159.

Johnstone, Barbara

2000 Qualitative Methods in Sociolinguistics. New York and Oxford: Oxford University Press.

Jones, Dale E., et al.

2002 Religious Congregations and Membership in the United States 2000: an enumeration by region, state, and county based on data reported for 149 religious bodies. Nashville, TN: Glenmary Research Center.

Keane, Webb

2004 "Language and Religion" In A Companion to Linguistic Anthropology, Alessandro Duranti (ed.). London: Blackwell Publishing. Pp. 431-448.

Keenan, Elinor

"Norm-makers, Norm-breakers: Uses of Speech by Men and Women in a Malagasy Community" In Explorations in the Ethnography of Speaking.
 Richard Bauman and Joel Sherzer, eds. Cambridge: Cambridge University Press. Pp. 125-143.

Keller, Eva

2005 The Road to Clarity: Seventh-Day Adventism in Madagascar. New York: Palgrave Macmillan.

Kirby, James E., Russell E. Richey, and Kenneth E. Rowe

1998 The Methodists. Westport, CT: Prager Publishers.

Kort, Wesley

1996 "Take, Read": Scripture, Textuality, and Cultural Practice. University Park: Pennsylvania State University Press.

Kristeva, Julia

1986 "Word, Dialogue, and Novel" In *The Kristeva Reader*. Toril Moi (ed.) Oxford: Oxford University Press. Pp. 34-61.

Kuipers, Joel

1990 Power in Performance: the creation of textual authority in Weyewa ritual speech. Philadelphia: University of Pennsylvania Press.

Lakoff, George and Mark Johnson

1980 Metaphors We Live By. Chicago: University of Chicago Press.

Lehtinen, Esa

2005 "Achieved Similarity: Describing Experience in Seventh-Day Adventist Bible Study." *Text* 25(3): 341-371.

Long, Elizabeth

2003 Book Clubs: Women and the Uses of Reading in Everyday Life. Chicago: University of Chicago Press.

Lutz, Catherine

1988 Unnatural Emotions: Everyday Sentiments on a Micronesian Atoll and their Challenge to Western Theory. Chicago: University of Chicago Press.

Malley, Brian

2004 How the Bible Works: an anthropological study of Evangelical Biblicism. Walnut Creek: AltaMira Press.

McGrath, Alister

In the Beginning: the Story of the King James Bible and How it Changed, a Nation, a Language, and a Culture. New York: Anchor Books.

Mertz, Elizabeth

"Recontextualization as Socialization: Text and Pragmatics in the Law School Classroom" In *Natural Histories of Discourse*. Michael Silverstein and Greg Urban (eds.) Chicago: University of Chicago Press. Pp. 229-249.

Metts, Wallis C.

Just a Little Talk With Jesus: an analysis of conversational narrative strategies used by evangelical college students. Unpublished Thesis (Ph.D.), Michigan State University. Department of Interdisciplinary Studies, College of Arts and Letters.

Miller, Donald

1997 Reinventing American Protestantism: Christianity in the New Millennium. Berkeley: University of California Press.

Mould, Tom

2003 Choctaw Prophecy: A Legacy of the Future. Tuscaloosa: University of Alabama Press.

Mouw, Richard J.

"The Bible in Twentieth-Century Protestantism: a preliminary taxonomy" In *The Bible in America: Essays in Cultural History*. Nathan O. Hatch and Mark A. Noll, eds. New York, Oxford: Oxford University Press. Pp. 139-162.

Muse, Erika A.

2005 The Evangelical Church in Boston's Chinatown: a discourse of language, gender, and identity. New York and London: Routledge Press.

Noll, Mark A.

- "The Image of the United States as a Biblical Nation, 1776-1865" In *The Bible in America: Essays in Cultural History*. Nathan O. Hatch and Mark A. Noll, eds. New York, Oxford: Oxford University Press. Pp. 39-58.
- 1992 A History of Christianity in the United States and Canada. Grand Rapids, MI: William B. Eerdmans Publishing Company.

Ochs, Elinor

"Transcription as Theory" In *Developmental Pragmatics*. Elinor Ochs and Bambi B. Schiefflen, eds. New York: Academic Press. Pp. 43-72.

Olson, Daniel V.A.

"Making Disciples in a Liberal Protestant Church" In "I Come Away Stronger": How Small Groups are Shaping American Religion. Robert Wuthnow, ed. Grand Rapids, MI: William B. Eerdmans Publishing Company. Pp. 125-147.

Philips, Susan

- 1974 "Warm Springs 'Indian Time': How the Regulation of Participation Affects the Progress of Events." In Explorations in the Ethnography of Speaking. Richard Bauman and Joel Sherzer, eds. Cambridge: Cambridge University Press. Pp. 92-109.
- 1998 Ideology in the Language of Judges: How judges practice law, politics, and courtroom control. New York, Oxford: Oxford University Press.

Puckett, Anita

2000 Seldom Ask, Never Tell: Labor and Discourse in Appalachia. Oxford: Oxford University Press.

Pulis, John

"'Citing[Sighting]-Up': Words, Sounds, and Reading Scripture in Jamaica." In *Religion, Diaspora, and Cultural Identity: A Reader in the Anglophone Caribbean*. Gordon and Breach Publishers. Pp. 357-401.

Radway, Janice

1984 Reading the Romance: Women, Patriarchy, and Popular Literature.
Chapel Hill: University of North Carolina Press.

Ricoeur, Paul

[1971] 1981 "The model of the text: meaningful action considered as text" IN Paul Ricoeur: Hermeneutics and the Human Sciences. John B. Thompson, ed. Cambridge: Cambridge University Press. Pp. 197-221.

Robbins, Joel

2004 Becoming Sinners: Christianity and Moral Torment in a Papua New Guinea Society. Berkeley: University of California Press.

Roberts, Fredric M.

2005 Be Not Afraid: building your church on faith and knowledge. Herndon, VA: Alban Institute Press.

Sherzer, Joel

- 1983 Kuna Ways of Speaking: an ethnographic perspective. Austin: University of Texas Press.
- 1987 "A Discourse-Centered Approach to the Study of Language and Culture" In *American Anthropologist* 89: 295-309.

Silverstein, Michael and Greg Urban

1996 "The Natural History of Discourse" IN *Natural Histories of Discourse*.

Michael Silverstein and Greg Urban (eds.) Chicago: University of Chicago
Press. Pp. 1-17.

Street, Brian (ed.)

1991 Cross-Cultural Approaches to Literacy. Cambridge: Cambridge University Press.

Svensson, Cai

1990 "The Bible and the real reader: World view and interpretive strategies" In Bible Reading in Sweden: Studies related to the translation of the New Testament 1981. Gunnar Hansson (ed.) Stockholm, Sweden: Almqvist and Wiksell International. Pp. 117-148.

Tannen, Deborah and Cynthia Wallet

[1987] 1999 "Interactive Frames and Knowledge Schemas in Interaction: Examples from a medical examination/interview. In *The Discourse Reader*, Adam Jaworski and Nikolas Coupland, eds. New York: Routledge Press. Pp. 346-366.

Theusen, Peter J.

1999 In Discordance with the Scriptures: American Protestant battles over translating the Bible. New York, Oxford: Oxford University Press.

Thompson, John

1991 Ideology and Modern Culture: Critical Social Theory in the Era of Mass Communications. Stanford: Stanford University Press.

Titon, Jeff Todd

1988 Powerhouse for God: Speech, Chant and Song in an Appalachian Baptist Church. Austin: University of Texas Press.

Treat, James (ed.)

1996 Native and Christian: Indigenous Voices on Religious Identity in the United States and Canada. New York: Routledge Press.

Urban, Greg

1991 A Discourse-Centered Approach to Culture: Native South American Myths and Rituals. Austin: University of Texas Press.

Warner, R. Stephen

"The Place of the Congregation in the Contemporary American Religious Configuration" In American Congregations: Volume 2: New Perspectives in the Study of Congregations. James P. Wind and James L. Lewis, eds. Chicago: University of Chicago Press. Pp. 54-99.

Wimbush, Vincent (ed.)

2000 African-Americans and the Bible: Sacred Texts and Social Textures. London and New York: Continuum International Publishing.

Wind, James P. and James W. Lewis (eds.)

1994 American Congregations: Volume 1: Portraits of Twelve Religious Communities. Chicago: University of Chicago Press.

Wuthnow, Robert

- 1988 The Restructuring of American Religion. Princeton, N.J.: Princeton University Press.
- 1994a Sharing the Journey: Support Groups and America's New Quest for Community. New York: Free Press.
- 1994b "I Come Away Stronger": How Small Groups are Shaping American Religion. Grand Rapids, MI: William B. Eerdmans Publishing Co.

